CREATE HEALTH WAYS OF WORKING INSIGHTS FROM TEN EHEALTH INNOVATION RESEARCH PROJECTS

WRITTEN BY:

WILKE VAN BEEST BERIT GODFROIJ MARIEKE ZIELHUIS DAAN ANDRIESSEN REMKO VAN DER LUGT

This work was supported by the Netherlands Organization for Health Research and Development (ZonMW).

COLOFON:

Editor: Tanja Enninga Layout book: Marry Bassa Layout tool: Charlotte van der Schoot Project support: Natascha Helmer Publisher: HU University of Applied Sciences Utrecht ISBN ebook 9789090360669 ISBN printed version 9789090362144

CONTENTS

	PREFACE	
L	INTRODUCTION	6
2	OVERVIEW PROJECTS	17
	OVERALL PORTRAITS	17
	EVERYDAY SOUNDS OF DEMENTIA	22
	DEMENTIA DYNAMICS IN DESIGN (DDD)	26
	NATALIE	30
	GROWING ROOTS	34
	SQUEALED	38
	TRACK, TRACE & TRIGGER (TTT)	Ч2
	ΡΑΟΟ	Ч6
	HEALTHY STORYTELLING FOR EHEALTH	50
	FOODSAMPLER	54
	GOAL	58
3	WAYS OF WORKING	62
Ч	WAYS TO INVOLVE PEOPLE	100
5	COLLABORATING ACROSS BOUNDARIES	127
6	CONCLUSION	152
7	HOW TO USE THE RESEARCH PATHWAY TOOL	168
	APPENDICES	206
	LIST OF FIGURES AND TABLES	

REFERENCES



PREFACE

"DESIGN TECHNIQUES OFFER A RICHNESS FOR WORKING TOGETHER ACROSS DISCIPLINARY BOUNDARIES."

-PIETER JAN STAPPERS

This book shows how ten projects connect design and health, across the boundaries of research and practice. People from design and health worked together on challenges around dementia, obesity and loneliness. This book is a collection of insights gained through the project Create- Health Innovation Ways of Working Analysis (CHIWaWA) and describes a multiple case study that looked at the richness of ways of working in research projects in which design and health researchers and professionals investigated eHealth solutions together.

The research was done by researchers from the Research Groups Research Competences and Co-Design of HU University of Applied Sciences. The advisory board consisted of Pieter Jan Stappers (Professor of Design Techniques at the Faculty of Industrial Design Engineering, TU Delft), Ellen Moors (Professor of Innovation and Sustainability at the Faculty of Geosciences, University of Utrecht), Jan-Willem van Dijk (Partner & Founder of Greenberry), Alain Dujardin (Creative Director & Partner of Greenberry) and Karin Alfenaar (Program Manager Regional Innovations Health Hub Utrecht). We would like to thank the members of the advisory board for sharing their insights during the start and the end of the project. We asked them to respond to the insights of the book from their own background and expertise. We are pleased to share some of these responses here:

In his reaction to the book, Pieter Jan Stappers mentions the importance of research into the application of creative ways of working: "With design thinking, part of the tools of design has become more widely applicable, but it is important that the tools are always part of deeper design techniques. That is easily lost when tools are simplified to a persona or journey template. That is precisely why it is great that research has been done into the application of creative ways of working in health research." Alain Dujardin indicates after reading the conclusions that he wants to emphasize that experience shows that it is important to trust designers to apply creative ways of working in projects in the healthcare context: *"It requires flexibility from healthcare professionals and researchers to be open to design methods."* Karin Alfenaar, in her response to the book, emphasizes the importance of having a conversation with all stakeholders: *"From the Health Hub Utrecht, we want participation of all stakeholders to be a part of all plans."*

To be able to introduce creativity and creative methods into eHealth innovation processes, one needs so-called boundary spanners or boundary brokers between disciplines. Ellen Moors mentions as important - and recognizable - insight from the book: *"The deployment of boundary brokers is crucial for pushing boundaries. Creativity brings in an element that allows you to go outside the standard routines of health care. Creativity can be a preparation for a yet unknown pathway, a challenge or way of working that we do not yet know, but may need in the future."*

With this book, we aim to provide insight into how Create Health ways of working can contribute to the innovation and research process and the people involved in it. We were able to gain these insights because we were allowed to engage with multiple projects and draw overarching conclusions. We therefore thank ZonMw for the opportunity to conduct meta-research into the use of creative ways of working and the added value of using these methods to achieve new innovations in healthcare. We hope that this meta-research will be helpful to researchers who are looking for solutions with and for healthcare from the design and health domain. In addition, we also hope that the insights will help grant makers to set up new research programs.

We would like to thank all project leaders, PhD students and post docs who wanted to talk to us about their project: Pieter van Gorp, Raoul Nuijten, Annemarie Braakman-Jansen, Christian Wrede, Thomas van Rompay, Josca van Houwelinge-Snippe, Emely de Vet, Lean Kramer, Rens Brankaert, Maarten Houben, Sebastiaan Peek, MyrteThoolen, Amy van Grieken, Sophie Korenhof, Job van 't Veer, Bard Wartena, Valentijn Visch, Niko Vegt and Natalia Romero Herrera, as well as many other people who were involved with the projects, one way or the other. Thank you all for sharing the lessons learned from your projects during our conversations and interviews, and during the final stage of the research project. We hope that we have done justice to your insights and hope that they empower people who read this book to strengthen their create-health collaboration in both research and practice.

June 2022

Wilke van Beest, Berit Godfroij, Marieke Zielhuis, Daan Andriessen, Remko van der Lugt INTRODUCTION

Monday morning at our ConceptSpace: After the first round of studying the project proposals and auxiliary information from 10 different projects, we have come together as a research team to spend a full day analyzing data. The co-design researchers prefer an open approach, utilizing the room as a "shared external memory," using scissors and tape to organize fragments on the wall, then looking for connections and generating mini-theories. The team members from the research methodology group are familiar with different approaches to analyzing data and developing coding schemes and are more familiar with identifying and coding fragments on the computer, using Atlas TI as a software program. We end up alternating between different tasks in the room, recording insights using Atlas TI, then printing these out for another round of interpretation. Even though this is a hassle, it helps us take time to engage with each other's approaches to research, which, in turn, helps develop shared understanding throughout the project. Moreover, the Create-Health research team needed some boundary-crossing events to start functioning as an interdisciplinary research team.

In recent years, the application of creative ways of working in the domains of healthcare and well-being has become increasingly popular, both in research and in practice. For instance, the personas, fictive personalities that represent needs or motives of people involved are constructed from actual user research. Another example is a patient journey, which is a variation of the user journey as a central tool in service design (Simonse, Albayrak & Starre, 2019). The patient journey is the pathway that a patient takes through the healthcare process, from (before) the beginning to (after) the end of the given care or treatment. These approaches have become common practice to such an extent that when I talked to a healthcare administrator, she mentioned that "The way WE do it in our field is to create a patient journey" (personal conversation with the director of a paramedic school). However, many of the creative tools and methods are used in instrumental ways without connecting to more structural applications or integrating the ways of working involved. This is amplified in research projects, where creative and healthcare research traditions are quite different in their respective epistemological stances and research approaches. Even when tools and methods have been incorporated, it remains unclear whether these approaches improve health innovation and research processes. Moreover, what are the obstacles, and how can we learn from research groups that successfully marry the different approaches? With the Create-Health program, the Dutch research funding agency ZonMw created a unique opportunity to study the application of creative ways of working in health research. The program aimed to bring parties from the creative and the health sectors together to develop knowledge building blocks on eHealth applications in elderly care. We followed 10 projects over the course of three years and studied the way they combined research and innovation methods from the creative and health sectors. This allowed us to answer the following main research question that we address in this book:

"WHAT WAYS OF WORKING ARE EMPLOYED IN RESEARCH PROJECTS ON EHEALTH INNOVATIONS IN WHICH RESEARCH AND PRACTICE PARTNERS FROM BOTH THE CREATIVE AND THE HEALTH SECTOR COLLABORATE, AND HOW DO STAKEHOLDERS BENEFIT FROM THESE WAYS OF WORKING?"

This chapter introduces and explains these creative ways of working in healthcare, describing what we set out to investigate and how we approached this meta-research.

PREJUDICE AT WORK

Making design work in the field of healthcare requires interdisciplinary research with close collaboration between researchers from the creative and the healthcare sector. However, the intensive collaboration between different disciplines is often not without its challenges, and healthcare and creative professionals come from different worlds that do not automatically align. This was illustrated in a small experiment we conducted at a Design for Health conference in October 2018. We challenged a group of 70 researchers and professionals working in the healthcare and creative industry to name as many prejudices and assumptions about the other sector as they could.

A total of 290 comments, evenly divided between both fields, were written down, folded into paper airplanes, and thrown on stage. The creatives considered their healthcare counterparts bureaucratic, conservative, and rule-oriented but also caring and people-oriented.

When it came to healthcare research, they believed it was evidence-based and scientific but also too focused on randomized controlled trials. The healthcare professionals viewed it as creative but also impractical and chaotic. When it came to creative research, they thought it lacked rigor, even to the extent that some used the word "dangerous." What constitutes creative ways of working? Is it a mindset? A skillset? Methods and tools? Or all of the above? We found that they are difficult to define. The ways of working are visible in the applied tools and methods, but they can also be seen as the skills and motivations that manifest in the ways people engage in innovation and research projects.

In this book, we will not try to demarcate or define creative ways of working but instead provide examples of working methods employed by the 10 projects. These examples show the diversity of ways of working that are possible when parties from the creative and health sector unite in research, as shown in Figure 1.1. The ways of working observed did not come strictly from the creative sector. In fact, in many cases, it was difficult to identify their origins. Based on the analysis of the 10 project proposals, we developed a card sorting exercise with the 10 project leaders and the steering committee of the Create-Health program. A textual analysis of the 10 project proposals revealed about 100 activities that could be considered ways of working. During the exercise, we tried to identify the origins of these ways of working by asking the participants whether they thought one was from the creative or the health sector. The general conclusion was that, in many cases, it was not possible to say. Therefore, we refer to the ways of working described in this book as "create healthier ways of working."



Figure 1.1: When creative and health sectors unite

The funding agency ZonMw did not define creative ways of working in their Create-Health call either, which gave the 10 participating projects room to choose their own research design and approach. However, the funding agency highlighted three aspects that all projects needed to comply with. These aspects defined the boundaries of the phenomena under investigation in our research. We will be looking at the following:

- The use of specific methods for research and innovation: The ZonMw call stated that "through an iterative and, at the same time, designerly research method (for example, research-through-design), projects can produce knowledge building blocks during the research period" (p. 5). Although not strictly a prerequisite, this sentence indicates that the funding agency was thinking of iterative research methods that apply design activities as part of the design. We will be looking at methods and approaches that apply these principles.
- The inclusion of the target group: Clients, patients, and healthcare professionals all needed to be involved in the research process. We will be looking at how beneficiaries and other actors involved are included in research activities.
- Collaboration: The call demanded that parties from both the creative and health sectors as well as parties from research and practice work together. The expectation is that this collaboration will create an interdisciplinary approach that will lead to better innovation. We will be looking at how this collaboration takes shape and whether the collaboration can be characterized as interdisciplinary, multidisciplinary, or transdisciplinary.

Thus, in this book, we look at the application of methods and approaches to research and innovation in Create-Health projects (Chapter 3), ways to involve target groups in research and innovation (Chapter 4), and ways to collaborate between creative and health sectors (Chapter 5).

OUR METHOD

In the project, which we named "Creative and Health Innovation: Ways of Working Analysis" (CHIWaWA), we conducted cross-cutting research on the use of Create-Health ways of working in health innovation research by studying 10 research projects on eHealth innovation in the Netherlands. Researchers from the research group "Research Competence" and the research group "Co-design" – both based at the Utrecht University of Applied Sciences in the Netherlands – conducted this meta-research between 2018 and 2022. CHIWaWA and the 10 research projects involved are part of the ZonMw program "Create-Health – eHealth Knowledge Base for a Healthy and Active Old Age." ZonMw is a Dutch funding organization for health research and care innovation.

RESEARCH DESIGN

Our study was set up as a multiple case study design (Yin, 2009). The selection of cases was convenience sampling, as the funding agency had already made the selection. However, we found out early on in the process that each case was unique in terms of the consortium members, the topic of the research, the research approach, and the collaboration method. Therefore, each case was studied independently, and a cross-case analysis was applied to the 10 projects. A consortium of various actors from academic and applied research institutes in the Netherlands was involved in each project, as well as partners from the creative industry and health sector. For each of the 10 projects, actors from the creative sector and health sector collaborated to create new eHealth products and services and to generate knowledge, bringing ways of working to the table from their own sectors. The general presupposition of the Create-Health program was that these creative ways of working in innovation and research might lead to a better quality of the innovations as well as further development of interdisciplinary practices between creative practices and research and health practices and research impact). The cases are presented in Chapter 2.

In our approach to this multiple case study, we were guided by two important principles:

- We involved the researchers from the 10 projects in our own project whenever possible. For example, project members from the 10 projects were invited to become co-authors of our publications
- 2. We tried to ensure that our research activities would benefit both the participating projects and our cross-cutting research. Therefore, we minimized the use of questionnaires and limited the number of interviews. We also used the joined meetings that ZonMw scheduled to gather additional data when possible.

Within the multiple case study we applied the following data gathering methods:

Exploratory literature review: We performed an initial literature search on collaborations between creative and healthcare professionals that involved related methods, boundary-crossing, and impact. This search guided the preparation of a checklist and protocol for both the analysis of documents and the first series of interviews.

Taking stock: We analyzed the 10 project proposals and held preliminary interviews with the project leaders, often accompanied by the main researcher. The interviews were done by two of the CHIWaWa team members during site visits. In addition to the general check-list questions, specific issues encountered in the project proposal were discussed. We also assigned one of our researchers as the main case-holder to maintain regular contact with the project leader.

Monitoring the projects: While monitoring the projects, we collected two kinds of configurational process data: research elicited data and naturally occurring data. Research elicited data was collected by questioning linked and unlinked actors from each of the 10 projects. The data was gathered both in formal and informal settings, in semi-structured interviews, but also by asking topic-specific questions in more informal settings (e.g., face-to-face during coffee breaks of knowledge sharing sessions and monitoring meetings). The informal questioning focused on inductively developed themes and events that emerged during the process. For instance, when project actors described specific events as very creative, we would ask questions about the ways of working, such as why they thought the ways of working were creative or how different actors collaborated. The naturally occurring data consisted of information that was observed or overheard through participative observation in knowledge sharing sessions and monitoring meetings, as well as report analyses from the 10 projects, including project proposals, presentation slides, progress, and other kinds of reports, publications, posters, and the websites from the 10 projects. We documented this data through field notes, audio and video recordings, and a digital project dossier.

Assessing impact: Once the 10 projects reached their knowledge extension phase, interviews were held with the project leaders, as well as one or more key users from each project, about their interaction with the project, their awareness, and interpretation of the results and the role of creative ways of working. In addition, the practical results and knowledge products developed by the 10 teams were analyzed.

Overall, 35 interviews were conducted. The project leader from each project was interviewed three times, as well as four practice partners and the ZonMw program leader. The interviews were conducted by following a topic list. In many cases, supporting material was used, including an actor map and a project timeline.

OUR THEORETICAL LENS

This book examines what happens when researchers and practitioners from the creative and healthcare sectors join forces in research and what this means for the ways of working. In a three-year period, the 10 projects took on many activities that we had to structure accordingly to study them. As a basis for this structure, we went back to one of the key characteristics of creative ways of working described above: the iterative way of working. In the iterative way of working, the researcher moves between theory and practice several times during the research. We found Nelson and Stolterman's (2003) distinction helpful here, as used by Jonas (2006), between *what should be* (theory), *what could be* (concepts), and *what is* (practice). We were able to plot the sequences of activities from the 10 projects in these three categories. We used the Research Pathway Model (RPM) in order to do so (Van Beest et al., 2021). The RPM is a process model that makes explicit the research steps taken toward impact. The model proposes that research takes place in three contexts: the theoretical context, the conceptual context, and the real-life practice context:

"In the theoretical context, the research is focused on creating, exploring, and delivering a better understanding of problems and related propositions for solutions, which are advanced and not verified in practice yet. The conceptual context translates presupposes solutions to a more specific prototype that is created, explored, and made.

In this context, researchers, project partners, and other stakeholders bring in experiential knowledge to translate theory to prototype, by themselves, together with the consortium, and with or without the end-users, but always in a protected niche as a brainstorming room, a pilot environment, or in the context of a pilot organization. The real-life practice context concerns the context in which the prototype is tested in the world of professional practice and/or the living environment of end-users" (Van Beest, 2021, p. 7).

The model also distinguishes three clusters of activities: "understand and create," "explore and test," and "deliver and implement." The "understand and create" activities refer to getting a grasp on the issue at hand, understanding the problem, and creating ideas. Some examples include a literature review (theoretical context), co-creation workshops (conceptual context), and contextual interviews with people in their own context (real-life practice context). The "explore and test" activities relate to exploring or validating an idea, concept, construct, or solution through research activities. Examples include double-blind lab experiments (theoretical context), a pilot in a living lab (conceptual context), and a prototype test in the environment of the end-user (real-life practice context). The "deliver and implement" activities entail the transfer of the insights gained. Examples from this category include writing academic journal articles (theoretical context), handing over a prototype to the market (conceptual context), and implementing a new protocol (real-life practice context). Examples are: academic journal articles (theoretical context), handing over a prototype to the market (conceptual context), implementing a new protocol (real-life practice context).

This results in a 3x3 matrix of activities for the different realms and activities (Figure 1.2). In working with the RPM together with researchers from the 10 projects, we found we needed to slightly adjust the names of the three types of activities. Van Beest et al. (2021) referred to these activities as create, explore, and deliver. Researchers with a design background were confused by these terms since many were used to the Double Diamond terminology from the British Design Counsel: discover, define, develop, and deliver. The first three Ds of the Double Diamond describe the process of understanding the issue, defining it, and codesigning. This is covered by the first row of activities. For the sake of clarity, we designated this row as "understand and create." The last D describes the process of testing solutions on a small scale, which is equivalent to our second row of activities. Therefore, we designated this row as "explore and test." The Double Diamond model does not include activities that produce and implement results from research. Because the RPM is about research and innovation, the RPM does include these types of activities due to its focus on research and innovation; therefore, we categorized these activities as "deliver and implement." These changes to the model were made to accommodate it to the specific context of the Create-Health projects and were approved by the authors of the RPM.

Van Beest et al. (2021) have demonstrated that different paths can be chosen and that there is no order of activities that is better than the other. In this book, we will use the RPM as a descriptive model to help provide structure across the different cases and sub-research questions.



Figure 1.2: Research Pathway Model (RPM)

Chapter 4 examines how the Create-Health projects involved end-users and other actors in their research. To better understand this process, we used the Service Dominant logic theory (Vargo & Lusch, 2004). This theory describes innovation projects as an exchange of services rather than the creation of goods. By looking at services delivered by all project participants, including representatives from the target group, we addressed an important question raised by these types of projects: "what's in it for me?"

To further describe how partners collaborate within these research pathways, we relied on the Boundary Crossing Theory, which is part of the Cultural-Historical Activity Theory (e.g., Akkerman & Bakker, 2011). This theory provides the concepts that we used to describe the characteristics of the collaboration between the partners (see Chapter 5).

OUR OWN COLLABORATION CHALLENGES

Our multidisciplinary team consisted of three main researchers, supervised by two professors and supported by auxiliary staff (graphic designers, research assistants, and project managers). Our research groups are strongly rooted in the Research-through-Design tradition from the design sciences (i.e., co-design) and the Design Science Research tradition of the organizational sciences. We had our own challenges to tackle by overcoming differences and finding synergy. For instance, in the early data interpretation meetings, the co-design team preferred to conduct data analysis through in-the-room, on-the-wall, open-ended coding with taping, cutting, and annotating involved. In contrast, the research methodology researchers were more inclined to develop a coding scheme in Atlas TI. By alternating between different methods, we developed collective ways of dealing with and interpreting data.

AND THEN COVID 19 HIT

We originally intended a highly participatory research process, with numerous workshops for data interpretation and triangulation with the projects' participating researchers. We organized a card sorting exercise for the first-year review meetings to discuss terms found in the project proposals. For the second-year session, we had prepared posters with the thematic ordering of issues found to be discussed with the project teams. However, that is when the first lockdown started, which dramatically changed the program dynamics. The projects struggled to find new ways to engage with their target groups and collaborate between health and creative settings. Person-to-person meetings became rare. Thus, the lockdown significantly impacted the Create-Health ways of working employed. We organized a knowledge exchange workshop to support the research groups and gain insight, which ultimately led to a white paper.

STRUCTURE OF THE BOOK

The book's structure is as follows: Chapter two provides an overview of the 10 projects and their key characteristics. Chapter three addresses the Create-Health ways of working and the insights gained during the research process. Chapter four looks at the collaborations with the target group(s) and consortium members. Chapter five focuses on the boundary-crossing issues between creative and healthcare academics and practitioners. Chapter six provides some overarching insights and conclusions. Finally, Chapter seven offers a practical guide for using the Research Pathway Model in other projects.

The reference list and the list of tables and figures for all chapters can be found in the appendixes.

CHAPTER 2

EHEALTH INNOVATIONS TO DEAL WITH ISSUES AROUND LIVING WITH DEMENTIA, LONELINESS, AND OBESITY

CREATE HEALTH WAYS OF WORKING

In the meta-study, we followed and studied 10 projects that are all part of the Create-Health program of ZonMw. The Create-Health program aims to contribute to the social challenge surrounding healthy and active aging. Create-Health brings parties together from the creative industry and the health care and welfare "industry," supporting knowledge development, knowledge sharing, gaining access to money, and supporting scaling up.

The research within Create-Health is focused on three themes, in which the elderly or aging people are the main target group:

- Living with dementia;
- Preventing loneliness in the frail elderly;
- Prevention of overweight and obesity.

The call suggests a broad choice of research topics within the themes, such as focusing on a user interface that works for people living with dementia or developing new validation methods to determine the effectiveness of applications. As a result, there are various ambitions and goals within the various projects, with different perspectives on innovation and on the timeframe of the innovation path.

Each of the 10 studied projects focused on one or two of the three themes. In a number of projects, we saw a combination of the themes "living with dementia" and "preventing loneliness."

Therefore, we have chosen to combine these two themes in the general description in this paragraph, with one side note: the project "Growing Roots" did not focus specifically on people living with dementia but rather on frail elderly more in general.

THEME: LIVING WITH DEMENTIA AND PREVENTING LONELINESS IN THE FRAIL ELDERLY

Within the last decade, it has become more common for elderly people to live at home rather than being admitted to a care home or a nursing home. The frail elderly are encouraged to live at home as long as possible, including those living with dementia. This is not only to reduce the increasing costs of health care but also to improve quality of life by continuing to live in familiar surroundings. Medical care and self-management are key to making it possible to live at home, in which eHealth could play a role. The projects within these two themes each address different aspects of the opportunities provided by eHealth. Most projects focus on the initial stage of application development. However, Everyday Sounds of Dementia also conducts research in a nursing home, where patients are often in more advanced stages of dementia. Several projects studied eHealth in relation to the possibilities and desirability of monitoring people. Because all kinds of ethical issues are involved, this is still in its early stages, and such projects take a more fundamental approach with a focus on knowledge development.

The following projects mainly focus on living with dementia and preventing loneliness in the frail elderly:

- Everyday Sounds of Dementia focuses on the use of sounds to calm, reduce agitation, stimulate, excite, and engage people living with dementia;
- Dementia Dynamics in Design (DDD) focuses on promoting the independence and quality of life of people with dementia by developing eHealth solutions that can prevent loneliness in people living with dementia (see below);
- Need Articulation Through Autonomy Loss in the Elderly (NATALIE) aims to develop a theoretical framework that describes design principles and methodological procedures that guide the design of applications to minimize difficulties in the dialogue between patients and (in)formal caregivers;
- Growing Roots: Connecting Elderly through Virtual Nature Spaces aims to develop a framework for creating a virtual natural environment that inspires feelings of connectedness and stimulates social contact among the frail elderly;

- Support Quality Care for Elderly Using Ambient Living Environment Data (SQUEALED) focuses on monitoring independent-living elderly persons via smart energy meters, which are already installed in >30% of all Dutch households;
- Track, Trace, and Trigger! (Unobtrusive Sensing Technologies to Monitor and Coach Elderly People with Dementia) aims to develop sensory technology to identify an individual's daily dynamic activities (track, trace) and trigger (social) behavior and communication with music cues;

THEME: PREVENTING OVERWEIGHT

Obesity is a complex problem. People who are overweight have a higher risk of developing health issues. The projects within this theme focus on knowledge development about the prevention of obesity and the development and maintenance of a healthy lifestyle. The four projects in this theme show a diversity of target groups and eHealth interventions that could support users in developing a healthy lifestyle.

The projects with a focus on this theme include the following:

- Designing Persuasive eHealth Agents for Coaching Older Adults Toward Dietary Behavior Change (PACO) studies the use and usefulness of virtual eHealth agents to prevent obesity in the elderly;
- Healthy Storytelling for eHealth proposes storytelling as a persuasive eHealth element to motivate people with low literacy skills for obesity prevention;
- *FoodSampler* aims to determine the best practices of dietary reporting in daily life and the contextual factors that influence dietary practices;
- Gamification for Obesity Prevention and Active Lifestyle (GOAL) focuses on gamification as a technique in eHealth solutions to tackle patient engagement and motivational issues.

PROJECT PORTRAITS

In each case portrait, the project is described in general terms, including the *topic* of the project, the project's target group, the theoretical *background* of the project, the project actors, what happens over the course of the project, and the *results*. An illustrative quote is added to vivify the factual project descriptions.

The Venn diagram (Figure 2.1) of the Actor map represents three contexts: the theoretical context (blue), the conceptual context (yellow), and the real-life practice context (orange-red). The contexts and colors match the RPM (see p. for a general introduction to the model). The project actors are plotted according to their main role in one or more of these contexts.

The project actors (Figure 2.2) are visualized not as individual participants but as representatives of a type of organization or a role in the project.



Figure 2.1: Actor Map Venn diagram



Figure 2.2: Project Actors

The project actors are divided into target group individuals and target group representatives (orange), practitioners such as health care organizations and creative agencies (green), and knowledge partners such as universities and universities of applied sciences (purple).

The attributes of the actors visualize the focus of their role in the project. The shield is used to visualize the representation of a certain target group. The pencil represents a design perspective, and the stethoscope represents a healthcare perspective. The tie stands for a partner in the project. Combined with the main roles, one figure could represent a university (purple), for instance, with a design research focus (pencil) or a healthcare focus (stethoscope).

The course of the project is described and visualized in a project timeline. As most projects have a duration of four years, the timeline highlights only the main events or turning points in the project (numbered). As a result, the reality is overly simplified. Based on the Research Pathway Model (RPM) (see Figure 2.3) and its corresponding colors, the timeline colors read as follows:



Figure 2.3: Research Pathway Model

The circles correspond to the theoretical context; solid blue stands for the "understand and create" phase, dotted blue for the "explore and test" phase, and striped blue for the "deliver and implement" phase. The same goes for the yellow circles in the conceptual context and the red for the practical context: solid for the "understand and create" phase, dotted for the "explore and test" phase, and striped for the "deliver and implement" phase.

The *results* are clustered according to the RPM into *knowledge building blocks* (results in the theoretical context), *prototypes* (tangible results in the conceptual context), and *practice* (*results*). In line with the aim of each project, not every case resulted in an evenly divided number of results for all three contexts.

EVERYDAY SOUNDS OF DEMENTIA

TOPIC & FOCUS

The project Everyday Sounds of Dementia focused on the use of everyday sounds – such as waves crashing or bird songs – to calm, reduce agitation, stimulate, excite and engage people living with dementia. The project investigated the added value of such sounds for people with dementia, and developed and evaluated audio-based technology to explore the opportunities for immediate meaningful impact in care practices. Additionally, the project developed insights into collaborative methods that enable people living with dementia and their caregivers to engage in co-design and participatory approaches.

"WHEN LISTENING TO THE SOUND OF THE WIND AT THE BEACH. A PARTICIPANT DESCRIBED A JOYFUL MOMENT SHE ONCE EXPERIENCED ON THE SEASIDE BOULEVARD."



The project drew on the large body of evidence about how music provides emotional and behavioral benefits for people living with dementia. The project built on the concept of soundscapes: (everyday) sounds that help people build unifying relationships with their environment (Schafer & Murray, 1977). The impact of sound, and how meaning can be created through soundscapes for people living with dementia, is under-researched. The need to involve people living with dementia, and their caregivers in the collaborative design of technology is increasingly being recognized (Treadaway et al., 2015; Kenning, 2017) but remains an under-addressed issue.

The project also drew on the view that well-designed technology can play a crucial role in supporting people to live independently at home (e.g., Brankaert, 2016). For an intervention to be successful, the design approach needs to take into account individual wants, needs, likes and dislikes, circumstances, personal context and environment, and existing relationships with family members and caregivers (Brankaert, 2016).

PROJECT ACTORS

The project management and core researchers of this project were part of a design research group at a university with expertise on design for dementia. Researchers from other universities were closely involved in the knowledge development, bringing expertise on design and dementia and the social and technical aspects of care innovation. The healthcare organizations were involved in gaining access to the target group, contributing to co-design activities and in employing prototypes. The target group of people with dementia was involved in co-creation activities in all studies. A network organization was tasked with dissemination to care and welfare organizations, knowledge institutes, companies (incl. SMEs) and government organizations. During the third study, a design practice partner was involved in the technical development of the final prototype.



TIMELINE

A Research-through-Design approach was adopted through which research artifacts (i.e., interactive, digital, or physical objects) were developed and used (Hummels et al., 2011). A participatory, person-centered, and reciprocal approach was used to involve people with dementia in the process (Kenning 2017), providing activities with interactive audio-based technology that stimulate agency, playfulness, and social engagement. The project developed further insights into this participatory and reciprocal approach: how to involve people with dementia not only by watching and observing them but by actively involving them in all phases. This was most prevalent in the final study, in which the researcher worked closely with three couples in user research, design, and evaluation.

The project consisted of three sequential studies. In all three studies, research artifacts were used. The first studied the phenomenon of sound & dementia and took place in a daycare context (sessions with the researcher present). In the second study in two different care homes, a research artifact was deployed in actual practice with a big role for caregivers in this use. The last study involved three couples (a person with dementia and a caregiver) who co-developed and used a research artifact in their home.



Figure 2.5: Timeline



Figure 2.6 (Left): The "Soundboard" plays everyday sounds when objects are on top of it, such as the sound of the sea when a shell is placed. This prototype was used in a workshop setting at a daycare facility to study sound in relation to dementia.

Figure 2.7 (Right): A research participant with dementia interacts with the "Tumbler", a prototype that fosters initiative and agency through the rediscovery of familiar everyday sounds. The research participants were closely involved in the development of this prototype.

RESULTS

KNOWLEDGE BUILDING BLOCKS

- Insights into how everyday sounds evoke memories, emotional experiences, and a sense of social belonging during activities at a daycare center (Houben et al., 2019).
- Insights into how everyday sounds stimulate meaningful conversations, playfulness, curiosity, and verbal and nonverbal contact with people in advanced stages of dementia (Houben et al., 2020).
- Demonstration of opportunities for everyday sounds and design in the home environment to provide social activities through storytelling and exploring selfhood and identity (Houben et al., 2022a, Houben et al., 2022b).
- Participatory co-design guidelines on how to engage participants in workshops and in their natural context to identify and collect personal and generic sounds, categorizing them according to source and potential emotional or behavioral impact (Houben et al., 2019).

PROTOTYPES

- Soundboard: a design artifact that enables people in the early to mid-stages of dementia to listen to and interact with various soundscapes (Figure 2.6).
- Tumbler: a prototype that enables people with dementia to initiate and participate in meaningful activities with their partner in everyday home settings (Figure 2.7).

DDD: DEMENTIA DYNAMICS IN DESIGN

TOPIC & FOCUS

The DDD project aimed to understand the use of technology regarding social health and social participation of independent-living people with dementia. The researchers explored how people with dementia accept and use technology to maintain social contacts and how the use of technology changes over time

as people go through various cognitive, physical, and social changes as they age. The DDD project designed a high-fidelity prototype of a supportive and sustainable eHealth solution "LivingMoments" to reduce the risk of loneliness by improving or maintaining social participation. DDD combined expertise in technology acceptance and product design research by applying a Research-through-Design approach to adapt and evaluate a theoretical framework. DDD had three objectives: 1) Improve knowledge about technology acceptance in people with dementia, 2) design eHealth solutions for social needs in people with dementia, and 3) allow third parties to apply this body of knowledge to product offerings and the implementation of technologies.

"QUICK AND DIRTY PROTOTYPING IS OFTEN USED IN CO-DESIGN, WHILE A WORKING PROTOTYPE PROVIDES MORE IN-DEPTH INSIGHTS INTO THE EXPERIENCES AT THE TIME OF USE."



The project partially drew on a theoretical framework regarding technology acceptance by older adults who are aging in place (Peek, 2017). The framework addresses the changes that elderly people experience and how these changes affect technology use (Peek et al., 2014; Peek et al., 2016; Peek et al., 2017a; Peek et al., 2017b). In DDD, researchers adapted the theoretical framework to the specific circumstances and characteristics of independent-living people with dementia. They also evaluated how the adapted framework can contribute and be enhanced by the design of supportive and sustainable eHealth solutions for preventing loneliness of independent-living people with dementia and their caregivers (formal caregivers and family members): How they experience technology use in relation to social participation.

PROJECT ACTORS

The core researchers of this project were a post-doc researcher at a university faculty in social behavioral sciences and a PhD student at a university faculty in industrial design. The PhD design researcher held weekly meetings with a supervisor (assistant professor) in the industrial design department, who was also involved in writing the project proposal with the behavioral sciences post-doc. The latter was project lead in DDD and supervised by a full professor in social behavioral sciences, who was the main applicant. Moreover, two other university partners were involved in a mental health organization and were part of the advisory board. A local innovation network for active and healthy aging was also part of the consortium. This practice partner was particularly involved in recruiting participants, disseminating results, and making connections with healthcare institutions, technology companies, and municipalities. Figure 2.8 shows the project actors mapped according to the theoretical, conceptual, and real-life practice contexts.



TIMELINE

The research design was an applied Research-through-Design and consisted of a process in which participatory design techniques were applied to design and evaluate supportive and sustainable eHealth solutions. Central to the project's way of working was the combination of longitudinal field research with design research. Longitudinal interviews and contextual interviews with design tools took place in parallel and informed each other.

Three rounds of interviews were conducted, during which the social behavioral research focused on the current situation and the same questions in each follow-up interview to observe change over time (which can be expected regarding the technology use by people living with dementia). The contextual interviews focused on creating a visual understanding with the participants through three tools: A social mapping tool, a diary probe, and a hobby-and-activity tool (Den Haan-Wintermans et al., 2019). Based on the findings of the combined study, the researchers developed a prototype (Living-Moments), which was due to COVID-19 an accelerated development, not leading to a quick-and-dirty prototype which is common in design research, but a high-fidelity prototype that participants could use in their homes without a researcher being there. Health experts and consortium members provided feedback on the first ideas of the prototype in an online co-design session. The prototype was tested for at least six weeks at the homes of people living with dementia and their care providers. Researchers from both disciplines published their findings (knowledge generated) in interdisciplinary outlets.



Figure 2.9: Timeline



Figure 2.10 (Left): LivingMoments prototype

Figure 2.11 (Right): Participants receive a postcard from family members via the LivingMoments prototype

RESULTS

KNOWLEDGE BUILDING BLOCKS

 DDD provided knowledge regarding the experiences, problems and needs of community-dwelling older adults with dementia that are related to their social participation and the use of communication technologies.

PROTOTYPES

- DDD delivered a stand-alone prototype of a social health research product, a device that enhances social participation by allowing the creation and receiving of instantly printable postcards: LivingMoments (see www.livingmoments.nl). LivingMoments supports people with dementia to establish and maintain social contact by bridging the gap between modern technology and historical but more familiar means of communication, such as sending postcards. LivingMoments won the accelerator award from IllionX during the TU/e contest 2021.
- Furthermore, DDD created a social mapping tool as a research artifact specifically designed to conduct qualitative research among community-dwelling older adults with dementia.

PRACTICE

 DDD delivered guidelines for designing the participation of community-dwelling older adults with dementia and experiences and tips for interdisciplinary research projects in particular projects that involve behavioral and design researchers.

NATALIE (NEEDS ARTICULATION THROUGH AUTONOMY LOSS IN ELDERLY)

TOPIC & FOCUS

The project NATALIE focused on the difficulties in the dialogue between patients and (in)formal caregivers. The project aimed to generate knowledge that helps to design adaptive, user-friendly applications for people with dementia that persuade them to overcome impediments to talking about their illness, support them to enter the dialogue with their (in)formal caregivers, and assist in creating better insight into what support they need to uphold independent living.

The project provided a framework based on the Research-through-Design approach that should make the design of suitable (digital) applications for this target group more appropriate and effective.

"ULTIMATELY IT WAS A THEORETICAL FRAMEWORK THAT WE TESTED, REFINED AND VALIDATED DURING THE RESEARCH-THROUGH-DESIGN PROCESSES. BUT THEN WE ALSO LOOKED AT HOW WE COULD BRIDGE THE GAP BETWEEN THE CREATIVE INDUS-TRY AND HEALTHCARE. THAT IS WHY WE STARTED TO FOCUS ON DESIGNING DESIGN GAMES." The theoretical background of the project was based on persuasive technology to develop user-friendly, motivating, and adaptive eHealth applications. The participation of people with dementia in decision-making about their own situation was not self-evident; overall, (in)formal caregivers tend to make decisions for them rather than with them (Dupuis et al., 2011). Many scholars argue that it is important to involve people with dementia during the eHealth innovation process (Malinowsky et al., 2013; Lindqvist et al., 2013; Span, 2016). The consortium's practice partner gathered valuable knowledge about specific applications that assist in the dialogue between elderly care consumers and caregivers and developed an application to establish suitable care planning (Van

den Berg, 2013). Based on the Care-Dependency Scale (Henderson's framework of human needs, Dijkstra et al., 1996; Dijkstra, 1998; Henderson, 1966), this application supports care professionals and patients in the shared decision-making process, focusing on activities the person can perform as independently as possible. Although Span and colleagues (2015, 2016) identified a set of user requirements for digital shared decision-making tools for people with dementia, knowledge about the use of persuasive technology as a conceptual perspective to identify the design principles needed to develop eHealth interventions that aid this patient-centered approach was still missing and therefore formed the starting point for this project.

PROJECT ACTORS

The project management and core researchers of this project were a PhD student and a researcher from a university of applied sciences. The researchers at the universities were part of the advisory board. One of the health organizations and the target group were involved in data collection during the project. A consortium was formed with related projects in the region to share knowledge and experiences from practice.



TIMELINE

The project started with a literature review to establish a framework and an explorative field study with healthcare professionals using personas, a day-in-the-life approach, contextual interviewing, cultural probes, and context mapping. This was followed by an exploratory phase: five studies in which design, development, and evaluation took place. These studies partly informed each other and partly ran parallel. The five studies were set up based on the Research-through-Design approach. Artifacts were designed together with the target group. Gaining knowledge from making these artifacts has been carried out to reduce the dependency of people with dementia and offers insight into a form of

discussion in which people with dementia and their informal caregivers can make joint decisions. In this co-creation process, the researchers used, among other things, objects that were created through the process of abstraction, which made them useful during the conversation.

The main purpose of the prototype was to develop principles from which the framework could be further developed and then expanded and refined. Based on this, a framework and a whitepaper were developed.



RESULTS

Based on the framework, the experience of autonomy that is created through communication and decision-making was examined. Balancing the experience of autonomy in communication and decision-making was an important starting point of the design process. The design principles and triggers to maintain a balance in autonomy were converted into a toolkit for the design sessions. During co-creation, we looked at how this toolkit was used and its concrete impact on the design choices for the prototypes from the design sessions (e.g., wicked circles and hackathons). Through these activities, the researchers gained insight into how wishes and needs can be supported and how triggers can be applied to promote decision-making.

KNOWLEDGE BUILDING BLOCKS

Knowledge that helps to design adaptive, user-friendly applications for people with dementia that accomplish the following:

- persuade them to overcome impediments to talking about their illness;
- support them to enter the dialogue with their (in)formal caregivers;
- and assist in creating better insights into what support they need to uphold independent living.

PROTOTYPES

Five demonstrators:

- The design game Remembrance focused on jointly designing tools that can assist with retrieving memories, understanding each other, and supporting the formulation of help and care questions on the basis of these memories.
- The design game "Vertel eens," a quartet game, was developed to strengthen the relationship between the caregiver and the person with early dementia and to get conversations going.
- My Lable Care is an extension of the Lable client system for residential care centers. With this digital client system, it was already possible for the resident's family to read along in the client file, but during the project, an extension was made in co-creation.

This extension makes the system more interactive, whereby the end-user can fill in their own file and share it with their (in)formal caregivers.

- The Digi-Hug is a game that was designed in collaboration with the health-care organization to find ways in which the nursing home residents could get in touch with their loved ones and relive meaningful activities together. In this way, a number of concepts emerged in which message streams, photos, videos, and music became the carriers of connection in the form of digital hugs.
- Timesteps is a smartphone with a specially designed Android operating system in which widgets can be placed. The smartphone is part of an ecosystem in the house that supports prospective and retrospective memory.

GROWING ROOTS:CONNECTING ELDERLY THROUGHVIRTUAL NATURE SPACES

TOPIC & FOCUS

The project Growing Roots: Connecting Elderly Through Virtual Nature Spaces focused on the potential of using encounters with nature to address loneliness in elderly people. The research aim of this project was the development of an evidence-based framework for the creation of a virtual nature environment that inspires feelings of connectedness and stimulates social contact among the frail elderly.

"SEVERAL TIMES, I HAVE OBSERVED THAT SOMEONE ACTUALLY DID NOT REALLY WANT TO PARTICIPATE IN THE RESEARCH PROJECT. HOW-EVER, IF SOMEONE STILL PARTICIPATED AND WATCHED THE VIDEO, THEY HAD THE FEELING THAT THEY HAD BEEN OUTSIDE (THROUGH THE VIRTUAL NATURE IMAGES). SO JUST WATCHING THE VIDEO FOR FOUR MINUTES HAS IN MANY CASES REALLY MADE A DIFFERENCE IN THE STATE OF MIND AT THAT MOMENT. WE CAN'T SAY ANYTHING ABOUT THE LONG TERM, BUT ACTUALLY ALL INDIVIDUALS WE VE SEEN HAVE BEEN POSITIVE ABOUT THE EXPERIENCE."



The theoretical background of the project was shaped by Kaplan and Kaplan's (1989, 2011) Attention Restoration Theory (ART), emotion research (Frederickson, 2001; Piff et al., 2015), work on nature experience (e.g., Van Rompay & Jol, 2016), and insights into how the elderly adopt and use new technologies (Jaschinski & Ben Allouch, 2015a; Graaf et al., 2017; Lange et al., 2010). The frail elderly who live at home are in need of nature's benefits but are often the least likely to receive them. Nature heals and combats feelings of loneliness and disconnectedness (Kaplan & Kaplan, 2011; Ulrich, 1984). Earlier research has indicated that when people feel related to nature, they also experience a greater sense of connectedness to other

people and the world at large (Aron et al., 1992; Zelenski & Nisbet, 2014). New, innovative technologies provide opportunities for accessing nature in a social and engaging manner. Several studies have suggested that Virtual Reality (VR) technology is highly suitable for addressing the social challenges faced by the elderly (Baños et al., 2012; Farris et al., 1994; Goldstein et al., 1997; Jung et al., 2009). However, fundamental research indicating which specific natural features are essential for delivering nature's benefits is scarce. This realization inspired previous research (Van Rompay & Jol, 2016), the subsequent development and research of a virtual nature environment (Tech4People, 2016), and this project.

PROJECT ACTORS

The project management and core researchers of this project were a PhD and a researcher at a university and a researcher at a university of applied sciences. The design partner was involved in building and giving advice about the prototype. The health organizations and the target group were involved in data collection during the project.



The project started with developing a theoretical framework in which relevant research findings were documented. In parallel, the project extended on existing knowledge through analyses of existing design and VR interventions, and exploratory techniques to uncover which nature elements provide people with a sense of connectedness. The project addressed expectations and needs of elderly people towards VR and ambient technology during surveys, focus groups and interviews at the care center and at home. Based on the qualitative findings, the outcomes were reflected on potential VR design approaches with the creative industry partner after the project assessed suitability of the approaches in co-design sessions with elderly people supervised by the healthcare partners. The VR environment was tested in an experimental setting, in a living lab situation and during interviews at home.



Figure 2.15: Timeline


Figure 2.16 (Left): Virtual nature space in a room setting Figure 2.17 (Right): Virtual nature space in VR

RESULTS

KNOWLEDGE BUILDING BLOCKS

An evidence-based framework for the creation of a virtual natural environment that inspires feelings of connectedness and stimulates social contact among frail elderly, which consists of knowledge building blocks about the following:

- Experimental condition: different nature conditions related to feelings of social connectedness were identified. Both types of nature and spaciousness would be relevant factors in relation to social aspirations.
- Spaciousness: the spaciousness of nature scenes was manipulated; both dense and spacious scenes were developed. The results show that spacious rather than dense scenes elicited significantly higher social aspirations.
- Type of nature: the type of nature was manipulated; tended and wild nature scenes were developed. The findings show that tended nature scenes elicited more social aspirations than wild nature scenes.

PROTOTYPES

- The application is available via the website www.growingroots.nl. The digital nature experience made with the game engine Unity was exported as a desktop PC build (.exe).
- A demonstrator project is being considered, through which the consortium can further develop another prototype.

SQUEALED: SUPPORT QUALITY-CARE FOR ELDERLY USING AMBIENT LIVING ENVIRONMENT DATA

TOPIC & FOCUS

The SQUEALED project aimed to develop an eHealth tool to support safe, independent living of older people using energy consumption data. The eHealth application uses energy consumption data to determine the older person's daily routine. When something seems off in the daily routine, a message can be sent to either the older person or the primary care provider (e.g., family members or formal care providers). The project aimed to study whether the eHealth tool could support independent living at home and enhance older people's safety and quality of life.

Research findings indicated that identifying disruptions in the daily routine in a reliable manner was challenging.

This was mostly due to the low overall energy consumption and irregular daily routines which appeared from the data collection performed among independent-living older people. The project provided insight into energy consumption among independent-living older persons and the challenges of detecting disruptions in daily routines. As a stand-alone system, an eHealth tool based on energy consumption needs to be further developed before it is suitable as a product. However, it might be suitable for specific subgroups of older people and/or combined with other sensors like water and gas to monitor daily activity.

"WE DECIDED TO REPLACE THE PRODUCT WITH A QUALITATIVE STUDY, A VIGNETTE STUDY IN WHICH WE PRESENTED FICTITIOUS PEOPLE USING THE SYSTEM." Most older citizens prefer to live independently for as long as possible. However, direct care providers often have safety concerns, especially when signs of dementia appear. After time, these include risks of falls, self-neglect, and social isolation (Kelsey et al., 2010). The signs of early dementia appear in daily activities, such as turning on the television at night, leaving the tap water running, and getting up earlier every day. The SQUEALED project investigated the possibility of monitoring older people using energy consumption patterns (e.g., using the television, light, and coffee machine), as registered by the smart energy meter. The project aimed to develop, design, and evaluate an eHealth solution monitoring older people's energy consumption patterns.

PROJECT ACTORS

The project management and core researchers of this project were a PhD student and a health sciences researcher (assistant professor) at the public health department of a university. The researchers worked together with researchers from another university who brought in expertise in design, autonomous aging, electrical engineering, mathematics, and computer science. Moreover, two additional university/care partners were part of the advisory board (i.e., departments of geriatrics and neuropsychology). Various practice partners, such as health care professionals working with older people, informal care organizations, a housing and a care organization, technical and creative partners, and the local government were involved in SQUEALED to develop the product's back-end, recruit participants, provide a context for evaluation studies, and disseminate results. Figure 2.18 shows the actors mapped on the theoretical, conceptual, and real-life practice context.



TIMELINE

By means of a participatory approach, SQUEALED gathered data on how the proposed intervention could contribute to the safety and quality of life of independent-living older persons. The research method combined a literature study with a longitudinal field study that consisted of interviews at people's homes and field tests with a prototype.

Although the research initially consisted of sequentially scheduled activities, the research took place more in parallel. In addition, during the development phase, the focus was on technical development rather than the creation of design solutions. The recruitment of participants also took some time. Collaboration with the partners in care and community was needed to reach the right people and inform them about the project. Using the collected data from the field research (i.e., collection of energy consumption from independent-living older people), the analyses revealed that many false notifications would be sent to family and formal caregivers when the system was implemented. Therefore, researchers decided it would not be useful to further develop the eHealth system into a product. Instead, they focused on the technical development and analysis of the energy consumption data using qualitative research methods. Technical partners developed research reports with graphics and visuals that provided insight into daily energy consumption and energy consumption routines. The core researchers created vignettes that presented fictitious people using the system, based on the reports. Participants were asked to respond to these vignettes. This research step was instead of the controlled trial with a prototype of the eHealth tool.



Figure 2.19: Timeline



Figure 2.20: Ambient Living Environment Data (Technolution bv).

RESULTS

KNOWLEDGE BUILDING BLOCKS

 Understanding that monitoring energy consumption is not enough to fully support older people in safe, independent living. The concept might be useful in combination with other sensors, or, when further developed, only for a part of the population. Further research will have to be performed to study how more accurate monitoring can take place (e.g. using additional data) and for which groups of older people the monitoring system is the most valuable.

PROTOTYPE

- Report with historical data of total energy consumption (technical data report)
- Report with visualizations of energy consumption patterns (service designs)

PRACTICE

• There is a need among older people, informal caregivers, and care providers for a monitoring system that can increase the (perceived) safety of independent-living older people without interfering in their daily life. Detecting activity based on energy consumption is challenging when there is a lower activity level or when people have an irregular day-night rhythms.

UNOBTRUSIVE SENSING TECHNOLOGIES TO MONITOR AND COACH OLDER ADULTS WITH DEMENTIA: TRACK, TRACE & TRIGGER

TOPIC & FOCUS

The project Track, Trace, and Trigger (TTT) aimed to develop an unobtrusive Wi-Fi channel state information (CSI) system designed to accurately track and trace deviations in the relevant behavior of persons with dementia. The researchers explored the possibilities of a platform for caregivers to provide information regarding their daily functioning and behavioral and emotional state (Braakman-Jansen et al., 2017). The project explored how sensor technology can assess and monitor behavior and inform caregivers of people with dementia while preventing potential obtrusions or privacy breaches. The CeHRes Roadmap for eHealth, a holistic, participatory development approach combining value-based design with persuasive technology, led to this project to realize a suitable fit between technology, context, and stakeholders (Van Gemert-Pijnen et al., 2011). The first three phases of the CeHRes Roadmap were included in this project: contextual inquiry, value specification, and design (Kip & Van Gemert-Pijnen, 2018). The project's ultimate goal was to optimize the independent living of people with dementia by providing meaningful support to (in)formal caregivers.

"WE HAD A FANTASTIC START TO THE LOCKDOWN WITH THE TECHNI-CIANS OF A HEALTHCARE ORGANIZATION. THEY WERE RIGHT IN THE MIDDLE OF IT. THEY UNDERSTOOD OUR NEW MONITORING TECHNOL-OGY. THEY UNDERSTOOD WHAT WE WANTED TO DO WITH IT AND IT SUITED THEIR OWN NEEDS. IT WORKED WELL BECAUSE THEY HAD THE BUDGET TO BE ABLE TO TALK TO US FREELY AND THINK ALONG WITH US. WE THEN INVITED THEM TO COME AND VISIT US IN OUR TECH MED CENTER. THEY HELPED US SET UP THE MEASUREMENTS IN THAT LAB. IF IT WEREN T FOR THE LOCKDOWN, THEY WOULD HAVE HELPED WITH ON-SITE MEASUREMENTS AS WELL." Current monitoring technology studies focus on positioning technologies (domotica) that are static and not context-aware. They can only track how often a certain facility is used but not by whom or when. Therefore, these systems provide data that is difficult to interpret in view of supporting people's health and well-being. As such, these technologies generate many false alarms, which results in overcharging the professional and informal caregivers. Therefore, the focus of the TTT project is to develop novel unobtrusive sensory technology to identify an individual's daily dynamic activities (Kamminga et al., 2016a; Kamminga et al., 2016b; Le et al., 2013; Le et al., 2014; Le et al., 2016; Nguyen et al., 2017; Salomons et al., 2016; Van Kleunen et al., 2016). This way, the user does not have to wear anything, and their privacy is potentially better protected since it is not easy to retrieve user identification from the obtained data, such as the reflected radio waves.

PROJECT ACTORS

The project management and core researchers of this project were a PhD student and a researcher in the Section Psychology, Health & Technology of the Faculty of Behavioral, Management and Social Sciences at a university. The research was executed in collaboration with another research group of the same university (Engineering: Pervasive Systems) and researchers from other universities abroad. Three university partners were part of the advisory board. The practice partner was involved by giving advice and facilitating the data collection and development of the prototype. One of the healthcare organizations was involved by giving advice about the organizational and technical processes. The health organizations and the target group were involved for data collection during the project.



TIMELINE

The research approach is described as mixed-methods research and participatory development design approach, and, therefore, they used the CeHREs roadmap, combining qualitative methods (e.g., contextual inquiry and value-based requirements for design) and quantitative methods (e.g., deep learning using Wi-Fi and sensor data). Due to the user-oriented and iterative approach, stakeholders and a representation of future users were involved throughout the research. Interviews and focus groups were started with informal caregivers and care providers to map out needs and barriers with regard to monitoring. Focus groups with experts and stakeholders were organized to arrive at the values and attributes of TeleResearch at home. In addition, a survey with informal caregivers was conducted via the platform of the practice partner. The results of the qualitative and quantitative methods were translated into system and service requirements for the sensing system, as well as preconditions for its successful implementation from a multi-stakeholder perspective. In a parallel process, the system that analyzed motion data was assessed in experimental set-ups involving test subjects (lab setting). Machine-learning techniques were applied to improve the system's accuracy.



Figure 2.22: Timeline

RESULTS

Despite the presence of barriers, formal and informal caregivers of people with dementia generally saw value in unobtrusive in-home monitoring, and felt that these systems could contribute to a shift from reactive to more proactive and less obtrusive care. The requirements that emerged from the project can inform the development of more acceptable and goal-directed in-home monitoring technologies to support home-based dementia care.



Figure 2.23: Unobtrusive Monitoring to Support Home-Based Dementia Care

KNOWLEDGE BUILDING BLOCKS

• Requirements for unobtrusive remote monitoring technology which optimizes aging in place of people with dementia, reduces the burden of care, and provides meaningful support to both formal and informal caregivers.

PROTOTYPE

Algorithm Wi-Fi technology

PACO: PERSUASIVE DIGITAL AGENTS TO COACH OLDER ADULTS TOWARDS DIETARY BEHAVIOR CHANGE

TOPIC & FOCUS

The PACO project studied the use and usefulness of virtual eHealth agents to support healthy diets for older adults and decrease their loneliness. Such agents are computer-generated and animated conversation partners for technology users. The project identified the different factors that explain virtual agents' potential success for dietary change in older adults and developed and evaluated a web-based prototype based on these factors. In this prototype, two virtual agents engage in dialogue with older adults in order to motivate them to change their dietary behavior and decrease their loneliness.

"BASED ON THE LITERATURE, WE MADE A LIST OF THE THINGS WE AIMED TO LEARN IN THE CO-CREATION SESSIONS, SUCH AS PEOPLE'S VALUES, THE FOOD CHOICES THEY MAKE, AND WHY THEY SOMETIMES MAKE UNHEALTHY CHOICES. THE CREATIVE INDUSTRY PARTNER TRANSLATED THIS TO METHODS: HOW CAN WE GET ANSWERS ON THAT, AND HOW CAN WE BEST ADDRESS THIS IN A SESSION? ONE OF THE THEMES WAS THE PHYSICAL LIMITATIONS THAT ELDER-LY PEOPLE EXPERIENCE AS THEY PREPARE FOOD. THIS IS VERY PRAC-TICAL, SUCH AS LIFTING HEAVY PANS OF WATER FOR COOKING." As theoretical background, the project drew on concepts from persuasive health communication. As compliance with health advice is important for positive health outcomes, the successful design of persuasive virtual agents can have significant health benefits. However, insights into how to design such agents for the eHealth context are non-existent. Coaching and mentoring older adults via virtual agents is a radically new approach that may be far more motivating and persuasive than utilizing text-based information (as is currently the main approach). These agents allow for rich and persuasive automated communication. In theory, virtual agents should be able to deliver behavior change techniques such as self-monitoring, goal setting, action planning, and feedback, but research is lacking. Therefore, the PACO project built upon the fundamental research into virtual agents as done by Battagolino and Bickmore (2015) and Vardoulakis et al. (2012) and drew on the Elaboration Likelihood Model (ELM) (Petty & Cacioppo, 1986) to explore how individuals with lower motivation or ability to process arguments can be effectively reached.

PROJECT ACTORS

The project management and the lead researcher were based in a research group at a university. This group has expertise in intervention strategies to promote healthy behavior. A company and knowledge institute with expertise in design for the care sector was closely involved in user research activities. A research and development company in rehabilitation technology and telemedicine was closely involved in the knowledge development and the development and testing of the prototype of the virtual agent. The target group representation organization was involved to gain access to the target group and was also closely involved in the theory building. The target group of elderly people was involved in co-creation and prototype evaluation activities.



TIMELINE

The research design consisted of a series of studies: a study to build the general model of persuasion, a study with online experiments with different prototypes, and a real-life randomized control trial. To develop a general model of persuasion for virtual eHealth agents, the project combined the results of a scoping review with a co-design approach with older adults and their caretakers. The online experiments contained both qualitative experiments in group interviews and a quantitative survey. The persuasiveness of the (online) prototype of a virtual agent was studied in a randomized control trial.

The project started with an immersion in both theory (scoping review) and field (interviews, co-design sessions, and focus groups). The results were translated into the design of a set of three initial prototypes. Digital mock-ups were proposed to the target group in qualitative experiments through group interviews, and a large survey (N = 732), which was quantitatively analyzed. The results were used to develop design guidelines.

Based on these guidelines, a final integrated prototype was developed, which was run in a two-month field evaluation of elderly people using the virtual agent at their homes. This evaluation was an unblinded randomized control trial in two cohorts, with 30 participants per cohort.



Figure 2.25: Timeline

Figure 2.26: One of the virtual agents in the PACO web application: Herman, the cook, who gives nutritional advice.



RESULTS

KNOWLEDGE BUILDING BLOCKS

- A general model of persuasion for virtual eHealth agents (Kramer et al., 2020)
- Guidelines to design a virtual agent for coaching older adults on dietary behavior change (Ter Stal et al., 2020).

PROTOTYPES

- A web application of a virtual agent (figure 2.26, Kramer et al., 2021) will be further developed in a follow-up research project.
- The co-design development of this agent is described in Kramer et al. (2021b), and the randomized control trial is described in Kramer (2021a).

HEALTHY STORYTELLING FOR EHEALTH

TOPIC & FOCUS

The Healthy Storytelling project studied the potential of interactive storytelling to include people with low literacy skills in eHealth and to increase their health literacy regarding obesity prevention. The project investigated how communicating medical information through stories can improve the health literacy that is required to prevent obesity.

"WHEN STORYTELLING IS USED AS A DESIGN ELEMENT, STORIES HAVE A DOUBLE FUNCTION. THEY CAN BE COLLECTED TO GAIN INSIGHT INTO THE MOTIVATION OF USERS. BUT THEY CAN ALSO BE USED AS A MOTIVATIONAL ELEMENT, TO GET PEOPLE TALKING.

FOR INSTANCE, STIGMAS ARE A VERY SENSITIVE ISSUE RE-GARDING OVERWEIGHT. YOU CAN ASK SOMEONE: TELL ME WHEN YOU FIRST FELT STIGMATIZED? NOT EVERYONE WILL WANT THAT, BECAUSE IT MAKES YOU FEEL BAD. A STIGMA IS SOMETHING THAT MAKES YOU FEEL LESS. YOU CAN ALSO CHOOSE TO TELL A STORY: SOMEBODY ELSE TOLD US, THAT THEY FEEL VERY MUCH OBSERVED AND JUDGED WHEN THEY EAT A PIECE OF PIE AT A PARTY." The project drew on the literature about health literacy and storytelling. Obesity is not only related to an unhealthy lifestyle but to many other factors, such as genetic disposition, hormonal disturbances, and medication. These factors are not always known by people with obesity or by the general public. It is important to be able to talk about obesity, which is often not easy. Many people feel shame and do not want to talk about their overweight. Middle-aged people with low literacy skills, in particular, have a severe risk for obesity development at a later stage in life. They face difficulties in understanding and acting upon medical information and eHealth communication. However, when medical information is communicated by stories, their health literacy improves (Haigh & Hardy, 2011). Communicating health information by means of stories is well-suited to the interest and capabilities of people with low literacy skills (Horvath, 1987; Macaulay, 2002) and increases its memorability (Gray, 2009) and believability (Slater et al., 2003), both of which positively influence healthy behavior. However, design principles for effectively integrating storytelling interactively in eHealth are yet unknown.

PROJECT ACTORS

The project management and core researchers of this project were all part of a design research group at a university. Several other research groups were involved as knowledge partners and advisers to this core research team, advising from the perspectives of persuasive communication and storytelling, psychology and eHealth, and obesity. The creative industry partner was involved in the theory development throughout the project and in the final study developing the prototype. The health organizations and the target group representation organizations were involved in gaining access to the target group and giving expert advice, for instance, on obesity or low literacy. A municipality was involved in the second study ("Torries") to link to a regional health initiative (school project) and to give advice based on earlier health intervention projects. The target group was involved in co-creation activities in the three studies, which involved obesity patients and people with low literacy skills



Figure 2.27: Actor Map

TIMELINE

This project took a Research-through-Design approach and consisted of several studies in which prototypes were developed and evaluated to explore the solution space for Healthy Storytelling. Two activities were intertwined in these studies: 1) gathering story content about the personal experiences of people with low literacy skills and (ex-)obesity patients, as well as related medical information on obesity, and 2) developing interactive formats to represent this content as persuasive stories in a digital Healthy Storytelling space. Each iteration addressed a specific theme: taboos, stigmas and myths, and stimulating the conversation about obesity. The project started with an immersion in both theory (literature) and field (experts review). Based on this, both an initial theoretical model and an ini-

tial storytelling design method were developed. Two main studies followed in which design, prototyping, and evaluation took place. During these studies, the storytelling design method and model were further developed. The first study, "Bal & Spriet" ("Ball & Stalk"), was about stigmas and the ability to talk about overweight. A board game was developed during this study which involved people from a low-socioeconomic status (low-SES) group. This study led to a spinoff, a student project that created a website with conversation starters about overweight: "Dikke Onzin" ("Fat Nonsense"). The largest and final study, "Torries", ran at primary schools. The intervention involved a game in which both children and parents (via these children) are reached to increase their health literacy.







Figure 2.29 (Left): The conversation game Bal & Spriet (Ball & Stalk) during development with the target group. Figure 2.30 (Right): Children engaged in a co-design session to develop the "Torries" intervention.



Figure 2.31 (Left): The game Bal & Spriet (Ball & Stalk), which works as a conversation starter about weight and its associated stigmas. The game consists of a foldable gameboard and an app. Players engage with two main characters, Ball and Stalk, who run into stigmatizing weight-related situations.



Figure 2.32 (Right): The Torries plushies, which are part of an intervention in the form of a teaching package on healthy eating for primary schools. The Torries are introduced to the children as imaginary creatures arriving from a fictional island. The children need to take their Torries home. This way, the intervention involves both children and their parents

RESULTS

KNOWLEDGE BUILDING BLOCKS

- A theoretical model which describes why and how interactive storytelling can contribute to obesity-related preventive eHealth interventions. It describes which storytelling elements, such as fictive personas, discussions, and diaries, can be used and why (Vegt et al., expected).
- A design method is used that applies interactive storytelling while considering the various desires, motivations, capacities, and cultural backgrounds of people (Van Boeijen et al., expected).

PROTOTYPES

- Three demonstrators: design proposals that show how preventive eHealth based on storytelling can be included in a design (Van Eijk, 2019; Vegt et al., 2021. Vegt et al., expected; Visch et al., expected).
- Interest from the municipality to further develop the demonstrator in the third study ("Torries") and embed it into education programs.

PRACTICE

• The game that was developed in the second study ("Bal & Spriet") is produced in low quantities and is being used at community centers.

FOODSAMPLER: DIETARY REPORTING IN DAILY LIFE

TOPIC & FOCUS

The FoodSampler project aimed to understand the practices of dietary reporting in daily life, that is, how people report their eating during the day. Dietary reporting is often used when dieticians work with people who want to address their weight. They use various eHealth tools such as apps. These apps give insight into and subsequently change food consumption habits. To better design such eHealth reporting tools and make sure that they are used by people in their daily activities, it is crucial to understand these reporting practices. The FoodSampler project aimed to incorporate not only what is normally recorded – what is relevant for a dietician – but also what is relevant for a patient or participant to report on, as well as the reporting qualities regarding patient privacy or data ownership. Relevant insights were captured through the knowledge building blocks that can be used to develop eHealth monitoring tools.

"WE DO NOT LOOK AT HOW A MONITORING TOOL IS GOING TO MAKE PEOPLE LOSE OR GAIN WEIGHT. WE JUST WANT TO MAKE THEM USE IT, AS WE UNDERSTAND THAT THIS DATA NEEDS TO BE GATHERED FROM THEM. WE WANT THEM TO REPORT NOT ONLY WHAT NORMAL-LY HAS BEEN RECORDED, BUT ALSO THE REASONS WHY. HEALTHCARE PROFESSIONALS OR DIETICIANS ALWAYS NOD AND SAY: "YEAH, THAT IS IT. IT IS ABOUT GETTING TO THE WHY." AND THEY SAY THEY DO TRY BUT THEY DON'T GET THE DOOR OPEN TO GET THERE. WE LOOK AT PARTICULAR QUALITIES OF REPORTING. SUCH AS: WHERE WOULD YOU LIKE TO REPORT IN YOUR HOME CONTEXT? WHEN YOU'RE OPENING YOUR FRIDGE, WHEN YOU'RE SITTING AT THE DIN-NER TABLE, WHEN YOU'RE LYING ON YOUR COUCH?" To study and understand monitoring food intake practices, the project drew on concepts from several areas. First, this project took the angle of "positive health" (Huber et al., 2016), addressing the shift to support the active involvement of people in their own health conditions. This is opposed to the majority of existing scientific and commercial food measurement systems, which focus on treatment rather than prevention and describe rather than explain overweight. Second, it drew on persuasive and user engagement theories (e.g., O'Brien & Toms, 2008). Third, it used in-situ reporting methods to investigate the ease-of-use and direct benefit of in-situ tools that integrate different techniques for collecting (Hektner et al., 2007) and reconstructing (Kahneman et al., 2004). Finally, it drew on concepts from mixed-method research (Creswell & Piano, 2011; Romero Herrera, 2017) about using a mixed set of data sources and collection techniques.

PROJECT ACTORS

The project management was part of a design research group at a university. This group has expertise in self-management, user engagement, self-reporting, and monitoring. Researchers from this group worked closely together with a research group at a university of applied sciences that has expertise in food measurements for overweight interventions. The creative industry partner was involved with the theory development throughout the project and in the conceptualization and development of the prototypes. Target group representation organizations and several expert organizations were involved in gaining access to the target group and giving expert advice, for instance, on obesity or dietary practice. The target group was involved in co-creation or activity evaluation in all studies. This involved obesity patients and people with low literacy skills. Sometimes in the practice context (at home), sometimes in the conceptual context (in focus groups), and sometimes in the theoretical context (in the lab at the university).



TIMELINE

Central to the project's way of working was the evaluation of interventions or prototypes in context through the so-called Living Lab design research approach (Krogstie, 2012, Romero Herrera, 2017). The project employed various methods to capture knowledge from, with, and by users regarding the day-to-day practices and experiences of a phenomena (in this case: the self-reporting dietary behavior).

End-users and professionals were explicitly involved as providers and users of information through the user-centric design approach. The project takes a research-through-design approach through which design artifacts (existing products as well as newly developed low and hi-fidelity prototypes) were iteratively developed and deployed in context. The project started with an immersion in both theory (literature) and field (interviews). This was followed by an exploratory phase: three studies in which design, prototyping, and evaluation took place. These studies partly informed each other and partly ran parallel. Based on the results, a final prototype is currently under development. This will run in a short evaluation in the home context (still in progress at the time of writing this publication). All data was collected in the end-users' context, interpreted, and evaluated with and by end-users (parents of overweight children and older adults at different stages of overweight) and food experts (dietitians and scientists).





Figure 2.35 (Left): A research participant engaged in a contextual interview.

Figure 2.36 (Right): A research participant engaged in the home lab setting at TU Delft to explore the qualities of interaction.

RESULTS

KNOWLEDGE BUILDING BLOCKS

 A conceptual model is developed which identifies the contextual aspects that influence dietary behavior and the specific needs of people with overweight and obesity regarding personal data dietary behavior (van Oers et al., 2019, Romero Herrera et al., 2018). Best practices are developed of contextual reporting (what to report when, where and how).

PROTOTYPES

• A demonstrator of novel engaging interactions is under development. The theoretical insights from FoodSampler will be applied in real-life tools for dieticians in follow-up projects to develop an enhanced proof-of-concept digital food coach.

PRACTICE

• The project reframed how the involved dieticians and eHealth developers see the future of dietary tools: minimizing people's reporting effort by adding advanced sensing technologies and increasing engagement through gaming and social techniques.

GOAL: GAMIFICATION FOR OVERWEIGHT PREVENTION AND ACTIVE LIFESTYLE

TOPIC & FOCUS

The GOAL-project aimed to gain un understanding of intervention strategies for overweight prevention in the aging population. Therefore, GOAL conducted two case studies and a series of gamification experiments with end-users of a mobile health (mHealth) app to measure engagement (e.g., the number of days end-users visited a mobile health (mHealth) app or the number of health behaviors that end-users registered in the app). With the experiments, the research team investigated the impact of four intervention strategies: How 1) reward mechanisms, 2) social comparison, 3) adaptive goal setting, and 4) personality tailoring can be employed to foster engagement with an mHealth app. Seven experiments

took place with (pre)adolescents with low socioeconomic status (SES) in educational contexts and one experiment with sedentary office workers in occupational contexts. The experiments lasted between four weeks and twenty weeks each between 2018–2021. Relevant insights were captured through knowledge building blocks that were used to propose a toolbox, "SciModeler," that estimates the potential impact of intervention strategies in a given context (i.e., based on existing empirical data). GOAL demonstrated that SciModeler could be used effectively (i.e., by querying its central database) to explore promising intervention strategies for a specific context.

"WE HAVE COLLECTED A LOT OF EMPIRICAL DATA AND INFORMATION ABOUT HEALTHY BEHAVIOR CHANGE IN PRACTICE, IN THE WILD, WITH REAL PEOPLE AND WITH APPS." Many eHealth tools have been developed to support and promote an active, healthy lifestyle. Popular health apps can, for example, count steps and track workouts. Engagement levels with mHealth apps typically collapse after short periods of time, and the effectiveness of an mHealth app largely depends on the specific combination of intervention strategies the app employs (Nuijten, 2022). The GOAL project investigated levels of engagement of various intervention strategies regarding overweight prevention through the use of gamification techniques, which refers to game strategies outside of a game context (Deterding et al., 2011). Important factors for effective gamification that lead to healthier lifestyles are the theoretical underpinning and personalization (Cugelman et al., 2013; DeSmet et al., 2014). However, many existing gamified and eHealth approaches have limited theoretical foundations (Tabak et al., 2015). GOAL addressed the theoretical gap between health behavior changes, smart, healthy living environments, and health game design.

PROJECT ACTORS

The project management and core researchers of this project were a PhD student and an information systems researcher (associate professor) in Industrial Engineering and Innovation Sciences at a university. The researchers worked together with design honors students from the Industrial Design department at the same university and a researcher in digital (game) technology for healthy urban living from another university. Additionally, there were four other university partners (two full professors and two associate professors) who were on the advisory board.

Many practice partners were involved in GOAL as sparring partners, including well-being organizations and end-user organizations like many municipalities and various secondary schools. The end-user organizations provided the context for the experimental studies and participant recruitment. Figure 2.37 shows the actors involved in GOAL, mapped according to the theoretical, conceptual, and real-life practice contexts.



GOAL researchers experimented with an existing mHealth platform called GameBus, which they incrementally improved by adding creative ideas and evaluating the ideas in practice. Central to the project's way of working was experimenting in practice with different health game configurations in the health app. The configurations were creative ideas as possible solutions for the four intervention strategies that were added to the health platform. For example, they experimented with the intervention strategy "reward mechanisms" by which badges could be earned and points collected that are then converted into gift cards to investigate the impact of the various ideas. Through the app, they monitored the progress of end-users regarding specific tasks, such as "go for a walk" or "eat an apple." They visualized the performance and engagement of the end-users when using different health game configurations; for example, are they more engaged when collecting badges or saving money for a gift card?



Figure 2.38: Timeline



Figure 2.39 (Left): Research participants engaged with the mHealth app. Figure 2.40 (Right): Research participants engaged in healthy tasks.



Figure 2.41: Research participants engaged in an experiment using the mHealth app.

RESULTS

KNOWLEDGE BUILDING BLOCKS

Empirical data on behavior change intervention strategies in health games were gathered during experiments in occupational and educational contexts. This resulted, among others, in the following knowledge building blocks:

- Insights into how gamification prevents overweight and supports an active lifestyle (Nuijten et al., 2018).
- A theoretical meta-model for consolidating scientific knowledge by linking empirical data with theoretical constructs (Nuijten & Van Gorp, 2021).
- Assessment of the influence of physical activity on the experiences with sampling response devices (Khanshan et al., 2021).
- Various evaluation studies.

PROTOTYPES

• A database toolbox for consolidating scientific knowledge on intervention strategies for obesity prevention in specific contexts has been created. The theoretical insights from GOAL have been applied to this real-life toolbox, "SciModeler," for researchers in follow-up projects to estimate the potential impact of intervention strategies in a given context (i.e., based on existing empirical data).

WAYS TO INNOVATE AND DO RESEARCH

MARIEKE ZIELHUIS

"I WAS HAPPY TO SEE 'RESEARCH THROUGH DESIGN' AS SUG-GESTED APPROACH WITHIN THIS PROGRAM. TO SEE WHAT IT BRINGS WHEN YOU USE THIS APPROACH, BUT ALSO TO DEVEL-OP MORE GUIDANCE ON HOW TO DO THIS EFFECTIVELY."

-DESIGN RESEARCHER

3.1 INTRODUCTION

In research projects on eHealth, such as those in this book, several practices and worlds come together. All have their own beliefs and ways to go about their work. In this chapter, we will see how this works out in their joint methods.

The two worlds that meet are those of research and innovation. They find each other in the shared goal to eventually improve peoples' lives. Innovation is the successful implementation of creative ideas within an organization (Amabile et al., 2016). The journey to this goal is a long-term effort, and research can contribute to a part of this journey (Moors, 2013). Combining these worlds can pose some problems as they essentially differ in the nature of their processes. Even though many companies try to organize innovation in clear-cut stage-gate models, it is hard to control and predict in practice (e.g. Van de Ven et al., 1999). Most research, conversely, aims for structure and control.

A strand of research that combines these two worlds is the research that originates in the field of design. Stappers and Van der Lugt (2006) illustrate how design became recognized as a driving factor in innovation and as a knowledge-growing activity. This resulted in design research as a branch of academic research. The role of design is central to this research. Such research comes in many flavors and goes by different names.

For some researchers, design and research are separate. In *design science research*, they are organized as separate activities which take turns over time. In this type of research, the focus of the research activities lies on validating solutions (Hevner, 2007; Peffers et al., 2008; Van Aken, 2005). The design process itself is a black box that is not part of the

research process. In other types of research, the two are intertwined. This is particularly the case in *Research through Design* (RtD) (Zimmerman et al., 2010; Stappers & Giaccardi, 2017). In this type of research, design and prototyping are seen as crucial to knowledge production (Durrant et al., 2017). In other words, the design process is a knowledge-generating activity in itself (Stappers & Van der Lugt, 2006; Stappers & Giaccardi, 2017; Zimmerman et al., 2007). This intertwinement of design and research is operationalized in many different ways.

As we see in these 10 eHealth projects, the world of research also meets the world of practice in various ways. Issues from real life often stand at the base of the research, and practical impact is part of the goals. In many cases, design research projects are aimed at societal issues (e.g. Roggema, 2017; Zimmerman et al., 2010). We also see practice partners join into the project. Even ways of working that are derived from practice find their way into the projects, for instance when methods from design practice are adopted (Wensveen & Matthews, 2015).

In addition, various challenges can arise when one attempts to integrate ways of working from the disciplines of healthcare and design. In particular, Blandford and colleagues (2018) pointed out several differences and related challenges in human-computer interaction. Healthcare research is typified by a focus on evaluation through randomized controlled trials and effect studies. However, this is not the only way of working in healthcare research. The paramedical professions, such as nursing, have a tradition of grounded research and other qualitative research approaches, which are also often pursued in design research. The difference is that in qualitative research in nursing science is conducted in a more structured and protocolized manner than is common in the field of design (e.g. McCann & Clark, 2003). In the design field, evaluation has a formative function. This provides insights on usability, usefulness and user experience. Differences can also be found in the approaches towards identifying user needs and developing interventions. Groeneveld and colleagues (2019) and Van der Lugt and Van der Laan (2017) describe how these differences also create challenges in collaborations with health practitioners. Collaboration between these domains can also be problematic because the design field lacks a clearly described and agreed upon set of methods. This makes it difficult to explain approaches to design research and the corresponding methods to non-design partners. For instance, the term 'co-design' is interpreted in many ways within the field of design (Koskela-Huotari, 2013; Mattelamki & Sleeswijk Visser, 2011; Sanders, 2008). If there is no consensus within the design field, it is even harder for partners outside of that field to understand what a certain method entails. The health research tradition, conversely, has clear ideas and method descriptions. As a result, the authors of design research papers typically need to dedicate many words to descriptions of method, whereas naming the specific methods that are used tends to suffice in healthcare papers (Blandford et al., 2018).

In short, in eHealth research collaborations, several practices, with their own beliefs, values and ways of working, come together. *Innovation* meets *research* in various design research approaches. The explorative world of *design research* meets the world of *health research*, with its randomized control trials and effect studies. And finally, ways of working from *research* are combined with ways of working from *practice*. This chapter investigates what happens when these worlds meet and the ways of working that are employed as a result. It answers the following question: *how do the ways of working in eHealth research projects add value to the quality of the process*?

We observed a variety of characteristic ways of working, which we organized in the following themes (Figure 3.1):

- 1) the goals of a project (Section 3.3) and, in light of those goals
- 2) the mix of methods (Section 3.4)
- 3) the role of prototypes (Section 3.5)
- 4) iterations and the research path over time (Section 3.6)

The four sections (Sections 3.3-3.6) present the emergent ways of working that we found within each theme. They add value to the projects, and they highlight points at which the different worlds, namely those of design and health and of research and practice, reinforce each other.



Figure 3.1: Four themes on ways of working in research-practice projects in which the design domain and the health domain meet.

In the following paragraphs, we will discuss each theme. First, we will explore the four themes from a theoretical perspective (Section 3.2). That exploration is followed by a discussion of observations from practice in relation to each of the four themes. In that section, we start with the end of the process, that is, with project goals, in mind. The main tips that we provide, which are based on the insights in this chapter, are summarized in Figure 3.16 in Section 3.7.

3.2 THEORETICAL PERSPECTIVE ON THE FOUR THEMES

PROJECT GOALS

The goals of a Create-Health research-practice collaboration that is directed towards an (eventually) meaningful eHealth innovation can be manifold. The Research Pathway Model (RPM) that we introduced in Chapter 1 highlights that projects can deliver outputs in three contexts: the theoretical context (papers, models and theory), the conceptual context (ideas, prototypes and such like), and in real-life practice (changes to practices). This entire range of outcomes could be beneficial for the practice situation in the long run: others may build on theoretical contributions to devise new solutions, and prototypes can be developed into actual products or services.

When researchers and practice partners collaborate, they can be expected to have different priorities. As Stokes (2017) argued, there need not be a trade-off between an 'eye for generalization' and an 'eye for application' in research. In fact, many design research projects aim for both. Beck and Stolterman (2016) argue that design publications often contain multiple knowledge claims of different kinds. Publications in the natural and social sciences, conversely, tend to contain singular knowledge claims of similar kinds. RtD projects, in particular, are often intended to instigate societal change (Brankaert & den Ouden, 2017). In doing this, they typically produce a wide range of outcomes. This range of outcomes also encompasses so-called 'intermediate knowledge,' such as guidelines and demonstrators (Gaver & Bowers, 2012; Hoök & Löwgren, 2012; Löwgren, 2013). Sleeswijk Visser (2018) shows how a single project can produce outcomes on all these levels.

The health and design domains also differ in the nature of the knowledge that they aim to generate. Healthcare researchers try to find evidence of effects, such as changes in behavior. The dominant approach in health is evidence-based practice or evidence-based medicine (e.g. Burns et al., 2015; Portney & Watkins 2000), in which different types of evidence are ranked. This approach has become prevalent in evaluations of non-pharmacological interventions, but it has also been criticized (e.g. Howick et al., 2009). Design research, conversely, concerns, among others, the quality of interactions (Blandford et al., 2018). Essentially, health research is oriented towards what we know now, and design research is oriented towards exploring possible futures. In the words of Koskinen and colleagues (Koskinen et al., 2011), 'designers focus on the creation of artifacts through a process of disciplined imagination.' By doing so, they open the solution space.

This also implies a difference in research objects. Design research typically produces knowledge not only about phenomena or solutions but also about approaches to design. The various audiences of these projects have different interests in them. All three types of knowledge, about phenomena, solutions, and approaches, are of interest to design professionals (Zielhuis et al., in press).

THE MIX OF METHODS

For design-health collaborations, there is not much guidance on how to deal with these differences in orientation, and to operationalize this in method use. Current guidance focuses on highlighting the differences between disciplines and the challenges of working with those differences. Little guidance is provided, especially for user research and intervention development. One of the main challenges that result from the different orientations is that disciplines have different views on what an adequate basis for moving forward is and on the manner in which research findings should be grounded.

Blandford and colleagues (2018) provide some helpful directions. As far as the overall approach is concerned, they suggest that creatives and health researchers find each other through a person-centered approach. In the healthcare domain, the importance of considering individual situations and needs has been emphasized (e.g. Council for Public Health and Society, 2017). Personal approaches also have advocates in the field of design (e.g. Smeenk et al., 2022). Blandford and colleagues (2018) suggest that those who evaluate interventions should make a distinction between immediately measurable effects, such as user actions and perceptions, and more distal outcomes, such as longer-term effects on health.

Methods from different disciplines can be mixed in many ways. In some cases, one of the disciplines essentially takes the lead. At the other extreme, disciplines really integrate their ways of working. Choi and Pak (2006) distinguish between three main ways of working with multiple disciplines. They describe multidisciplinarity as a salad bowl in which the different ingredients are clearly distinguishable. Activities stay within the boundaries of the field. Interdisciplinarity is described as a melting pot or a stew in which the ingredients are only partly distinguishable. Links are formed between the disciplines to form a coordinated and coherent whole. Finally, transdisciplinarity is presented as a cake in which the ingredients form a final product of a different kind. This approach integrates various disciplines and transcends traditional boundaries. McComb and Jablokow (2022) argue that all degrees of disciplinarity can be valuable.

THE ROLE OF PROTOTYPES

The prototypes that play parts in these projects should be assessed in the light of the goals

of the projects. This can be difficult when objectives are manifold and viewed differently by partners (as discussed previously). By 'prototypes,' we mean the created objects (often tangible) that are used in research and can realize the (inter)action that is studied (Stappers & Giaccardi, 2017). Prototypes (both physical and digital) play an important role in many design research projects, and they can have different functions (Stappers & Giaccardi, 2017; Wensveen & Matthews, 2015). These are sometimes considered to be important end goals of a project—a prototype that can be developed further into a product. This said, in many cases, these prototypes are a means of inquiry or exploration (e.g. Keller, 2005). The artifacts that are created function as embodiments of possible futures (Koskinen et al., 2011). For practice partners, the role of prototypes in a particular project may not always be clear. In some cases, the *process* of prototyping is a vehicle for inquiry (Wensveen & Matthews, 2015). For instance, it can provide insights about the design approach. Horst (2011) describes the process of crafting prototypes as 'platforms for participation' in iterative design. Reay and colleagues (2017) illustrate how prototyping can facilitate collaboration between designers and clinicians (and support the boundary-crossing process—see Chapter 5).

ITERATIONS AND THE RESEARCH PATH OVER TIME

Bringing ways of working from different disciplines together involves activities that extend beyond the level of method mixing. There is also the matter of deciding how to move forward and how to ground research findings. This problem is about building a research path over time.

Iterating is a characteristic way of moving forward in much of design research. Researchers who work in a design research context are accustomed to design and research being interwoven, which makes for an agile and dynamic process. However, this process can take many different forms, especially when design is closely tied to the research process. This is particularly the case in RtD. Stappers and Giaccardi (2017) distinguish various approaches to RtD projects that all exhibit different ways of iterating. These differences are related to their different goal orientation. Some projects iterate on successive prototypes, which can be left behind when moving towards a final product. Other projects take a programmatic approach in which a conceptual frame is filled with explorations in design. These design explorations are also seen as iterations. The research process in design science, conversely, is characterized by cycles of observation and analysis that revolve around a specific design proposal—developing and building, using, testing and evaluating, and receiving and integrating feedback. Since all of these iterations have different goals, the criteria for moving forward are also different.

In the health disciplines, experiences of and views on working with iterations differ (Mann et al., 2018). Blandford and colleagues (2018) describe how many health researchers see

the development of complex (non-pharmacological) interventions as a sequential process that culminates in a randomized controlled trial that determines effectiveness. This view has gradually been broadened as more researchers have seen the necessity of iterative development.

Unfortunately, there is not much guidance on adopting an iterative approach when working across boundaries and towards health interventions. Hermsen and colleagues (2020) show how an agile approach leads to rapid progress and successful stakeholder inclusion when a multidisciplinary team is developing health interventions. However, they note that the integration of user research and scientific evidence in the development process is still a challenge as speed seems to be favored over rigor.

3.3 PROJECT 60ALS

"WE TRY TO FIND A BALANCE BETWEEN PRACTICAL APPLICATION AND FUNDAMENTAL RESEARCH. THIS COMBINATION IS WHAT STANDS OUT IN THIS PROGRAM AND WHAT REALLY ADDS VALUE."

-RESEARCHER, GROWING ROOTS

"WE REFINED AND VALIDATED A THEORETICAL FRAMEWORK AND AD-DRESSED THE PRACTICAL APPLICATION. HOW TO CONNECT THE CREATIVE INDUSTRY AND HEALTHCARE? THAT IS WHY WE FOCUSED ON DEVELOPING DESIGN GAMES."

-DESIGN RESEARCHER, NATALIE

Although project partners are united by an interest in new knowledge, we found that they actually have different interests, which can be hard to combine. We found several ways in which projects manage to overcome this difficulty.

FOCUS ON THEORY AND RELEVANCE TO PRACTICE

The call of ZonMw provides an opportunity to focus on theory development. Since previous programs apparently focused too much on delivering solutions for practice and, in the words of the program manager, 'skipped over the knowledge production,' projects are sought that can produce knowledge building blocks in order to contribute to solutions to problems in healthcare practice. All design and development activities should be aimed towards knowledge production.

Many of the researchers who were involved welcomed this focus on theory development. They explained that this is not always the case in health-design eHealth collaborations. Instead, the focus is often on the development of a specific product, such as an app. They sometimes feel the need to take a step back and to develop the knowledge that is needed before going into product development. For instance, the project *Track*, *Trace*, *Trigger* deals with technological explorations and ethical issues—which aspects of the lifestyle, health and safety of individuals with dementia can be measured through the use of innovative and unobtrusive technology, and how does the target group perceive and accept them? The researchers see developing a product proposal as premature in this early stage of the innovation path. One of the researchers explained that they were glad that the call did not require 'jumping to solutions.'

Although the Create-Health program was not intended to result in actual product development or implementation, most researchers aim to achieve a combination of generic ('fundamental') knowledge and relevance to practice. All projects were conducted in close cooperation with practice partners, who brought their perspectives from practice into the project. In addition, most researchers felt driven to be relevant to practice and to contribute to it. In these projects, which interact with practice, they were often able to achieve these results. An example from the *GOAL* project follows (see Box 3.1).

This example shows how a researcher can combine a focus on theory development with a direct impact on practice. A focus on theory is also not problematic for most of the practice partners who are involved. Design practitioners and health practitioners recognize the need for an evidence base in healthcare innovations. For instance, the design practice partner in the project *Growing Roots* observed, '*This was a chance to develop an evidence-based mechanic that we can seamlessly implement in other games and VR environments that we make* [...] *That is the first thing healthcare clients ask for: evidence based.*' The design practice partner in *FoodSampler* accepted this focus on theory development and expected practical value in the long run: 'I see this more as building up knowledge for follow-up projects than that it helps us now.' They were explicitly involved in this knowledge-development process within the project. Care organizations and care professionals also understand the need for an evidence base in healthcare innovation. Moreover, most of them had worked with the researchers in the past and understood the theoretical focus of the collaboration.

DIRECT IMPACT IN A THEORETICAL PROJECT

Project: GOAL

Although the project aims at the development of a theoretical model and tools for researchers, it is very important for the main researcher that the project and the research impact the lives of individuals positively. This means that they also value a direct impact on practice. While they were working on the two main outcomes of the project, this impact took shape in various ways.

One of the two main outcomes is empirical knowledge about behavior-change intervention strategies in health games. This knowledge was gathered from experiments in occupational and educational contexts. Several experiments with health games were ran in a practice context. They had different configurations. For instance, subjects could earn badges as rewards or collect points that could be converted into gift cards. The researcher made the following observation:



Figure 3.2: one of the experiments with health games ran at schools

For the past four years, my personal goal has been to restore the connection between living healthy and feeling energetic. I have tried to inspire people to live healthier lives and feel energized through a mobile health app. I feel that I have made some valuable contributions to this challenge. I am very proud to have reached over 800 individuals with my digital lifestyle programs throughout this period. My programs have encouraged some of these individuals to engage in a healthier lifestyle.

The other main outcome of this project is a

toolbox, SciModeler. This toolbox is a metamodel and database that helps researchers or health innovators to estimate the potential impact of intervention strategies in a given context (i.e., based on existing empirical data). This can be used to develop and evaluate personalized eHealth interventions to prevent obesity. By providing this toolbox, the researchers hope to affect the lives of even more individuals.

A VARIETY OF GOALS AND OUTCOMES

The new knowledge that was developed is not presented solely in the form of models or theory. The projects aimed for a broad range of outputs, which varies, first of all, in terms of

70

content. We see that many projects produce knowledge about three topics: about phenomena, about design approaches or about solutions. Knowledge about a phenomenon yields insights on issues such as loneliness or persuasion. Knowledge about design approaches, for instance, may result in the development of guidelines on design for individuals with dementia or the use of storytelling as a design element (*Healthy Storytelling*). Finally, knowledge about a solution refers to descriptions of a particular solution but also to the requirements or components of new solutions or even a broader solution space for future projects.

These various types of knowledge are disseminated in a variety of formats. Apart from theoretical contributions (such as models that are described in papers) and prototypes for eHealth solutions, the projects yielded various types of in-between outputs, such as design guidelines. A researcher from the *PACO* project explained that the project was geared not only towards developing a virtual agent but also towards creating and sharing the design guidelines on which that agent is based: 'It is most important to share the knowledge and experience that we developed during the project. In our case, about designing eHealth applications, and virtual agent as such.' Such guidelines are often shared through academic papers as well as at events and workshops that are aimed at an audience of design practitioners. The *Growing Roots* project developed content for a VR environment that can be reused in other VR solutions for older adults. This is an example of an 'intermediate' result. The researchers on the *NATALIE* project developed their knowledge of design approaches into design games that other designers can use (see Box 3.2).

A DESIGN GAME THAT MAKES THEORY ACTIONABLE

BOX 3.2

Project: NATALIE

The project entails the development of knowledge about the experience of autonomy in communication and decision-making. The researchers have developed a set of design games to use these insights in practice, for audiences of both creatives and health practitioners. How to maintain balance in autonomy?

With the design games, we aim to make this larger framework actionable: how can you use this in practice? There are a lot of aspects that you need to keep in mind when working with products or client systems that aim to facilitate the communication. How can you provide an overview of all the aspects for someone from the creative industry or the caregiver or the people whom it concerns? The design game guides you along all these conceptual aspects and makes clear how to recognize the principles in products. So, the game helps you apply the theoretical framework in an analytic and a designerly manner. The various project partners sometimes exhibited different interests in these various knowledge outcomes. This can complicate matters. These different interests sometimes lead to partners having different goals in certain steps of the research. This tendency is particularly pronounced in the steps of analysis. For instance, in the *PACO* project, the design practice partner was interested in the values of older adults that are relevant to food. This would be helpful in similar design projects. The university partner was more focused on the theoretical model of a phenomenon (in this case, changing behavior). As a result, the design practice partner focused on the analysis of the optimal translation of these values, whereas the university partner focused on identifying the consequences for the theoretical model of the phenomenon of eating behavior.

OUTCOMES FOR DIFFERENT AUDIENCES

These various types of knowledge outcomes are also informative for different audiences in different ways. Since most projects aim to impact multiple audiences, it is hard to combine their different interests. In these projects, the researchers aim to inform the fields of design and healthcare research as well as the practice of design and healthcare. In particular, the design-practice audience seems to occupy a secondary position, behind the other audiences. We identified several challenges that stand in the way of informing the practice of design effectively. These are described in more detail in Zielhuis et al. (2022). For instance, differences between design research and design practice are sometimes not recognized, which makes it difficult to address the needs of the latter. This problem will be discussed in greater detail in Chapter 5. In additions, researchers on some projects refrain from reaching out to a design-practice audience until after the end of a project, whereas practitioners prefer to be involved at a much earlier stage. We also found that the involvement of a design practice partner in a project does not necessarily mean that the partner in question will assume the role in order to represent the interests of design practice as a whole.
CONCLUSIONS: PROJECT GOALS

We found that a focus on theory can be combined with relevance for practice in projects that are conducted in close cooperation with practice partners. Partners join these projects because they have some interest in knowledge development. However, the domains of research and practice and of design and health aim at different types of knowledge. These differences are on two dimensions: the *topic* and the *orientation* of the desired knowledge. If project partners are clear about those dimensions from the outset, they have the opportunity to discuss them early.

The first dimension is *topic*—what is it about? Projects can produce knowledge about three general topics: a phenomenon, design approaches or solutions. For health researchers, the approach of design-science research, in particular, is close to home, in that it focuses on developing knowledge about specific solutions. When a specific solution is part of the work of the design-practice partners (when it is their design), this can also be of interest to them.

Knowledge with a different *orientation* can be produced on each of these topics, which results in the second dimension on which the domains differ. Some researchers are more intent on developing knowledge about what we know now, and others are oriented towards the future. Design researchers in the tradition of RtD, in particular, develop knowledge on all three topics but mostly strive to explore possible futures. While they develop insights about phenomena and design approaches, they also produce information about a range of interventions, which create a broader solution space for future projects. They often develop knowledge about several solutions in this process.

Finally, we found how the different interests of audiences can lead to similar difficulties. In order to inform a broad audience, researchers need to consider the needs and interests of different groups. Combining these interests can be challenging. Researchers can draw on the examples of projects, such as that of Sleeswijk Visser (2017), that distinguish project outcomes on various levels. They can also use many examples of intermediate knowledge, such as design games or guidelines, to create knowledge with generative qualities.

Table 3.1 summarizes the main insights, the ways of working that we found in the projects, and the relevant literature. In the next section, we will explore the manner in which researchers employ mixed methods to work towards these goals.

TOPIC	INSIGHT	LITERATURE	EMERGENT PRACTICE FROM THE PROJECTS OR LITERATURE
THE GOALS			
Focus on theory	Focus on theory is relevant to practi- ce. However, the domains of research and practice and of design and health differ in the type of knowledge that they aim for.	Evidence-based practice (Burns et al., 2015; Portney & Watkins 2000); differences between healthcare and HCI (Blandford et al., 2018); future orientati- on of design (Koski- nen et al., 2011)	 Be clear about the type of topic of the intended output: knowledge about phenomena (e.g. dementia), about design approaches (e.g. using storytelling), about solutions (e.g. a specific eHealth application). Often, projects aim for a combination of the above but have different emphases. Be clear about orientation: now oriented (what do we know about this intervention now) or future oriented (how does this open the solution space).
Various audiences	In much of design research, particularly RtD, knowledge is typically captured by a range of outco- mes, from theory through guidelines to technology de- monstrators and pro- totypes. It is difficult to ensure that these outcomes serve the different interests of the various audien- ces.	Range of outcomes in RtD projects (Sleeswijk Visser, 2018); inter- mediate knowledge (Gaver & Bowers, 2012; Hoök & Löw- gren, 2012; Löwgren, 2013); serving design practice (Zielhuis et al., 2022)	 Targeting different audiences through specific outcomes. Consider the range of available options: models, design guideli- nes, design requirements, design games, demonstrators, etc. "We view the design guidelines as the main outcome." "With design games, we aim to make this larger framework action- able."

 Ces.
 Image: Ces.

 Table 3.1: Insights on the ways of working that deal with goals in eHealth projects. Emergent practices from the projects and references to the relevant literature are added.

3.4 MIXING METHODS THROUGHOUT THE PROJECT

"THIS SUBSIDY SAW THE VALUE OF GENERATIVE AND EXPLORATIVE METHODS."

-END REPORT, FOODSAMPLER

"HOW DO YOU PUT EVERYDAY PEOPLE ACTUALLY IN THE LEAD OF A DESIGN PROCESS? YOU HARDLY SEE THAT IN RESEARCH PROJECTS THAT WORK WITH CONCRETE SOLUTIONS. OUR PROCESS PROVIDES AN ANSWER FOR THIS."

-DESIGN RESEARCHER, EVERYDAY SOUNDS OF DEMENTIA

In order to work towards the goals described in the previous section, the projects combined methods from different disciplines in various ways. The researchers used four main approaches to address the differences between now-oriented and future-oriented perspectives. To map these approaches, we use the RPM, which distinguishes between various research activities and contexts.

In all projects, activities take place in all three contexts that the RPM distinguishes (see Figure 3.3): the theoretical, the conceptual and the real-life-practice context. The colored dots in the timelines in Chapter 2 are related to these various contexts (blue = theory, orange = conceptual and red = practice). Researchers place much emphasis on activities in the practice context, that is, on field research. We also see a variety of activities in the conceptual context. In terms of the RPM, we see this both in activities to 'understand and create' and in activities to 'explore and test.' Activity in the theoretical context is largely focused on 'understand and create.' A few studies involved lab experiments, as in the *FoodSampler* and *Track, Trace, Trigger* projects. The following depiction of the RPM grid indicates that a variety of different types of activities are placed in each square.



Figure 3.3: Various activities in the top two rows of the RPM matrix ('Understand & Create' and 'Explore & Test').

This variety of activities can be organized in line with their orientation, be it future or present. The future orientation is particularly pronounced in the various generative methods that are used to help individuals to express their needs and desires. Co-constructing stories is an example of the generative method. It is central to the Healthy Storytelling project, but it was also used in FoodSampler: 'We incorporated this in our interviews, and it was the most successful method we had. It is a way to be able to ask about intimate things without asking about themselves but about someone else. It did not take more than two minutes before people started talking about themselves.' Introducing a prototype to practice is also a good way of gaining information on user needs. In Box 3.3, which follows, we illustrate this proposition by reference to the Everyday Sounds of Dementia project, in which various prototypes were used to elicit responses that are hard to obtain through inquiry.

GENERATIVE WAYS OF DOING USER RESEARCH

Project: Everyday Sounds of Dementia

The final Tumbler study concerns a co-design process that involves three couples. The researcher works with various generative methods.



Figure 3.4: the Soundbox that was personalized in the co-design sessions in the Tumbler study.

With each couple, we did an activity in which we personalized the soundscapes. In each session, we built further on their reactions. We began with a very generic soundscape. This elicited all sorts of personal reactions, with people talking about their holidays in Spain at the sea. We personalized the soundscape by adding Spanish sounds. We did three sessions like this. To do each session, we developed a sort of sound box with switches that we could personalize.

In Figure 3.5, we added this 'perspective' to the RPM as an additional layer. The two layers show the difference between a future-oriented and explorative perspective (the top layer) and a now-oriented and pre-structured approach (bottom layer). We found four main strategies for addressing these different perspectives. They are described in the following sections.



Figure 3.5: 'Perspective' as an additional layer of the RPM. Various strategies were found to address these different perspectives in the categories 'Create & Understand' and 'Explore & Test.'

A SEQUENTIAL APPROACH

The first approach is to work consecutively: first do one thing, then another. In 'explore and test,' the strategy is often to perform the more qualitative evaluation first and to conduct an effect study later. Most researchers structure their evaluations so as to obtain indications of effect, usability or use experience. In Box 3.4, we illustrate this proposition by reference to the *Healthy Storytelling* project. The design partners are experienced in evaluating prototypes that are still in development qualitatively, whereas the health partners are experienced in evaluating the effects of developed interventions. Combining this phase of designing and developing an intervention with an effect evaluation within the scope of these projects turns out to be challenging. A conclusive effect study is often planned as a follow-up. In effect studies and in many qualitative studies, most projects aim at evaluations in real-life contexts. An evaluation in a real-life context is still planned for the *FoodSampler* project. The researchers conducted an evaluation with professional experts within the projects, using concrete prototypes.

QUALITATIVE EVALUATING ON THREE LEVELS

BOX 3.4

Project: Healthy Storytelling

In the second study of this project, the researchers designed an intervention for primary school that is aimed at healthy eating. It takes the form of a teaching package. Stuffed toys called Torries are central elements of this package. The Torries are introduced to the



children as having arrived from an imaginary island. The children need to take care of their Torries at home. With this intervention, the researchers aim to reach not only children but also their parents.

They evaluated the intervention on three levels: 'We looked at the usability of the intervention ("Is it doable for teachers?"), at the experience for the

Figure 3.6: the stuffed toys named 'Torries' as part of an intervention in primary schools towards healthy eating.

children ("How was their involvement?") and, finally, at the effect ("Is there a learning outcome or a behavioral change?").' The researchers reported, 'Evaluating generated knowledge with [dieticians] by means of design artifacts proved to be successful in discussing more deeply the pros and cons of possible concepts. This allows them to be more concrete with regards to their expectations and concerns of the concepts.'

The researchers on the *PACO* project merged the strategies for integrating results and took a sequential approach. The project began with a co-design approach to developing a virtual agent, which was evaluated in a randomized control study to determine its effect. This evaluation was inconclusive, but it did provide some indications of the effects and pointed to relevant directions for further development and evaluation. In fact, the researchers combined different perspectives in the evaluation study of the virtual agent that was developed during the project. They used different methods. Each took a different perspective to evaluating the agent (see Box 3.5).

DIFFERENT PERSPECTIVES ON EVALUATING A VIRTUAL AGENT Project: PACO

BOX 3.5



Figure 3.7: one of the virtual agents in the developed app: Herman, the cook.

In order to evaluate the prototype of the virtual agent that was developed during the project, the researchers combined questionnaires with qualitative interviews. The evaluation focused on the qualities of the interaction as well as on indications of effect.

We studied the use: could we explain why people use the app or why they don't, and how this is related to the use experience, how they though privacy was handled, how the agent looked and so on. We also wanted to know whether they felt less lonely or had started to eat more healthy. We did not want a questionnaire of 20 pages, which

is very usual in the food field. This limited us in what we could ask. The interviews provided much more insight, especially about food and loneliness. People made very specific changes: they started to cook with fresh ingredients. Some described the use of the app as a wake-up call. (Researcher, PACO)

In this way, the two research methods had different foci. The interview focused more on the qualities of the interaction, and the questionnaire focused more on indications of effect.

In this approach, the quality of the handover is a challenge. Each party needs to understand the input of the other. A particular challenge is to transfer insights so as to inform designers. This is especially difficult when there is a clear cut between phases, such as a handover of design requirements to other project partners or even beyond the project. Usually, such requirements or guidelines evolve gradually during a project. A warm handover to those responsible for design and development is helpful.

INTEGRATING RESULTS

The second strategy is to integrate the results of different activities into joint results. We find this in 'understand and create.' Several projects show that grounding a design process in previously developed models and theories while keeping an open mind and learning from users is challenging. One of those projects is PACO. A researcher explained, 'It is a tension. Co-creation is a nice technique to connect to the everyday reality of your target group, but it still gives a fairly limited representation of this group at a particular moment, while, at the other hand, you have scientific literature about behavior, about behavior change, about what is and isn't effective. This is sometimes not in line with what you find in co-creation.' A researcher on the Track, Trace, Trigger project described a similar tension: 'There is not really a satisfactory way to draw up requirements, especially requirements that are derived from both user research and literature.' The researchers involved project partners with different backgrounds, especially ones that possess technical expertise, in the project in order to integrate these different types of input into the requirements.

One of the ways in which results can be integrated is illustrated by the *DDD* project. In this project, the social researchers aimed to conduct longitudinal research with a pre-structured focus. They combined this with a context-mapping approach that the design researchers introduced. In the studies, this resulted in cross-fertilization between different orientations. Box 3.6 describes how the results were integrated in various steps, a 'ping-pong' approach.

A PING-PONG APPROACH TO LONGITUDINAL RESEARCH Project: DDD

The project aimed to capture changes in technology use among individuals with dementia in relation to their social participation. During the project, the focus came to lie more on mapping the (desired) social networks of individuals in the early stages of dementia. The focus on technology became less prominent.

In this research, the social researchers aimed to conduct longitudinal studies and to capture

changes in this use of technology over time. Researchers returned to ask the same questions over a period of time. To attain rich insights and to explore design opportunities, the design researchers proposed to conduct more explorative context-mapping activities. Combining the orientation towards the present of the first approach with the future orientation of the second posed a challenge. They combined the two by adopting an approach in which they alternated between context-mapping activities and more traditional interviews. This resulted in a 'ping-pong' approach, in which the focus of the co-design studies was determined by the preceding interview. Both activities contributed to a shared set of insights,

which developed gradually during the process. These insights were one-on-one insights about several individuals.

In Godfroij et al. (2022), the researchers elaborate on the process of creating shared understanding through this approach. For instance, they write on the role of design tools and tangible objects.



Figure 3.8. In the project, persona creation was shaped by both the longitudinal interviews and the context-mapping activities.

INTEGRATING METHODS

The third approach is to integrate different ways of working into a combined method. We find this in 'understand and create' and in 'explore and test.' In particular, user research is almost never a unidisciplinary effort by one of the partners. Often, we see health actors and creatives combine ways of working. In projects that are not led by creative partners, they are brought in to design but also to conduct user research and to report on the perspective of users, that is, to capture rich insights on what individuals find important. The health partners provide access to the target group and domain expertise. What do we know about a certain target group or health domain? What are the sensitivities, and what works best? In the FoodSampler project, the dietary expertise of one university and the contextual-inquiry expertise of another were used to set up and execute interviews jointly. This required some effort to be expended in order to find a method that both parties would be comfortable with: "Their understanding of running interviews was very different from ours. For them, an interview was more a matter of closed questions, very structured, so they could compare the different participants. We work with contextual interviews, cameras and pencils so they

can map things. It took a bit of time to find our combination, but I think it was one of the nicest studies we did, in terms of the data it generated. We had to learn from each other's method and to come to a method that we were both comfortable with to run. The main issue was not that they did not believe in what we were doing, but [that] they did not feel comfortable running something like that" (Researcher). The researchers also conducted joint analysis sessions, in which they also had to integrate different data analysis practices.

A PERSONAL APPROACH

The last approach can be used in various research stages. Common ground between the health and design domains is found by using a person-centered approach. Such an approach of N=1 (or N=low) is helpful in several projects. This approach can be helpful when it is difficult to access a target group (e.g. individuals with dementia) and when it is necessary to build a more durable relationship with participants. In the *NATALIE* project, the first iteration in one of the design cases was even conducted with a single older adult.

VITA CUSHION: DIRECT INTERACTION BETWEEN RESEARCHER AND PARTICIPANTS USING A PROTOTYPE

Project: Everyday Sounds of Dementia

In the second study, the researchers wanted to explore the context of the care homes in which individuals with dementia live. They used the prototype VITA cushion to achieve this goal. The researcher explains how they focused on working directly with the individuals with dementia:



Figure 3.9: a research participant in interaction with the VITA cushion

Many evaluation studies are conducted by proxy: the caregivers use a prototype for a certain period and report on the use. This is biased because the success of the product is perceived from the perspective of the caregivers or within a care organization.

ROX 3.7

I entered a care home as a design researcher and was able to offer the VITA cushion directly to the person with dementia. I sat with this person and observed how they interacted with the cushion. This way, you find out things that you don't discover when you take only the perspec-

tive of the caregiver. Of course, we also add the other perspective. Beforehand, we did a workshop with caregivers about their views on the prototype, and we did exit interviews with them afterwards. In *Everyday Sounds of Dementia*, the researchers also opted for a personal approach in all three studies. Box 3.7 describes how this worked out in the second study. The researcher explained how some insights about effect indications, about usability and about use experience can only come to light when researcher and participants interact directly with a prototype. In this way, the researcher can observe developments that a care professional would fail to notice or ones that a participant would not report. Building a relationship right from the start can be very useful when researchers need to work closely with the same individuals over a long period of time, as in longitudinal interviews or in co-design trajectories. The final study from *Everyday Sounds of Dementia* shows how a person-centered approach can be used throughout all phases and with a small set of participants. In that study, the researchers worked with three couples. Chapter 4 zooms in on various service-based ways of working and added value for research participants.

This personal approach also helps in framing the research for various domains. Publishing design research can be challenging, especially when researchers wish to reach an audience beyond the design field, for instance by publishing in health journals. The co-design approach can be particularly difficult to frame. A researcher on Everyday Sounds of Dementia noted, 'It is a challenge to let design research land in other domains. We describe a process from co-design to prototype to evaluation. Usually, we get the feedback that each of these steps is not thorough enough. Medical journals do not even see this as proper research! In the human-computer-interaction field, you see a lot of qualitative, often ethnographic studies, and many evaluation studies. Studies that take you along in such a co-creation process are not easy to find.' Emphasizing the person-centered approach helps to provide a frame because it is recognized in different domains.

CONCLUSIONS: MIXING METHODS

We find four successful ways of dealing with the differences between the various practices from now-oriented and future-oriented perspectives. We find them in several activities through the projects. In these combinations, appropriate use is made of the strengths of different areas of expertise, such as those of health and design and of research and practice. The four strategies are as follows: a sequential approach (first one, then the other), the integration of results, the integration of methods and a personal approach. The first can be recognized as a *multidisciplinary* approach, as described by Choi and Pak (2006). The second can be recognized as *interdisciplinary*, and the third as *transdisciplinary*. The personal approach can be applied to all levels of disciplinarity. The examples in this study show how different levels of disciplinarity can be found within a single project.

The sequential approach is mostly found in evaluations. They are mostly conducted sequentially, in line with the work of Blandford and colleagues (2018). Evaluations that are aimed at gaining indications of effect, usability and use experience provide rich information at an early stage. For a full-blown effect study, the prototype needs to be in a more final form and not in the development phase. In some cases, a combination is found.

The integration of shared results is found in user research and design activities. The combination of different types of evidence is a prominent challenge. Along similar lines, Austin and colleagues (2020) suggest four strategies for combining evidence-based and experience-based inputs. When the requirements for a specific design appear to be conflicting, the options are to satisfy one need but not the other (selecting), to retain multiple options in the design (combining), to design a new and coherent functionality that serves both needs (integrating) and to redefine perspectives (reframing).

We found examples of the integration of methods and a *transdisciplinary* approach in every project phase. In several projects, researchers combined methods to conduct user research and to analyze findings together. Finding a method that both parties are comfortable with requires some effort.

Overall, taking a personal approach is a recurring strategy in which design and health researchers find each other. The N=1 (or N=low) approach is not very popular in design research, but it was found to be helpful in several projects. This approach also resonates with the values of each domain. Smeenk and colleagues (2022) provide a tool for operationalizing this perspective, the empathy compass. Insights about working with a small number of participants in times of social distancing (due to Covid) are described in a white paper by Godfroij et al. (2020).

Finally, it is important that research partners discuss the various methods, because methods, particularly in the design domain, are not clearly defined (Sanders, 2008). Such discussions help to understand what the methods entail better as well as to identify means of combining them.

Table 3.2 summarizes the main insights, the ways to mix methods that we found in the projects and the relevant literature. The following section focuses on the role of prototypes within these methods.

MIXING METHODS					
TOPIC	INSIGHT	LITERATURE	BEST PRACTICE FROM THE PROJECTS		
Mixing methods in all phases	Mixing methods is mixing per- spectives	Differences between hu- man-computer interaction and healthcare (Blandford et al., 2018); interpretations of methods (Sanders, 2008); future orientation of design (Koskinen et al., 2011); levels of working with multiple disci- plines (Choi & Pak, 2006).	• Discuss the difference bet- ween a future-oriented and a now-oriented perspective to facilitate discussion about stra- tegies for addressing them.		
Grounding in both evi- dence and experience	Grounding in both literature and user rese- arch is a challen- ge and requires the active parti- cipation of both disciplines.	Challenges in identifying user needs and development (Groeneveld et al., 2019; Van der Lugt & Van der Laan, 2017); combination strate- gies: selecting, combining, integrating or reframing the bottom-up and top-down evi- dence (Austin et al., 2020).	 Interdisciplinary integration of results, for instance through a ping-pong approach (example of DDD). Transdisciplinary integration of methods, for instance by preparing and conducting interviews, co-design sessions and analysis sessions jointly. 		
Different forms of evaluation	Health research aims to produce effect studies; design research is aimed at effect indications and insights on usability and use experience.	Differences between hu- man-computer interaction and healthcare (Blandford et al., 2018).	 A multidisciplinary and sequential approach (preferably with a warm handover): first qualitative evaluations of indications of effect, usability, use experience and RCT as follow-up effect study. Transdisciplinary integration of methods: combining a study of effects and of quality of interaction. 		
Personal approach	Some insights can only come to light when researcher and participants engage in direct and personal interactions.	Empathy compass (Smeenk et al, 2022).	• A personal approach when re- searchers need to work closely with the same individuals over a longer period of time, as in longitudinal interviews or in co-design trajectories.		

Table 3.2: Insights on ways to mix methods in eHealth projects. Emergent practices from the projects and references to the relevant literature are added.

3.5 PROTOTYPES WITH VARIOUS FUNCTIONS

"FOR US, PROTOTYPES ARE FOREMOST RESEARCH MEANS. WE CALL THEM ARTIFACTS, PROBES OR PROTOTYPES..."

-RESEARCHER, DDD

The previous sections have already shown that artifacts and prototypes play a central part in these projects. This section zooms in on this variety of prototypes. Three different types, which have different purposes, can be distinguished: products as solutions, research artifacts and proofs of concept. We found that each type requires a different way of working.

DIFFERENT PURPOSES OF PROTOTYPES

When it comes to the results of some projects, the eventual development of a specific product or technology as a solution to a practice problem is of primary importance. The virtual agent in the PACO project provides one example, and the VR environment in the Growing Roots project provides another. These types of prototypes are meant to be delivered as products that will eventually produce a solution to the practice problem. The idea is that these products, being solutions, will eventually be developed into functioning and market-ready products. Development into market-ready projects takes time. Researchers on the Everyday Sounds of Dementia project described the VITA cushion as a previously developed research product, and they explained how it is different from an actual commercial product: 'There is no company, no production chain, no service model. But we have used it now for over two years.' The actual market rollout of a final prototype is certainly the goal of the long-term innovation path, but it is often not attained within the timeframe of a research project. Sometimes, achieving the outcome in question requires several consecutive projects. In the projects that are presented in this book, the focus is on thorough theory development to ground the final product or technology. For instance, the key question in the PACO project is whether the digital virtual agent leads to behavioral change.

In other projects, prototypes have the primarily goal of enabling a phenomenon to be studied and are used to trigger responses and interactions. We call these research artifacts. The Healthy Storytelling project and Everyday Sounds of Dementia supply salient examples of such research artifacts. In these projects, these artifacts are used to study the respective

phenomena of the role of storytelling in discussions of obesity, and the relationship between sound and wellbeing. Other examples are the GameBus app in *GOAL* and some of the prototypes in *FoodSampler*. In the latter project, several research artifacts were intended as means of gathering knowledge about a specific aspect of an interaction through the use of a monitoring device, which is described in Box 3.8. The resulting design proposals often incorporate knowledge that is hard to convey exclusively through words. That is why in some cases, these prototypes are used as *demonstrators* to communicate with an audience that is outside of the project.

EXPERIENTIAL PROTOTYPES

Project: FoodSampler



Figure 3.10: one of the research participants in interaction with an experiential prototype in a lab setting.

An initial framework was developed early in the project. This framework included the contextual factors that influence dietary behavior and the specific needs of overweight and obese individuals regarding the way they report about dietary behavior.

ROX 3.8

Several prototypes were developed in one of the explorative studies. The prototypes all address very specific aspects of the framework. The researcher explained, *'Two*

prototypes explore the concept of engagement, another one the concept of efficiency, another does a bit of both. So, we're understanding different aspects of this idea of efficiency and engagement.'

The prototypes were described as experiential prototypes. The term 'experiential' means that they are mock-ups, developed purely to study an interaction. The researchers use them in lab studies and as video demonstrators, in combination with narratives, to provide context.

This approach also determined the scope of the analysis of the results from the user research. This research was not aimed at identifying requirements for a specific product but at finding out which themes were relevant to the exploration: 'The main goal of this analysis was to extract the themes we wanted to bring into the design interventions.' To complicate matters, some prototypes start out as *research artifacts* but end up as *products as solutions.* Some of them receive such a warm welcome by practitioners that they decide to develop them further as products. The Torries from *Healthy Storytelling* were intended as research artifacts to be used at primary schools within the context of the study. The municipality expressed considerable interest in them. As a result, they may be developed further as a tool to reach parents and children and inform them about the topic of healthy eating. Of course, researchers are pleased by this interest, but they have limited opportunities to engage in further development. Living Moments from *DDD* is another example of a research artifact that might become a product.

In some cases, prototypes are meant to demonstrate the feasibility of certain technologies or methods or of the opportunities that they create. These are indicated to be proofs of concept or technology sketches. The prototype Tumbler in Everyday Sounds of Dementia is an example of this tendency. The prototype demonstrates the opportunities for everyday sounds and design in home environments. One of the researchers explains, 'I would call it a technology sketch. It is a first manifestation of the insights that we gained during an intensive co-design process.' In this case, the insights concern the ways in which social activities can be used to explore selfhood and identity. In the project Track, Trace, Trigger, a technology sketch was developed to visualize the concept of novel and unobtrusive athome monitoring technology. This project investigates the possibilities of using unobtrusive monitoring to support older adults who live independently as well as their caregivers. In this early stage of the innovation path, developing a product proposal would be immature. One of the researchers said that they were glad that the call did not require 'jumping to solutions' and that it provided an opportunity for thorough investigation. The project followed two parallel paths. One explored the aspects of daily life that can be measured accurately by using unobtrusive tracing technology; the other inquired what potential users need and how they would perceive and accept the technology.

The technology sketch serves as an input for studies of the target group. However, it was not meant as a solution because the insights from the user research had not yet been integrated.

DIFFERENT PROTOTYPE FUNCTIONS REQUIRE DIFFERENT

WAYS OF WORKING

Developing eHealth solutions often requires work with digital prototypes. It takes expertise and effort to develop such prototypes, especially eHealth prototypes, which require much technology in forms such as apps and platforms. Projects often combine the expertise of researchers and practice partners to facilitate prototyping. We find different ways of working to be appropriate for different types of prototypes: three ways of working are appropriate for developing *research artifacts*, and one way of working is appropriate for developing prototypes into *products as solutions*.

First, researchers can develop new, quick-and-dirty and low-fi prototypes as research artifacts. These allow for easy, quick and local adaptations. Such adaptations can work well when researchers need to study an aspect of a particular phenomenon, as in the *FoodSampler* lab studies that involved experiential prototypes.

Researchers can also develop high-fi working prototypes as research artifacts. This development can produce better insights, especially when research is conducted in the real-life-practice context in order to understand the broader setting of an experience. The prototype Living Moments in *DDD* is an example. A design researcher states, '*Quick-anddirty prototyping is often used in co-design, while a working prototype provides more in-depth insights in the experiences at the time of use.*' A partner with extensive design expertise is needed to develop such prototypes. This individual should be either a design researcher or a practice partner from industry. In the *Healthy Storytelling* project, the design-practice partner took an active role in the development of the Torries as part of the second main study. Professional design agencies provide the expertise that is necessary to develop a solution that is genuinely useful and practically feasible.

Using existing prototypes or products (digital or physical) can save effort and time in design. The VITA cushion that was used in *Everyday Sounds of Dementia* (Figure 3.9) provides an example. In the *GOAL* project, an existing eHealth platform, GameBus, was used as a basis. The researcher explained, 'I am quite happy that we did that, that we did not start from scratch. We kept adding creative ideas to this platform and evaluating these. That was the continuous cycle: learning from this, which triggers new ideas, implementing these in the platform, see what it does. Does it work, how does it help? Learn and adapt.' Using existing products can also prove difficult. In the NATALIE project, the aim was to conduct research with an existing eHealth product. As in the *FoodSampler* project, using an existing platform made it difficult to introduce the quick adaptations and explorations that were intended. A researcher on the NATALIE project explained, 'We thought that we would be able to

prototype in a very agile and flexible way with an existing product, but this turned out to be harder than we thought. This is really a clash with how creative industry works, especially with things that are up and running. That is why we used more rudimentary and easily changeable prototypes in the end.' Students and design researchers developed these early and explorative low-tech iterations of *FoodSampler*.

In some cases, researchers work with a practice partner from industry that has a vested interest in the eventual product. In the Growing Roots project, the design-practice partner (a game developer) was involved in the continuous adaptation and finetuning of the VR environment throughout the project (Box 3.9). As illustrated by this example, such partners do not necessarily implement the particular prototype in question as a product, but they can use its elements in similar products. The PACO project followed a similar approach, in that an R&D company was involved in the long-term development of the virtual agent. The expertise and the facilities of these professional technology developers contributed to a digital platform that provided a stable basis for the research. In Everyday Sounds of Dementia, the researchers sought the light involvement of various partners from the design industry who had similar products. They hosted a few sessions with these partners, and their advice helped to improve the prototype that was used within the research project. At the same time, the sessions were used to share insights from the project with these practice partners.

ADAPTING AND FINETUNING THE VR ENVIRONMENT

Project: Growing Roots

This project focused on the creation of a virtual nature environment. The researchers had the goal to develop an evidence-based environment, which inspires feelings of connectedness and stimulates social contact amongst frail elderly.

A game development company was involved in the development of this VR environment. To fine-tune to the specific context, and to the specific needs Figure 3.11: research participants with the virtual nature environment.

of older adults, this environment had to facilitate dynamic adjustments so as to allow for quick iterations. The professional game developer was involved throughout the project in a continuous adapting and finetuning of the VR environment, based on the input from the researchers.



The design professional was especially interested in the focus of this project on building an evidence-base. In his practice of developing (serious) games within healthcare, he finds that is very important: "That is the first thing healthcare clients ask for: evidence-base". He explains how he will not necessarily implement this particular VR environment as a product, but he can use elements of the developed content: "This was a chance to develop an evidence-based mechanic that we can seamlessly implement in other games and VR environments that we make". This is also of interest for a broader design practice audience.

CONCLUSIONS: THE ROLE OF PROTOTYPES

Working with prototypes is characteristic of the ways of working that predominate in the creative domain. Making tangible things can facilitate shared understanding considerably, but it can also lead to confusion. Partners from other domains are familiar with prototypes as first versions of eventual products. However, prototypes have other functions, such as the elicitation of responses and interactions. We showed how a single project can involve prototypes with different purposes and how these different purposes require different ways of working.

It is important to discuss the purpose of the prototypes in the project, especially when working with high-fi prototypes and practice partners. Will a care institution end up with a working solution for their local situation? Can the high-fidelity prototype that their clients love stay with them? For funding parties, it is important to recognize the need to work with an existing platform or product and to facilitate involvement.

Table 3.3 summarizes the main insights, the ways to deal with the prototypes that we identified in the projects and the relevant literature. The next section focuses on the final theme, namely the manner in which projects are iterated.

THE ROLE OF PROTOTYPES					
TOPIC	INSIGHT	LITERATURE	PRACTICE FROM THE PROJECTS		
Prototype functions	 Prototypes can have different functions, which require different ways of working. Differentiate between: Product as a solution: to develop (eventually) into functioning and market-ready products; Research artifact: to study a phenomenon and to trigger responses and interactions <i>a prototype to explore the concept of engagement, another one the concept of efficiency";</i> Technology sketch: to demonstrate the feasibility of certain technologies or methods and the opportunities that they create <i>"It is a first manifestation of the insights that we gained".</i> 	 Prototypes in RtD (Stappers & Giaccardi, 2017); prototy- ping as a means of inquiry (Wensveen & Matthews, 2015); proto- typing to cross boundaries (Reay et al., 2017). 	 To study a phenomenon with a research artifact: Develop new, quick-and-dirty and low-fi prototypes "we used more rudimentary and easily changeable prototypes in the end" Develop a high-fi working prototype provides more in-depth insights in the experiences at the time of use" Use existing prototypes or products if they allow for quick adaptations "We kept adding creative ideas to this platform and evaluate these" To develop successive and incremental versions of an eventual product as solution: Work with partner from industry with a vested interest in the product/platform 		

Table 3.3: Insights on ways of dealing with the role of prototypes in eHealth projects. Emergent practices from the projects and references to the relevant literature are added.

3.6 ITERATIONS AND THE RESEARCH PATH OVER TIME

"THE STRENGTH OF DESIGNERS IS THAT THEY WORK ITERATIVELY: LEARNING FASTER BY MAKING MISTAKES FASTER. THIS WAY, YOU PROCEED MUCH FASTER."

-DESIGN RESEARCHER

The previous sections described how the project partners work towards the various goals, how they mix methods along the way and how they use various types of prototypes. The path that they take while doing this always exhibits some features of iterative processes. In Chapter 2, these research paths were depicted in a simplified manner and as fairly linear. As we zoom in on them in this section, the different types of iteration become clearer. We show how these differences are related to the different types of evidence that are sought and to the way in which activities move between the three contexts.

DIFFERENT GOALS REQUIRE DIFFERENT ITERATIVE PATHS

In the projects, we found two main ways of iterating. The first is the iterative development of a specific eHealth innovation that is evidence based. The second is the iterative process of filling a conceptual framework through design explorations.

Several projects focused on the development of a specific eHealth innovation. A single design case was central throughout those projects. *Growing Roots*, in which a VR environment for older adults was developed, provides an example, as does the *PACO* project, which aimed to produce a virtual dietary behavior coach for older adults. The iterative process in *PACO* is outlined in Figure 3.12. The prototype that was developed during the project can be seen as a *product as a solution* (see Section 3.5): a prototype that is created with an eye on eventual development into a specific product or technology. In particular, this approach is used in projects that are led by researchers from healthcare or the behavioral sciences who aim to develop evidence-based interventions. Integrating the insights from user research as well as from scientific evidence into the development process (see Section 3.3) is a matter of particular concern in projects that pursue such goals. This problem is addressed through the iterative process.



Figure 3.12: The iterative pathway of the PACO project, in which a virtual agent was developed.

In other projects, an iterative approach is used not only at the product level but also at the theory level. Design cases function as iterations or loops through which a theoretical framework is gradually developed. In the *Healthy Storytelling* project, the design iterations are intended to explore a storytelling approach to design (see Figure 3.13). Both of the two main design cases are iterations that develop this framework. The prototypes in these studies are primarily research artifacts (see Section 3.5) that purport to trigger responses and interactions. Since both were received favorably in the practice context, they might also evolve into actual products. The spin-off project *Dikke Onzin* ('Fat Nonsense') was directed more towards a product as a solution, in the form of a website for overweight individuals. Each design case also entails the use of an iterative process to arrive at prototypes. This type of iterative approach is also used in several other projects that are led by design researchers. These projects are particularly unpredictable and difficult to plan. A researcher explains, 'It's an iterative process in which ends and means alternate. So, you need to check every couple of months. Where am I? Can I go to the next phase? This is more difficult to manage than a tight evaluation study.'



Figure 3.13: The iterative pathway of the Healthy Storytelling project, in which a storytelling design model was developed.

MOVING BETWEEN THREE CONTEXTS

These two different ways of iterating also entail different ways of moving between the three contexts of theory, concepts and real-life practice. We illustrate this by using the 3x3 matrix of the RPM. The first type of iteration, which is directed towards evidence-based innovation, is depicted on the left-hand side of Figure 3.14. The iteration is aimed at developing a prototype that is grounded in both theory and practice. This prototype is tested thoroughly, preferably in a real-life-practice context. The iterations loosely follow a pattern whereby the movement is from theory through concepts to practice. The other type of iteration starts with a theoretical framework. Explorations can take place in various contexts and always lead to conclusions that are related to the framework. This pattern alternates between various contexts but always loops back to the theoretical one.



Figure 3.14: Two ways of iterating and moving between contexts. Left: iterations that are aimed at developing a testable prototype, which is evaluated favorably in practice context but sometimes in a conceptual context. Right: iterations that start from a theoretical framework, whereby explorations from various contexts produce conclusions that are relevant to the framework.

The timelines in Chapter 2 also show that, as far as these contexts are concerned, every study has a different orientation. The representations of some studies are replete with yellow dots (the conceptual context), whereas others contain more red dots (real-life practice).

Sometimes, researchers can start engaging with real-life practice early. The project NA-TALIE, in which researchers worked with existing client-communication software, provides an example. The Everyday Sounds of Dementia project supplies an illustration of the different orientations of the three studies (see Box 3.10). In each study, choices were made for various reasons. Such choices depend on goals as well as on opportunities.

STUDYING SOUNDS AND DEMENTIA IN VARIOUS CONTEXTS

Project: Everyday Sounds of Dementia

The project is oriented towards developing knowledge about the relationship between sounds and dementia. The project consists of three studies that have different orientations.

- 1. The Soundboard was developed to study a theoretical concept and concerns the theoretical and conceptual contexts. The Soundboard was used as research artifact, in focus groups and in workshops.
- 2. The second study has a sharper focus on practice. The researchers used the VITA cushion in the care-home context. The co-design sessions that explored how the cushion could be used in that context took place in a conceptual setting.
- 3. The third study, *Tumbler*, focuses on opening a solution space, that is, on exploring how solutions can be developed in close contact with individuals who have dementia. To this end, the research took place in a real-life-practice context, that is, in the homes of individuals. In this case, both development and testing took place in a real-life practice context.



Figure 3.15: Timeline of the Everyday Sounds of Dementia project, in which the colored dots depict the various contexts in which the activities took place. The first study is directed primary at the conceptual (yellow) context. The other two studies are directed chiefly at the real-life-practice (red) context.

In some cases, studies run in parallel because each calls for work in a different context. This held true in the *Track*, *Trace*, *Trigger* project. The researchers planned parallel engineering and qualitative user-oriented studies from the start. The technical study concerned the theoretical context, with tests and development in a laboratory setting. The qualitative study, conversely, unfolded in the conceptual and real-life-practice realms.

CONCLUSIONS: RESEARCH PATHWAYS OVER TIME

Iterating is characteristic of design research, and we find evidence of this proposition in all of the projects. There are different ways of iterating. The examples show how different goals require different ways of iterating. The RPM helps to show these differences because we can map the manner in which the research pathways operate differently across the three contexts. We find two main pathways, which are both non-linear processes of moving between the three contexts. The first way entails the iterative development of a specific evidence-based eHealth innovation. In this approach, we recognize both the design science approach and the evidence-based approach of healthcare studies. The other way is the iterative process of fleshing out a conceptual framework by conducting explorations in design, in line with the process described by Stappers and colleagues (2015). Section 3.2 showed how both health and design research are geared towards theory, although the types of theory differ. The two different approaches show that the term 'theory oriented' can also be operationalized in different ways.

Table 3.4 summarizes the main insights, the ways of iterating that we found in the projects and the relevant literature.

ITERATING: "WORK ITERATIVELY: LEARNING FASTER BY MAKING MISTAKES FASTER"				
TOPIC	INSIGHT	LITERATURE	PRACTICES FROM THE PROJECTS	
Ways to build evidence over time	Different goals require different ways of ite- rating. There is always a non-linear process of moving bet- ween theory, concepts and practice.	Iterating in health vs HCI (Blandford et al., 2018); design and research interwoven in RtD (Stappers & Giaccardi, 2017; Zimmerman et al., 2007); agile approach in multidisciplinary healthcare innovation teams— agile process in design sprints (Hermsen et al, 2020); explora- tions with frameworks (Stap- pers et al., 2015).	 Explore a framework through open and explorative approach: iterating between theoretical, conceptual and practical con- text (to open solution space and develop theory; "an iterative process in which ends and means alternate" 	
			 Develop an evidence-based prototype iteratively that moves, roughly, from theory through concept to practice <i>"a chance to develop an evidence-ba- sed mechanic'"</i> 	

Table 3.4: Insights on ways of iterating in eHealth projects. Emergent practices from the projects and references to the relevant literature are added.

3.7 MAIN INSIGHTS

This chapter addressed the following research question: *how do the ways of working in eHealth research projects add value to the quality of the process?* We distinguished between several practices that can help to address the challenges in Create-Health collaborations, as identified by Blandford and colleagues (2018) or Groeneveld and colleagues (2019). These practices are organized in four themes. We found that Create-Health ways of working can be characterized by 1) a focus on theory, with a wide range of outputs for various audiences and, in light of these goals, by 2) various ways of working with prototypes, 3) method mixing in all phases and with different levels of disciplinarity and 4) an iterative research path over time.

The practices from each theme are presented in Figure 3.16. In the early stages of a project, research partners can discuss the four themes to create common understandings. They can use the overview of the emergent practices as an inspiration for their own efforts in innovation and research. Funding bodies can consider these topics when formulating criteria for

calls, when steering projects and when monitoring and assessing them as well as when they create opportunities for follow-up projects.



Figure 3.16: Tips on ways of working in eHealth projects that emerge from the four themes.

CHAPTER "

CREATE-HEALTH WAYS TO INVOLVE PEOPLE IN RESEARCH PROJECTS ON EHEALTH

WILKE VAN BEEST

"WITH A TARGET GROUP OF ELDERLY PEOPLE, YOU HAVE TO MAKE SURE THAT YOU INVOLVE THEM WITHOUT ASKING TOO MUCH OF ANYONE. IF YOU HAVE A TANGIBLE PROTOTYPE, IT COULD BE A GOOD TIME TO INVOLVE THESE OLDER ADULTS: BUT IF YOU DON'T HAVE THAT YET, IT MIGHT BE TOO ABSTRACT, AND IT COULD BE BETTER TO WORK WITH REPRESENTATIVES."

-TRACK, TRACE, TRIGGER PROJECT

4.1 INTRODUCTION

The Create-Health ways of working strongly emphasize the life experiences of those whom the innovation may affect. Therefore, the aim is often to involve the target group and other stakeholders in the research projects. Active collaboration with users in the innovation process supports the development and implementation of technologies that are suitable for the context and which are of high quality and of established value (Pagliari, 2007). There is evidence to suggest that the involvement and engagement of actors at an early stage of the innovation process leads to improvements in both healthcare practices and outcomes for the target group of patients (Boaz et al., 2018; Kok et al., 2016).

Early stakeholder involvement, interdisciplinary collaboration and business modeling are essential for the implementation of eHealth innovations (Pieterse et al., 2018). Stakeholder engagement is considered important. However, in healthcare, it is problematic in the early stages of innovation processes (Concannon et al., 2019). Creating long-term value in eHealth is often a complex task. All key stakeholders have to agree on and commit to a plan to create value with the eHealth application in the context of their own situations or interests, which is often problematic due to conflicting values (Geissbuhler, 2013).

Zooming in on the early stages of eHealth innovation processes and the role that research projects play in eHealth, a variety of studies have shown that the contributions and efforts of actors have to be taken into account (e.g. Kok & Schuit, 2012; Saapen & Van Drooge, 2011). At the same time, involving stakeholders from multiple disciplines and with disparate interests in eHealth research projects is challenging (Nielsen & Mathiassen, 2013; Van Limburg et al., 2015). Many research projects on eHealth are still monodisciplinary and expert driven, and the need to involve a diverse range of stakeholders is often neglected in practice (Pieterse et al., 2018). Moreover, even when stakeholders are involved in the early stages of an eHealth innovation processes, they are not always involved in setting the goals of the project, which would enable them to co-create value for themselves. Therefore, even if stakeholders are involved in research projects on eHealth, the question of when, how and in which role to involve them still poses challenges (Pieterse et al., 2018).

One specific aspect of co-creation has to do with the manner in which stakeholders can benefit directly from participating in activities in research projects on eHealth, besides the ultimate benefits of the implemented results of such innovation projects (Nicolas et al., 2019). Elderly individuals are mostly assumed to accept what designers and researchers offer to them, and it is therefore the tasks of designers and researchers to understand and meet their needs as participants who co-create (Peine et al., 2014; Sanders & Stappers, 2008). This chapter investigates the Create-Health ways of working that were adopted to involve the target group and other stakeholders in research projects on eHealth as well as the manner in which stakeholders can benefit directly from participating in project activities. It answers the following question: how do the Create-Health ways of working create value for the target group and other stakeholders during their involvement in research projects?

In research projects on eHealth, involving the target group and other stakeholders is often not straightforward. We studied the following dimensions in order to involve individuals in the 10 research projects of the Create-Health program:

- Ways to find and involve individuals who can participate and
- Ways to create value for the target group and other stakeholders when they participate.

The aim of this chapter is to find ways in which the research projects can be of value to actors who are involved in the process of co-creative research. Before we delve into the examples, which are drawn from the 10 research projects, we discuss the two themes from a theoretical perspective. We start with the theoretical perspective on ways to find and involve the target group and other stakeholders. Then, we discuss the value that is created for all stakeholders in research projects on eHealth.

4.2 THEORETICAL PERSPECTIVE ON THE TWO THEMES

Understanding how Create-Health ways of working can support the involvement of the target group and other stakeholders in research projects on eHealth and the manner in which stakeholders can benefit directly from participating in activities in such research projects requires two problems to be discussed. The first concerns the means of finding and involving the target group and other stakeholders into Create-Health and the research pathway of research projects on eHealth. Therefore, in Section 4.2.1, we investigate the theory of stakeholder participation. In Section 4.2.2, we discuss service-dominant (SD) logic as a way to rethink the role of researchers in creating value for everyone who is already involved in eHealth research projects.

4.2.1 STAKEHOLDER PARTICIPATION IN RESEARCH PROJECTS ON EHEALTH

In this section, we discuss the roles of the individuals who are involved in research projects on eHealth. The Create-Health program involved four types of stakeholders in each of the 10 projects: 1) research partners from the healthcare domain, 2) research partners from the design domain, 3) creative partners (at least one undertaking) and 4) representatives of the target group. There is a growing consensus among scholars that research-based innovation projects require an iterative process of interaction between scientists and stakeholders, which should entail reciprocity in terms of knowledge and experience (D'Este et al., 2018; Nijland et al., 2011). Therefore, it seems relevant to explain which stakeholders are important and what their various roles during the project are (D'Este et al. 2018).

The literature provides a few typologies of actors in healthcare. Several frameworks advocate a user-centered approach that focuses on the needs of two actors as end users of innovations, namely patients and healthcare professionals (Van Gemert-Pijnen et al., 2011). Other frameworks refer to a variety of actors in order to document the complex relationships between political, social, organizational and technical contexts in innovation (Van Gemert-Pijnen et al., 2011). The labels for actors that are used in the literature include 'patients,' 'healthcare professionals,' 'informal caregivers,' 'buyers,' 'sellers,' 'producers,' 'suppliers,' 'managers' and many other specific terms.

However, the use of these restrictive pre-assigned labels does not do justice to the complex and evolving nature of the healthcare context (Vargo et al., 2020). Therefore, Vargo and colleagues (2020) refer to stakeholders as 'actors' that provide services to each other. Vargo and colleagues (2020) provide a classification of these actors that is sensitive to their role in the value exchange that service provision entails. They are thus classified as economic actors, social actors and beneficiary actors. This classification seems appropriate for research projects on eHealth, in which different complementary interests play a role (Pieterse et al., 2018). The health expert or professional may be associated with social actors. IT specialists, such as software and hardware developers as well as designers, may be associated with economic actors. The patient, their relatives and informal caregivers may be associated with beneficiary actors. Institutional arrangements may be connected to consortia, in which researchers work alongside different actors, but also with the wider system (e.g. the government or health insurance companies).

Beyond actor types, it is useful to account for the types of interactions that occur between actors and project. Actors can engage in the innovation process at different stages and to different degrees. Dewaele and colleagues (2021) argue that the more involved an actor, the more likely it is that the research-based innovation project will meet their needs, that it will be picked up and that it will have an impact. However, Merkel and Kucharski (2019) contend that full involvement is not always feasible and manageable. Therefore, researchers should ask themselves who to involve and when as well as how and why they should involve these actors. The target group of older individuals provides a salient example (Merkel & Kucharski, 2019). By asking these questions, researchers are able to identify the forms of interaction that are most suitable for a particular actor during specific phases of the research-based innovation project (Deweale et al., 2021).

Arnstein's 'ladder of participation' (1969) can be used as a tool for analyzing participation levels in the healthcare context (De Wit et al., 2015; Lemmens et al., 2015). Smits and colleagues (2020) make several adjustments on the basis of a narrative review so as to design a practical tool that is intended for research projects. Instead of 'levels,' they write about 'roles.' To emphasize the equality of the roles, they position the ladder horizontally rather than vertically and only include roles that involve working together on a project, which means that respondents (who are present without possessing any real understanding of the project) are excluded. The five roles are: listeners, who are given information; co-thinkers, who are asked to share their opinions; advisors, who provide (un)solicited advice; partners, who work as equals; and decision-makers, who take the initiative and/or make decisions (Smits et al., 2020).

In summary, we include three types of stakeholders that participate in research projects on eHealth: economic actors, social actors and beneficiary actors (Vargo et al., 2020). Researchers who coordinate research projects on eHealth should ask themselves whom they should involve from which of the three actor groups and when as well as how and why they should involve them (Merkel & Kucharski, 2019).

In Section 4.3, we delve into examples from the 10 projects to find ways to involve individuals in research projects on eHealth. However, we first discuss SD logic in Section 4.2.2. It provides a theoretical perspective that enables us to rethink the role of researchers in creating value for everyone who is already involved in research projects on eHealth.

4.2.2 SERVICE-DOMINANT LOGIC AS A WAY TO ADD VALUE FOR EVERYONE INVOLVED

Researchers can use the theory of service-dominant (SD) logic (Nicolas et al., 2019; Vargo & Lusch, 2004) to rethink the relationship between themselves, the actors and their role in research projects on eHealth. Vargo and Lusch (2004) introduce SD logic as an alternative to the theory of goods-dominant (GD) logic. In doing this, they propose a shift from a goods-centered model that focuses on the value that is created by the exchange of products towards a service-centered model. SD logic and the service-centered model assume that value is created during the performance and exchange of services, that is, the application of competences for the benefit of others, and which may—or may not—occur in tandem with the application of products (Vargo & Lusch 2004, 2008). Vargo and Lusch (2004) argue that when services are exchanged, value is always co-created rather than created and subsequently delivered by a single actor. The exchanges that take place between individuals who are involved in the project may unfold throughout the entire innovation process, from the idea phase to commercialization and beyond (Aarikka-Stenroos & Sandberg, 2012; Reypens et al., 2016).

Joiner and Lusch (2016) point out that GD logic dominates healthcare innovation, and its primary focus is still on the delivery of goods: 'GD logic as applied to healthcare is about nouns. Hospital rooms, outpatient clinics, medications, medical devices, medical images, laboratory tests, doctors, nurses, electronic health records, accountable care organizations, genomics, and on and on. SD logic is about verbs: healing, caring, monitoring, resting, walking, talking, eating, sleeping, visiting, learning, feeling, curing, thinking, sharing, recovering, and dying' (Joiner & Lusch, 2016, p. 27). In the context of eHealth, for example, the GD logic focuses on tools rather than on the value that they create when they function. The language that is used in healthcare are in line with the values behind eHealth tools, such as preventing obesity, living at home for a longer time and reducing loneliness. These values are not applicable solely to the moment in which an eHealth tool is implemented in practice; they can also play a role in research-based innovation projects.

In particular, in innovation processes, such as research projects on eHealth, the term 'service' in SD refers to the way of working during an innovation process (Vargo & Lusch, 2016). Therefore, in this case, 'service' means the exchange of competences (knowledge, skills and tools) during the process. The focus is not on the output of projects (Vargo et al., 2020). The SD logic thus emphasizes the application of knowledge and skills as well as of supportive products, artefacts or prototypes for the benefit of others, all in line with the values behind the project, in our case obesity, living at home with dementia for longer periods of time and reducing loneliness (Vargo & Lusch, 2004). Previous research has established that researchers and/or designers can play a role in the coordination of the co-creation process, in order to understand and meeting the needs of other actors in the process (Peine et al., 2014; Sanders & Stappers, 2008).



Figure 4.1: Exchange of knowledge, skills and tools between disciplines.

The theoretical perspectives on stakeholder involvement in eHealth research projects are summarized in Table 4.1.

Researchers should think of who, when, how and why they should involve specific actors:

SECTION 4.3 OF THIS CHAPTER:	
Who are involved:	Creative industry partnersHealth and welfare partnersThe target group
In which specific research step in the resear	rch pathway (when)?
How are they involved:	 Listener Co-thinker, Advisor Partner Decision maker
SECTION 4.4 OF THIS CHAPTER	

Why are they involved: the value for the actor involved

Table 4.1: Stakeholder involvement in research projects on eHealth.

As Table 1 shows, in Section 4.3, we explore examples from the 10 projects to find ways to involve individuals in eHealth research projects. Section 4.4 discusses ways to create value for those individuals.

4.3 WAYS TO RECRUIT INDIVIDUALS AND INVOLVE THEM IN PARTICIPATING

Several researchers from the 10 projects experienced the recruitment and involvement of participants from the target group as a difficult task, especially when it came to engaging members of the target group of the intended eHealth innovation, such as overweight individuals (especially those with a lower SES), individuals with dementia or individuals with feelings of loneliness. The example from the *PACO* project in Box 4.1 concerns attempts to approach such individuals.

ATTEMPTS TO APPROACH MEMBERS OF THE TARGET GROUP Project: PACO

BOX 41



Figure 4.2: Co-creation session, PACO project

A researcher on the project explained that it had been difficult to find members of the target group, that is, individuals who want to lose weight: 'We tried to find people via various ways: via social media, we tried to find people... In some newspapers, we were allowed to publish a two-page article with a photo and all, and we had received a call list from the National Foundation for the Elderly for people who were open to research and whom we were allowed to call to ask if they wanted to participate. Via these diverse ways, thousands of people were contacted. In the end, it only resulted in 52 people who were actually registered. We really aimed for a minimum of 60, preferably 70, so that we could still correct for the dropouts during the research projects. So, we had hoped for more participants but were unable to find more. At a certain point, we also had to start the study, to prevent further delays.' Furthermore, not only was it often difficult to find members of the target group, it was also often difficult to keep the target group engaged during the research project, as in PACO: 'I found it remarkable that participants who participated in the co-creation session indicated what they would like to see in an application. We then developed this further, but then they no longer wanted to participate in the evaluation of the app.'

In the example in Box 4.1, individuals were able to think about their needs and preferences for an application but did not want to be involved in the evaluation study for that application. In this section, we discuss ways to recruit individuals and ways to keep them involved in research projects on eHealth.

When we looked at the types of stakeholders within the 10 projects, we saw similarities and differences. In some projects, the primary researchers were all from the same single discipline (either health research or design research). Other core project teams encompassed a combination of disciplines. However, if we look at the three types of stakeholders that we saw as being theoretically necessary in research projects on eHealth (see Section 4.2.1), it is clear that they feature in all projects (also because the research call prescribes their involvement). We stated that there were three types of actors: economic actors (creative partners), social actors (healthcare or welfare organizations) and beneficiary actors (the target group, their relatives and/or informal caregivers). For ease of exposition, we retain the terms that were used in the ZonMw call and refer to the target group alongside their relatives and/or informal caregivers (beneficiary partners) as well as to healthcare and welfare organizations (social partners) and creative partners (economic partners), as in Figure 4. 3.



TARGET GROUP: PATIENTS, CLIENTS AND INFORMAL CAREGIVERS AS WELL AS CITIZENS OR REPRESENTATIVES OF TARGET GROUPS



HEALTH PARTNERS: HEALTHCARE Organizations, welfare partners and municipalities



CREATIVE PARTNERS: PROFESSIONAL DESIGN AGENCIES, PROFESSIONAL TECHNOLOGICAL DEVELOPERS AND CREATIVE INDUSTRY PARTNERS

Returning to the roles identified by Smits and colleagues (2020), all of the researchers in the core teams of the 10 projects were decision-makers. Researchers (mostly professors) who hold roles on advisory boards or as supervisors of the PhDs who were involved were also engaged within the 10 project consortia. As discussed in Section 4.3.1, researchers who coordinate research projects on eHealth should ask themselves which of the three actor groups to involve and when as well as how and why they should involve them (Merkel & Kucharski, 2019). The choice of research methods and ways of working in the projects had several motivations, which involved specific actors (creative partners, healthcare and welfare organizations (including healthcare professionals), and the target group with their relatives and/or informal caregivers) in specific roles, namely as listeners, as co-thinkers, as advisors, as partners and as decision-makers. We use the RPM to map the specific actors and their roles in the research projects on eHealth.

As noted in Chapter 3, activities in all of the projects took place in all three contexts that the RPM distinguishes (see Table 4.2), namely the theoretical context, the conceptual context and the real-life-practice context. The colored dots on the timelines in Chapter 2 are related to these contexts (blue=theory, orange=conceptual and red=practice). In the actor maps in Chapter 2, actors are also placed in these contexts. Mapping the actors by reference to the RPM (Van Beest et al., 2021) provides insights into the questions of who should be involved in research steps and in certain contexts as well as when and how they should be involved. To understand the reasons for involving the target group and stakeholders in certain research steps, we discuss how members of the target group and other stakeholders were recruited and engaged in the projects for each research context.
	THEORETICAL CONTEXT	CONCEPTUAL CONTEXT	REAL-LIFE PRACTICE Context
CREATE & UNDERSTAND	Create theoretical understanding Methods e.g.: Desk research Modeling Literature review Expert panel Delphi study	Create concept Methods e.g.: • Focus groups • Co-creation sessions • Co-design sessions • Design interviews	Create understanding of real-life practice Methods e.g.: Interviews Observations Mood reporting Diaries Surveys
EXPLORE & TEST	Explore or test theory or concept in controlled environment Methods e.g.: • Lab tests • Observations • Activity tracking • Data monitoring	Explore or test a proto- type Methods e.g.: • Pilots • Co-design sessions	 Explore or test solution in real-life practice context Methods e.g.: Observations Activity tracking Data monitoring RCT
DELIVER & IMPLEMENT	 Deliver theory or knowledge Methods e.g.: Dissemination activi- ties: conference con- tributions, papers, presentations Guidelines Knowledge building blocks 	 Deliver a prototype Methlods e.g.: Prepare follow-up research Deliver prototype to developer Deliver algorithm to research team 	Implement change in real-life-practice context Methods e.g.: • Data monitoring • Evaluation research • Change management

Table 4.2: RPM (Van Beest et al., 2021), supplemented with methods from Chapter 3

4.3.1 INVOLVEMENT IN THE THEORETICAL CONTEXT

The theoretical context of the RPM is about research activities, with a focus on theoretical concepts or on gaining, testing and/or sharing knowledge in a controlled environment (living lab research) or in a controlled manner (literature reviews, Delphi studies and publications). In general, the involvement of the target group and other stakeholders in the theoretical context is limited. As we explained in one of our earlier works, 'In the theoretical context the research is focused on creating, exploring and delivering a better understanding of problems and related propositions for solutions, which are advanced and not verified in practice yet' (Van Beest et al., 2021, p. 9). When the goal of research is to find out more about the way a tool works or about its working mechanism, regardless of how the tool in guestion works in the actual environment, it could be helpful to conduct tests in a controlled setting first or to conduct literature research on possible working mechanisms. However, it is possible to involve stakeholders or research participants in the theoretical context, alongside the researcher, for example by engaging in joint analysis, by publishing or by engaging research participants in a living lab situation. Testing in a controlled environment ('Explore a concept or a theory in a controlled environment') with students or colleagues, for example, could be a suitable solution that avoids the need to disturb the target group in some cases. By explaining involvement in the theoretical context in relation to the 10 research projects on eHealth, we discussed the parties who were involved, the steps in which they were involved and the roles of the stakeholders and the participants in the research.

Who is involved in the theoretical context?

In most projects, both creative and health researchers were active in the theoretical context. In Chapter 5, we elaborate on the cooperation between these disciplines. In the *PACO* and *Healthy Storytelling* projects, the practice partner was also involved in theory development. In the *PACO* project, the practice partner was a research and development company. In that project (*PACO*), the representative of the target group was involved in theory development as well. In *Healthy Storytelling*, the practice partner was from the creative industry. In both projects, healthcare partners were not involved in the steps of the research that concerned the theoretical context. Some projects, such as *Track*, *Trace*, *Trigger* as well as *Growing Roots* and *FoodSampler*, involved students or young individuals in some research steps. In some cases, these individuals acted as representatives of the target group. Sometimes, this was done for practical reasons, say because the target group was difficult to reach. Sometimes, a conscious decision was made to have a conversation about future scenarios with younger target groups. These representatives of the target group were also involved in research activities that were related to this theoretical research context, such as lab tests and activity tracking. In which steps are people Involved in the theoretical context?

In two projects (*Storytelling* and *PACO*), the practice partner was involved in research activities to create theoretical knowledge. The *PACO* project also involved a representative of the target group in this step. In this project, the representative of the target group, who was a PhD student, was co-authoring a paper on the research methodology of co-design.

TEST WITH REPRESENTATIVES OF THE TARGET GROUP

BOX 4.2

Project: Growing Roots

In the project *Growing Roots* the researchers were first looking for the mechanisms of the experience of nature in a VR environment as the researcher explained: "In order to map the experience of nature, we were more looking for generalizable effects of that experience of nature in the VR environment, before we want to explore this with the aging population". In the project *Growing Roots* researchers deliberately chose for a representation of the target group in the theoretical context in which students or middle-aged people represent the target group in a living lab study or in online research settings. The prototype in figure 4.4 was used on a screen. The future VR environment will be part of the room as in figure 4.5, another experience which was not the focus in this phase of the research project.







Figure 4. 5: VR environment in a room

Some projects involved the participants in the research step 'test or explore theory or a concept in a controlled environment.' The reason for choosing to conduct research with the participants in the theoretical context is that the usability of the tool or concept (the potential scenario) should be tested by individuals from outside of the project team in order to understand more about the manner in which individuals in general experience future solutions, such as Wi-Fi monitoring. In some projects, the future scenario was too abstract for the target group, and the research was too fundamental, as in the example that is described in Box 4.2.

Depending on the research question, it was sometimes more helpful to test technologies, low-fidelity prototypes or mechanisms with a more neutral target group in the early phase of the eHealth innovation process. However, not all projects saved the target group for more fundamental questions. In the *Healthy Storytelling* project, a deliberate choice was made to conduct research with the intended users, with research methods adapted to the target group. It is unclear whether every type of research lends itself to target-group involvement. The requirements for evidence-based innovations in the healthcare context, when combined with ethical dilemmas, raise questions about the burden that is imposed on the target group.

In some of the projects, the creative partners had a role in the 'Deliver and Implement' step. One example is the Everyday Sounds of Dementia project, which did not involve design-creative partners. Instead, creative practice was reached by the network partner who was responsible for dissemination.

The actors and the research methods are mapped onto the RPM in Table 4.3. In all projects, the researchers published research papers, guidelines and requirements. The target group and the other stakeholders were not involved in this step.

THEORETICAL CONTEXT				
Create theoretical understanding	ACTORS: Researchers and, in one project, a practice partner	Research activities or methods: scoping review, literature review, field and theory immersion		
Create understanding of a theory or concept in a controlled environ- ment	ACTORS: Researchers and, in three projects, represen- tatives of the target group	Research activities or methods: living lab, machine learning, home-lab situation		
Deliver theory or knowledge	ACTORS: Researchers, network partners, creative partner	Research activities or methods: delivering knowledge, building blocks, papers, conference visits, guidelines		

Table 4.3: The theoretical context of the RPM

Role of individuals in the theoretical context

When we look at the role of stakeholders in the theoretical context, it becomes apparent that it differs with the type of stakeholder. Creative partners are involved in these research steps as partners in dissemination. Both the researcher and the practice partner have an interest in understanding theory. In the step 'Create understanding of a theory or concept in a controlled environment,' the representatives of the target group are mainly involved as research participants.

4.3.2 INVOLVEMENT IN THE CONCEPTUAL CONTEXT

The conceptual context translates the presupposed solutions into a more specific prototype that is created, explored and made. In this context, researchers and other stakeholders provide (experiential or theoretical) knowledge to translate theory to prototype by themselves, together with the consortium and with or without the end users, but always in a protected niche such as a brainstorming room, a pilot environment or in the context of a pilot organization (Van Beest et al., 2021). The focus of the conceptual context is to understand needs, requirements and preferences about a technology. This was seen, for example, in the proposal for the *Growing Roots* project, which describes the main goal of the research: 'Based on staff and user feedback, aspects such as ease and type of interaction will be fine-tuned.' The focus on the eHealth application could be partly explained by the specific demand of the funding agency for projects that work on creating the 'knowledge building blocks' that are needed to develop and/or implement eHealth applications. Similarly to Section 4.3.1.1, which concerns the theoretical context, we explain the involvement of individuals in the conceptual context, the steps in which they were involved and their roles in the 10 research projects on eHealth.

Who was involved in the conceptual context?

Creative researchers were active in the conceptual context of the projects in which they were involved. In four projects, the health researchers (who were mostly psychology oriented) were also active in the conceptual context. As mentioned previously, we elaborate on the cooperation between these disciplines in Chapter 5. All kinds of stakeholders, including the creative partners, healthcare and welfare organizations (including healthcare professionals) and the target group (alongside their relatives and/or informal caregivers) were involved in the conceptual context, as shown in Figure 4.6.



Figure 4.6: Stakeholders involved in the conceptual context, including creative partners, healthcare or welfare organizations (including healthcare professionals) and the target group (with their relatives and/or informal caregivers).

In which steps were individuals involved in the conceptual context?

In the contextual context, most creative partners were involved in the 'Create a concept' research step (see Table 4.4). The healthcare or welfare organizations (including healthcare professionals) and the target group (with their relatives and/or informal caregivers) were also involved in this step as part of, among others, co-design sessions, interviews or focus groups. The creative partners were mostly involved in the development of prototypes. However, in the *FoodSampler* project, the collaboration with the creative partner focused on the development of design studies. The prototypes were meant as design artefacts that would generate knowledge (RtD) and therefore required flexible requirements and quick iterations. The university decided to develop the prototypes internally. Due to the ease of accessing the prototyping infrastructure and related resources, the university was more flexible in performing iterations than the creative partner. As a whole, the creative partners were largely not involved in the 'Explore and Test' step and therefore had little interaction with the care partners, who were mainly involved in that step. In some of the projects, the creative partners had a role in the 'Deliver and Implement' step.

CONCEPTUAL CONTEXT		
Create a concept	ACTORS: Researchers, tar- get group, creative partners, healthcare and welfare organi- zations	Research activities or methods: Service design, technical design, interviews supported by the prototype, interviews to understand needs and preferences regarding a tool, co-design sessions, visualizing routines, sessions with experts on the solution, focus groups, contex- tual interviews, expert reviews, student design projects
Test or explore a concept	ACTORS: Researchers, target group, healthcare and welfare organizations	Research activities or methods: work- shops with care professionals, workshops with the target group, use in pilot, usa- bility study with target group, mock-up study
Deliver a concept	ACTOR: Researchers, creative partners	Research activities or methods: delivery of an algorithm, delivery of a prototype, working on follow-up project

Table 4.4: The conceptual context of the RPM

The practice partner in *DDD* was also involved in dissemination activities and in the delivery of a prototype. Initially, the aim of *DDD* was to deliver a low-fi prototype. However, COVID-19 restrictions resulted in one of the research artefacts being developed as a hi-fi prototype. The low-fi prototype required greater involvement on the part of the researchers because that prototype could not be used on its own—researchers had to be present.

However, this was not possible due to the COVID-19 contact restrictions. The project team decided to develop the prototype further so that individuals would be able to test it by themselves at home, without the researchers' help. In this case, conducting an iterative testing cycle with participants was not possible, so the step of delivering a prototype came earlier in the process. An application was also delivered in the *PACO* project, which was the basis of a new service that the practice partner would offer. The application will be developed further in the *PHArA-ON* project, in which it will be applied in practice. The practice partner will develop and implement the application on its own devices. The healthcare partners in some projects became enthusiastic and wanted to retain the prototype, with the support of practice-based research, in order to develop it further.

4.3.3 INVOLVEMENT IN THE REAL-LIFE-PRACTICE CONTEXT

The real-life-practice context concerns the setting in which researchers aim to understand the target group and its environment. Methods that have researchers observe the environment or conduct interviews aimed at understanding the target group are part of research on the real-life-practice context. In this type of research, tools or prototypes can be used as research artefacts or as RtD objects, but the 'thing' is mostly not intended to be developed further. The other methods that are applicable to this research context examine the functioning of a prototype or product in its intended environment, that is, the real-life-practice context. The issue is not whether an eHealth tool works but how it fits into the environment. A condition for this type of research is that the tool, process or intervention must have already been developed in such a way that practitioners can use it independently. Therefore, implementation research (or evaluation research) is also a part of the real-life-practice context. However, the 10 projects of the Create-Health program are aimed at delivering knowledge building blocks for the ultimate implementation of eHealth, not on implementing eHealth tools in the real-life-practice context.

Who was involved in the real-life-practice context?

In almost all of the projects, it emerged that it would be complicated to involve the target group. This was observed in respect of research activities in the conceptual context as well as in the real-life-practice context. It was complicated to involve the target group because topics such as obesity, dementia or a lower SES are delicate. An empathically non-judgmental tone and attitude are necessary during contact, which must be reflected in the research methods and in the design. A few projects revealed that flexibility is important for finding appropriate partners. In the *FoodSampler* project, researchers from the healthcare sector and end-user organizations that were partners of the project helped find healthcare professionals (mostly dietitians) and overweight individuals. The tendency to involve this target group and expert organizations and to work with them was helpful for finding the best individuals for the project.

COVID-19 made involving the target group, as well as healthcare and welfare organizations, in research projects more challenging. The challenges resulted from the burden on the healthcare sector and the need for social distancing. Since the healthcare sector was overloaded, there was less time to engage healthcare professionals, to recruit members of the target group and/or to facilitate the research project. In addition, many nursing homes, community centers and schools were closed. Therefore, the target groups of elderly and overweight individuals became more vulnerable.

In which steps were individuals involved in the real-life-practice context?

Several projects began by involving members of the target group in order to understand their needs and preferences about their situation and their views on the future. For example, in the *Track, Trace, Trigger* project, informal caregivers were consulted in interviews and through a survey (*'Create real-life understanding'*). Initially, the researchers would also carry out measurements in practice, but these could not continue due to COVID-19 and the resultant policies. In the *Healthy Storytelling* project, the researchers first built on their relationship with the target group and then invited them to co-creative sessions. As explained in Box 4.3, *'one could say that researchers' first step into the world of real-life practice is to understand the situation of the target group before they invited the target group into the world of concepts for a co-creative session.'*

FIRST STEP INTO THE WORLD OF REAL-LIFE PRACTICE BEFORE INVITING THE TARGET GROUP INTO THE WORLD OF CONCEPTS

BOX 4.3

Project: Healthy Storytelling

Researchers of the project Healthy Storytelling indicated that – in their case – involving people with a lower SES indication at individual level was stigmatising and it hindered the possibilities of contact with the target group. Therefore, they recommend that this criterion be applied mainly at the neighbourhood level and not at the individual level. An empathetic approach was essential to reach people of the target group: "Before people from vulnerable groups can be included in co-creative sessions, it is first necessary to reach them. In this project we did this by building trust for a couple of weeks with the visitors of neighbourhood centres before introducing research methods." In this example one could say that researchers first step into the world of real-life practice before they invited the target group into the world of concepts for a co-creative session.

Almost all projects involved the target group and healthcare partners in the research step 'Explore a solution in real-life practice' (see Table 4.5), in which they would test a hi-fi prototype in the healthcare context together with the stakeholders associated with that specific context. The researchers on the Track, Trace, Trigger project took a different approach by involving different target groups at different points. The project began by engaging informal and formal caregivers (who also served as representatives of their loved ones or of clients with dementia). When the technological concept became more concrete, individuals with dementia and a broader group of relevant stakeholders were involved in focus groups and interviews, building on the previous steps: 'With a target group of older adults, you have to make sure that you involve them without asking too much of anyone. If you have a more concrete technology concept, it could be a good time to involve these more vulnerable group of people; but if you don't have that yet, it might be too abstract, and it could be better to work with representatives.' The researchers acted in this manner on purpose because they knew from previous research that the timing of the involvement of end users and stakeholders is important. For instance, collecting the views of older adults on more concrete technological concepts often works better than asking them to envision abstract scenarios. However, the researchers did not choose to present a hi-fi prototype during the interviews and the focus groups because their goal was to enable participants to think beyond technical possibilities.

In the *Everyday Sounds of Dementia* project, the target group and the healthcare professionals were involved in the real-life-practice context (care homes). In the real-life-practice context, the researchers employed an existing product, which could be used by healthcare professionals and people with dementia. Both tools worked well during the research project and were appreciated by the intended end users, healthcare professionals and informal caregivers or relatives. However, in the real-life context, the participants did not always understand that there was no implemented product or implementation-ready solution yet.

REAL-LIFE-PRACTICE CONTEXT				
Create real-life understanding	ACTORS: Researchers, target group, healthcare and welfare organizations	Research activities or me- thods: surveys, longitudi- nal interviews, interviews, focus groups		
Test or explore a solution in real-life practice	ACTORS: Researchers, target group, healthcare and welfare organizations	Research activities or methods: participatory ob- servation, deployment in care homes, longitudinal testing at home, longi- tudinal field evaluation, testing research artefacts at schools, randomized controlled trials, evaluati- on research		
Implement a solution in real-life practice	ACTOR: Researchers	Research activities or me- thods: mapping require- ments for implementation, discussions with ethical committees of the EU, investigating possibilities for adaptations of laws and regulations		

Table 4.5: The real-life practice context of the RPM

In this case, the real-life-practice context called for the implementation of a prototype, but the step of implementation was difficult to achieve and was therefore not pursued. As a researcher on the Healthy Storytelling project noted, 'The step from a demonstrator to a scalable product could not been taken in a few steps. We are now going to see whether it is possible to scale up on a small scale first, i.e. to 10 schools, maybe with an evaluation. Then, we need money, and [we] need someone who will carry out the evaluation. Then, we can apply for a quality mark from RIVM for the intervention, and then it could be scaled up much later afterwards, but then you do need parties that can manage it, and that is a bit of a problem.' Although the transition from prototype to product is difficult (and therefore mostly not the intention of the researchers because of the focus on knowledge building blocks rather than end products), in some projects, such as DDD, the prototype was developed into a high-fidelity prototype that could be used by individuals (as we described in Section 4.3.2, p. 14) more rapidly than intended.

The implementation of products, services or processes was not the purpose of the projects. The set-up of the research call, for which researchers could submit applications, focused on knowledge building blocks 'for the purpose of developing fundamental knowledge – known as "knowledge building blocks" – on which to base the development or improvement and implementation of e-health applications which are intended to support the day-to-day functioning of people as they grow older, both now and in the near future' (Research Call, p.14).

This resulted in various types of outcomes, ranging from theory and guidelines to prototypes (which we discussed in Section 3.3). In the *Track, Trace, Trigger* project, the researchers studied the conditions for implementation. This said, they were active in creating and understanding their solution and in testing technical possibilities in that phase of the project.

Role of individuals in the real-life-practice context

Many patients or clients are often no longer completely independent, which means that they were usually in the presence of another individual during data collection. This does not always work well, as observed in *Track*, *Trace*, *Trigger*. *'The subsidy provider advised us to include patients as well for the interviews, but it did not work in combination with the informal caregivers*. Both do not really give their opinion about the situation in the presence of the other.' In the Everyday Sounds of Dementia project, the combination of a client and an informal caregiver worked well during a workshop in which the client and the informal caregiver related sounds to memories. Nevertheless, it should be noted that, here too, the researchers had to prevent the informal caregiver from speaking for the individual with dementia. In this case, one could say that the individuals with dementia acted more as participants than as co-thinkers or listeners (Smits et al., 2021).

In the real-life practice context, the healthcare and welfare partners were involved in the research methods in which the target group was involved, as well as in participant recruitment and in facilitating the sessions. Some network partners, municipalities and target-group representatives were involved in the real-life-practice context by recruiting members of the target group.

4.3.4 INVOLVEMENT AS A LAYER IN THE RPM

The involvement of the target group and other stakeholders was recognized in the theoretical context, the conceptual context and the real-life-practice context. The target group was mostly involved in the steps 'Create a concept,' 'Explore a concept in a pilot,' 'Understanding real-life practice' and 'Explore a solution in real-life practice.' The healthcare and welfare partners were involved in the same research steps as the target group. The creative partners were mostly involved in the conceptual context. However, we saw that the creative partners, the healthcare partners and the target group were almost never involved together in co-design or other research activities (Figure 4.7).



Figure 4.7: Involvement as a layer in the RPM.

In the next section, we delve into the role of researchers in creating value for everyone who is already involved in research projects on eHealth.

4.4 WAYS TO CREATE VALUE FOR THE TARGET6ROUP AND OTHER STAKEHOLDERSWHEN THEY PARTICIPATE

The first part of our research was done in co-creation, and it was mainly about values. An application was to be developed about healthy eating, healthy living. How do you define these? Together with participants, who could be participants in that app, we formulated those values in co-creation. Therefore, of course, we needed participants, and asked the elderly fund to recruit these people. We wondered: in what kind of setting could we formulate those values? What would such a conversation look like? How can you discuss these values in a pleasant way for everyone? We thought it through carefully and opted for a rural setting at a location where food was used: at a farm with a tea garden. We made it a kind of relaxing trip in a pleasant environment. There, we interviewed the participants, had conversations. There was a lot of enthusiasm for that. We had enough registrations for this study; people wanted to participate in this (PACO project).

In this example, the researchers found a way to involve the target group in a manner that was pleasant for all involved actors. They found a way to create a reciprocal experience by

making individuals enthusiastic about the project activities, and they were able to formulate values together. We use the concept of SD logic (Vargo & Lusch, 2004, 2008, 2016) to analyze the manner in which research projects could create value for all stakeholders who are involved during a research project. In this way, we want to provide examples from the 10 research projects in which different stakeholders contributed and benefitted in such a way that the project was not just valuable for the researchers but also for everyone else who was involved.

From an SD-logic perspective, one could argue that the project in the example above integrated end values into a way of working. The project aims to change dietary behavior among older adults. By talking to individuals about their own values of healthy eating and healthy living, they were made to think about their dietary behavior, which is what the final app that the research project is intended to yield aims to do. For the research participants, participation in the study therefore also contributed to their healthy ageing independently of the eHealth application, which is being developed and may be implemented. This section explains why individuals would become involved in research projects on eHealth and why they would like to be so involved.

4.4.1 VALUES IN CREATE-HEALTH WAYS OF WORKING FOR THE TARGET GROUP

The projects involved the target group in data collection or in value co-creation. Some researchers chose research methods so that they would meet the needs of the participants directly. For example, some projects (like DDD) involved building a relationship deliberately in order to be able to conduct longitudinal interviews. Others opted for means of creating a comfortable environment, such as the PACO project in the example at the beginning of Section 4.4. Likewise, the project leader of a loneliness prevention project (SQUEALED) invited the elderly for coffee breaks, in which there was time to talk and to get comfortable: the first step is that the elderly receives a letter inviting them for a meeting with coffee and cake.' In the example of SQUEALED, the researchers contributed to loneliness prevention by opting for a welcoming approach during interviews. We identify the interview method as a preferred way of working in projects that involve collaboration with the elderly as a target group. For example, in Track, Trace, Trigger, 'one of the lessons learned for us as researchers, was that an "old-school" interview worked the best for this target group. They want to talk about their experiences, their lives, and it brings us a lot of rich information.' Exchanging values in an interview setting is recognized as a means of sharing experiential knowledge and of enacting the values that underlie the project, namely to prevent loneliness and/or to promote healthy ageing.

Personal contact between researchers and participants was also seen as an important means of involving the target group. The *PACO* project team concluded that personal contact with participants (especially older ones) was very important for the smooth running of the study: 'during the evaluation study, for example, we called all participants after they had created an account to ask if everything was clear. This solved many problems, but also any miscommunication. We also made video messages, which were much appreciated.' This example shows that the role of researchers in understanding and meeting the needs of actors was important not only during research activities but also before they had begun and after they had concluded. This example is in line with Vargo and Lusch's (2004, 2016) service concept, which is not about a product, a service or the execution of a method but about an entire process.

All these examples of value exchange with a view to meeting the needs of participants are related to service ways of working, that is, to interactions in which valuable contributions and benefits are produced with and for everyone who is involved. However, not all research methods are conductive to the attainment of this goal. Suitability seems to vary with the type of actor. In Track, Trace, Trigger, using surveys of informal caregivers to find out about their issues was perceived to work well. In Growing Roots, the use of a survey of elderly individuals was counterproductive. The elderly participants found it difficult to complete the questionnaire, and it disturbed them. In this particular example, one could say that value exchange was absent. As a result, the research team adjusted its strategy and adopted a service way of working by serving coffee and supporting an activity at the healthcare organization. Through these activities, the researchers collected data from the target group by conducting interviews in a manner that was pleasant for the target group. The survey in the Track, Trace, Trigger project operated in a different way and had a different target group. The researchers used a platform that enabled informal caregivers to collect survey data. The informal caregivers were already using the platform and could provide their perspective on the survey in a way that was not burdensome or time consuming. Eventually, both research teams (Track Trace Trigger and Growing Roots) collected data in a way that was attuned to the daily lives of the participants.

In sum, it is important to identify appealing forms of research activities that are appropriate to the needs and preferences of actors in order to involve the target group. On the whole, members of the target group favored involvement in the research projects as long as the research method focused on their own situation and on the present rather than the future. SD logic assumes that value is created during the performance and exchange of services. When a way of working fits into the daily life and the habits of the research participants (their own processes), they are enabled to exchange knowledge in a more equal way. It should be noted that surveys involve little direct exchange, an SD-logic prescription, but the example of *Track*, *Trace*, *Trigger* shows that informal caregivers participated in a survey that provided room for their (experimental) knowledge. Moreover, several respondents gave their contact details so that they could be contacted for follow-up research. In this particu-

lar example, the theory assumes that the research participants contribute by sharing their knowledge and benefit from being provided with a setting in which they can tell their story so that something can be done with it.

4.4.2 VALUES IN CREATE-HEALTH WAYS OF WORKING FOR HEALTHCARE AND WELFARE PARTNERS

The healthcare and welfare partners mostly contributed by recruiting members of the target group and by facilitating the data collection sessions as well as by sharing their knowledge and experience during focus groups and interviews.

In some projects, the researchers introduced a new working method or service into the healthcare organization. For example, in *Everyday Sounds of Dementia*, the researchers introduced a workshop for elderly individuals and their informal caregivers or relatives. The project team set up an activity together with the healthcare organization in which they supported contact between healthcare professionals, individuals with dementia and their relatives: *'We see that the activities we carried out in our design research are appreciated by the care partner and that the first workshops were even seen as a possible permanent activity within the program of the daytime activities.'*

We recognized that if the target group and the health and welfare organizations experienced a way of working that suited their daily practice, they would indicate that they want to use it or engage with it more often. However, we also recognized that the examples of implementing a service approach are mainly at the interpersonal level, with engagement limited to a few actors. The researchers on the *NATALIE* project wanted to implement a helpdesk at one of the healthcare organizations (at the organizational level) to support the elderly as well as healthcare professionals by using the prototype *in situ*. However, the implementation of the helpdesk was pre-empted by the active adoption of another tool at the organization at the same time. The interpersonal-level exchange supported value co-creation for those who were involved in research projects, but it was difficult to expand this value creation to other contexts, stakeholders or even phases of the research projects on eHealth.

Many researchers from the 10 projects acknowledged that the input from professionals was valuable and important in the healthcare context, especially when it came to implementation issues that affect the integration of new technology into that context, as in the *Track*, *Trace*, *Trigger* project. The technical staff at one of the healthcare organizations that were involved in this project was released from other work for a substantial number of hours in order to spend time with the researchers in order to identify the possibilities and the limitations of integrating eHealth at the organization. The researchers could only obtain this kind of information from individuals who were engaged in healthcare processes on a daily basis

and who had a bottom-up perspective on the systems and processes involved. Moreover, for the successful implementation of eHealth, the early involvement of technical staff, such as IT specialists as well as software and hardware developers, seemed essential to ensuring that everyone would be on the same page.

In sum, exchanges of knowledge, skills and tools were part of the collaboration between the healthcare and welfare partners and the researchers. We recognized that usefulness for the organization was an important factor for the healthcare and welfare partners who were involved in those projects. As a researcher on the *Track*, *Trace*, *Trigger* project explained: 'Usually, you have to entice healthcare workers to talk to you. That is difficult and time consuming. We had a completely different experience at a healthcare organization that we did not officially work with. They had people from the technical staff who really had time to think with us (...) Dedicated time from technical healthcare professionals, who contribute not directly scientifically, but really based on a practical question and need, that works well.' Therefore, the involvement of healthcare and welfare partners created value not necessarily by facilitating data collection but rather by providing an equal partner who would think about the manner in which an eHealth application could be integrated into the healthcare context in the future. Similarly to SD logic, the theory recognizes the reciprocity of the relationship between the different stakeholders.

4.4.3 VALUES IN CREATE-HEALTH WAYS OF WORKING FOR CREATIVE PARTNERS

The creative partners were mostly involved in the development of prototypes. Some projects worked with an eHealth application that had already been developed by a creative practice partner. As may be expected, research on most eHealth innovations and their development take more time than the duration of the projects in this call. As a result, it is not yet certain whether the research product that is being developed will be implemented as a solution. Creative partners often do not benefit economically from research projects and may instead benefit from design guidelines and by gaining insights during the process.

Most of the creative partners invested in the design and development of a prototype with an eye on the outcomes of the project, in particular the implementation of the eHealth application. One of the researchers on the *Growing Roots* project mentioned the long-term investment of the partner who designed and built the prototype: 'Our partner invests hours, material, equipment, yet does not earn anything in the short term. I think he sees the benefit in the long term.' We recognized that creative partners had to invest in Create-Health projects to develop the necessary research artefacts, prototypes and research products as solutions. In sum, as far as the target group, the health and welfare organizations and the researchers are concerned, the benefit of involvement in research-based innovation projects seems to be related to service ways of working during the projects or in the near future. For creative partners, the benefits accrue in the long run, when the eHealth application is implemented and scaled up. Since the creative partners are often not the ones who reap the direct economic benefits of a project, it is desirable, in such cases, for researchers to consider how the project can be made as useful as possible for those partners.

4.5 OVERALL INSIGHTS

The involvement of the target group and the other stakeholders was recognized in all of the research contexts that we analyzed using the RPM (Van Beest et al., 2021), namely the theoretical context, the conceptual context and the real-life-practice context. However, we saw that the creative partners, the healthcare partners and the members of target group were almost never involved together in co-design or other research activities. One of the researchers (from the Track, Trace, Trigger project) mentioned that the connection between the stakeholder groups is important for finding solutions to healthcare issues successfully: 'The difficulty with this type of innovation is that the stakeholder groups also say, "something has to come that really connects all those groups." The easiest thing, and almost all researchers do [it], is to make something for informal caregivers and then again for professionals. Then, you end up on a website or a nice app, but we don't want that now. It really has to go a step further.' In this quotation, the researcher mentioned that most eHealth applications are meant for a specific stakeholder group, but more stakeholder groups mostly deal with healthcare problems. Therefore, the solution to the issues in healthcare should target multiple stakeholders as well as the Create-Health ways of working. In Chapter 5 we delve into collaboration between stakeholders.

By studying the projects through the lens of SD logic, we found approaches to interaction through which we could create valuable contributions and benefits with and for everyone involved; working with a 'survey' as just a 'good,' that is, a 'thing' that is not underlain by service logic, worked less well. In our research, we found that the different stakeholders put different emphases on project values. The target group found value in the research activities during the project, the social stakeholders found it in the integration of activities into the daily healthcare process, either during the project or in the short term, and the economic stakeholders found long-term value in the evidence-based development of the eHealth application.

We found that, researchers are able to deliver value for their target group during projects, even in fundamental research projects of this kind. In this chapter, we provided some exam-

ples in order to present insights on the way in which SD logic can be used as a concept for rethinking the relationship between actors and researchers. We did not want to use SD logic as a prescriptive theory but as a way of seeing Create-Health ways of working from a service perspective by designing the research pathway as a process. We believe that researchers could use such service-based ways of working even more explicitly. The examples in this chapter can inspire them.

Not all of the researchers on the projects intended to follow a service-based approach. However, in the analysis of these 10 projects, the theoretical lens of SD logic highlights examples of researchers acting on the basis of the service approach and/or the value that is the ultimate objective of research and development, that is, supporting the day-today functioning of individuals as they grow older. We believe that it is valuable to embed these ways of working into projects, especially in fundamental research projects in which the collected knowledge is necessary for further development. Earlier research shows that target groups are mostly assumed to accept what designers and researchers offer (Peine et al., 2014; Sanders & Stappers, 2008). In our research, we recognized the importance of understanding and meeting their needs as participants who already engage in co-creation during the process of data collection. In a number of projects, we found that a deliberate choice was made not to work with patients or clients but with informal caregivers, relatives or representatives of the target group—the subject of the study was still too abstract or too hypothetical. Overall, we found that the needs of the target group can be met by identifying appealing forms of research activities that are suited to the preferences of the actors and by adopting service ways of working, mainly at the individual and interpersonal level.

THE ADDED VALUE OF COLLABORATING ACROSS BOUNDARIES

BERIT GODEROID

It took a bit to find our combination, but I think it was one of the nicest studies we did in terms of data it generated. We had to learn from each other's method and to come to a method that we were both comfortable with to run (Researcher, FoodSampler).

This chapter focuses on the manner in which partners from various backgrounds work together across disciplinary boundaries and on the benefits that such collaborations bring for a project. It investigates the following question: *how do project partners in Create-Health innovation collaborate across boundaries, and how does it add value to interdisciplinary collaboration*? It addresses the close collaborations between researchers and practice partners from the creative and the healthcare sector within the project.

As noted in the introductory chapter, intensive collaboration between different disciplines is often not without obstacles—healthcare and creative professionals come from different worlds that are not automatically aligned. This chapter describes these collaborations by focusing on boundary crossing. The term 'boundary crossing' was introduced to denote the manner in which professionals at work may need to 'enter onto territory in which we are unfamiliar and, to some significant extent therefore unqualified' (Suchman, 1994, p. 25) and 'face the challenge of negotiating and combining ingredients from different contexts to achieve hybrid situations' (Engeström et al., 1995, p. 319). It includes the efforts made by individuals or groups who occupy boundaries to establish or restore continuity in action or interaction across practices (Bakker & Akkerman, 2013).

This chapter describes the way that Create-Health collaboration took shape across disciplinary boundaries. It provides examples of boundary crossing from the 10 projects, with the objective of stimulating learning in the creative and health sectors on creative ways of working on interdisciplinary projects.

5.1 BOUNDARY-CROSSING LEARNING MECHANISMS

To investigate how partners collaborate in eHealth innovation, we adopted the theoretical lens of boundary crossing theory (Star, 1988). To analyze and structure the data from this theoretical perspective, we took four potential learning mechanisms that can unfold at boundaries: identification, coordination, reflection and transformation (Akkerman & Bakker, 2011). Akkerman and Bakker (2011) examine boundaries in education-research terms, for example by studying interdisciplinary student projects, but these learning mechanisms can also be applied to understand the development of interdisciplinarity in the Create-Health research teams. According to Akkerman and Bakker (2011, p.132) 'these mechanisms show various ways in which sociocultural differences and resulting discontinuities in action and interaction can come to function as resources for development of intersecting identities and practices.' Table 5.1 summarizes the four potential learning mechanisms by distinguishing between meaning-oriented and practice-based mechanisms. Meaning-oriented mechanisms focus on sense-making and on understanding perspectives and identities, that is, on understanding other ways of working in other disciplines, and practice-based mechanisms focus on activities, such as the integration of tools into a combined way of working (Akkerman & Bakker, 2011). The first type of mechanism includes identification and reflection, which address, among others, creating understanding and reflecting, over time, on what partners contribute and in what way. The second type of mechanism includes coordination and transformation, which address the division of tasks, learning from the other discipline and novel ways of working.

MEANING ORIENTED	PRACTICE BASED	
FOCUS ON PERSPECTIVES AND IDENTITIES	FOCUS ON ACTIVITIES	
 Identification: Diverse practices Constructing and reconstructing boundaries Explication and visibility 	 Coordination: Cooperative exchanges Transcending boundaries Smooth, effortless and routine process People or objects moving back and forth 	
Reflection:Expanding perspectivesTranscending boundariesExplication and visibility	 Transformation: Co-development of (new) practices Transcending boundaries that need to be encountered and contested Confrontation Continuous joint work 	

Table 5.1. Boundary-crossing learning mechanisms (adapted from Akkerman & Bakker, 2011).

To capture the specific meaning of each mechanism more precisely and to recognize the relevant activities more easily, we operationalized the mechanisms as processes in which the actors who are involved *identify* complementary practices and expertise, *coordinate* tasks within the structure of the consortia, *reflect* on the evolving dynamics of the collaboration, and *transform* ways of working by interdisciplinary learning. Table 5.2 shows the structure of this chapter, which follows these mechanism processes.

SECTION	SUBJECT (APPLIED TO EHEALTH INNOVATION RESEARCH PROJECTS)	BOUNDARY-CROSSING LEARNING-MECHANISM PROCESSES	
5.2	Identification of different practices in Create-Health collaboration	(Re)construct boundaries	The process of creating an under- standing of the manner in which diverse practices differ and the way in which partners with different back- grounds explicate differences and make them visible to (re)construct boundaries (identification in Akker- man & Bakker, 2011).
5.3	Coordination of tasks within research projects	Connect practices →→	How do project partners connect practices and organize smooth col- laboration between them, aiming at cooperative exchange (e.g. making use of the same data) and different practices that can generate benefits (e.g. by applying different methods of analysis; coordination in Akkerman & Bakker, 2011)?
5.4	Reflection on collaboration	Exchange evolving perspectives	Explicitly redefining and exchang- ing evolving perspectives over time and making these processes visible, resulting in expanded perspectives that inform future practice (reflection in Akkerman & Bakker, 2011).
5.5	Transformation by interdisciplinary learning	Integrate practices	Changing practices or even creat- ing new in-between practices with continuous joint work at the bound- ary that needs to be encountered and contested. Often starts with a confrontation, with a deficit or with a boundary problem (transformation in Akkerman & Bakker, 2011).

Table 5.2: Structure of chapter, in line with the four boundary-crossing learning mechanisms (adapted from Akkerman & Bakker, 2011 and Akkerman & Bruining, 2016).

The next four sections describe findings for each of the specific mechanism processes that are listed in Table 5.2, as well as contextualizing them within the relevant literature. The findings in each of the four sections describe differences and discontinuities in action and interaction between the practices that were involved in the 10 Create-Health research projects, thus identifying boundary-crossing issues and the development of intersecting identities and practices in the projects. This chapter concludes (Section 5.6) by summarizing the lessons on collaborating across boundaries that were learned from the 10 Create-Health research projects.

5.2 IDENTIFICATION OF PRACTICES IN CREATE-HEALTH COLLABORATIONS

Akkerman and Bakker (2011) define *'identification'* as coming to know what the diverse practices are about in relation to one another and as being about constructing and reconstructing boundaries. Identification mainly reflects meaning-oriented learning processes by focusing on perspectives and identities. According to Akkerman and Bakker, identification, which involves the explication and visibility of perspectives, seems *'conditional for transformation, because in the latter boundaries need to be encountered and contested before being put to use for codeveloping practices'* (2011, p.151). In the Create-Health projects, the diverse practices, as demanded by the call, originated from creative and health research organizations, representatives of the target group (e.g. patients, clients, or organizations for the elderly) and practice partners (undertakings), such as technology partners and design-practice partners. The project partners operate in the context of science and academia, in healthcare, and in technology and design, which are domains that demand interdisciplinary and cross-sectional work (Akkerman and Bakker, 2011).

(RE)CONSTRUCTING BOUNDARIES

In the 10 Create-Health projects, we observed two differences between diverse practices and our expectations: 1) there is much nuance in and overlap between research disciplines, and 2) the identification of practices takes place between and within research and practice organizations as well as within domains. Table 5.3 summarizes these findings, which we will discuss shortly by relating them to the relevant literature.

TOPIC	INSIGHT	LITERATURE	EMERGENT PRACTICE FROM THE PROJECTS OR LITERATURE
(Re)constructing consuming.	boundaries: the ident	ification of practices in Create-H	Health collaborations is time
Much nuance in and overlap between re- search disci- plines	Many researchers did not identify themselves with 'creatives' or 'health.' In many cases, it was diffi- cult to identify the origins of ways of working.	Value is always co-created, jointly and reciprocally, in interactions between providers and beneficiaries through the integration of resources and the applica- tion of competences (Vargo & Lusch,2008)	 It takes time and effort to develop shared understandings of differences in interests. The ways to explicate differences and discontinuities to cross boundaries are e.g.: Evaluation sessions, Lengthy discussions, Listing differences.
Differences in practice between and within organi- zations	Identification of practices takes place between and within research and practice organi- zations as well as within domains. Project partners ex- change knowledge but can have dif- ferent interests and approaches, and creativity can be a part of individuals, regardless of the research discipline in which they work.	Boundary-crossing learning mechanisms on the organi- zational, interpersonal and intrapersonal levels (Akker- man & Bruining, 2016) Integration of knowledge, skills and attitudes (Baart- man & De Bruin, 2011) Knowledge from academic design research projects does not always help design professionals to improve their work (Zielhuis et al., 2022)	 Identification includes an inquiry into the manner in which different practices contribute to Create-Health research collaborations. Collaboration across boundaries includes the identification of competences, capabilities and skills that individuals bring to a project.

Table 5.3: Insights on the identification of practices that are related to (re)constructing boundaries in Create-Health research collaborations. Emergent practices from the projects and references to the relevant literature are added.

MUCH NUANCE IN AND OVERLAP BETWEEN RESEARCH DISCIPLINES

The interests and approaches observed in the 10 Create-Health projects did not, strictly speaking, originate from the creative or the health sector. In fact, in many cases, it was difficult to identify their origins (see also card sort, Chapter 1). For example, we observed social-science research organizations that provided knowledge about health and research organizations, which operated between domains.

Maybe it can be explained in that way because of the university faculty, which is an engineering faculty, yet I think we don't have to label ourselves like that! I don't feel affiliated with create or health. I am a technical business consultant. I do not have hardcore knowledge of health, and I am neither a designer. I know a bit about both, and I know people in both worlds. I try to navigate between them (GOAL).

We are categorized as 'health' in this project, but the kind of research we conduct is social research. The real healthcare partner is the social housing and care organization (SQUEALED).

It is not that I am the health researcher and he is the design researcher. I am a psychologist who had nothing to do with health until I got my PhD. Since then, I conduct technology acceptance research in health context. He makes things, and I investigate whether that is accepted and used (DDD).

Researchers explicated the differences between the practices that are involved in a project. For example, 'they seek for solutions, while we would like to understand the problem' (DDD) and 'they immediately start to draw. And we, well, there are a lot of whiteboards here in the room, but nobody ever uses them' (SQUEALED). In the DDD project, the researchers listed the differences between the interests of the practices (see Table 5.4).

BEHAVIORAL RESEARCHERS INTERESTS	DESIGN RESEARCHERS INTERESTS	
Interviewing participants	Action research with participants	
Finding problems	Seeking solutions	
Understanding the present	Shaping the future	

Table 5.4 Differences between the interests of behavioral and design researchers (listed by DDD researchers).

According to the *DDD* researchers, explicating differences was necessary because of a shared desire to publish about their experiences of collaboration in interdisciplinary outlets. They had a strong desire to combine the interests of behavioral and design research. As one researcher put it, 'I feel like we are pushing each other to do that. So, not letting go to achieve that.' This process of explicating and identifying practices within projects takes time, in which researchers sometime (re)construct boundaries by resisting other practices.

'The design researcher immediately wanted to apply design activities in my field research. I am educated not to influence the field research, and I did not want participants of my field study to be influenced by design activities, in the sense that, in the journals [in which] I publish, that is not done, so I resisted for a long time' (DDD). These processes of creating understanding by (re)constructing boundaries can be related to the literature on SD logic (e.g. Vargo & Lusch., 2008), which stresses that value is always co-created, jointly and reciprocally, through the integration of resources and the application of competences (Vargo & Lusch, .2008).

We observed various researchers in health research organizations, the social sciences and information-systems engineering, all of whom conducted action research. Furthermore, in the *DDD* project, a researcher in the social sciences and a design researcher created 'kind-of personas,' and the social sciences researchers in the *SQUEALED* project created vignettes, which is akin to persona creation, commonly known as a design activity. In this project, we also observed design partners with experience in health who had a theoretical advisory role rather than a creative (design solutions) one as well as design-practice partners that delivered data analysis in the form of graphs, activities that are often not directly recognized as forms of design. These examples from practice falsify assumptions about the differences between creative and health researchers (see Chapter 1). Boundaries between the specific practices were (re)constructed in each project.

DIFFERENCES IN PRACTICE BETWEEN AND WITHIN ORGANIZATIONS

Beyond the diversity of practices between creative and health research organizations, there is also diversity of practices within organizations (between individuals) and within individuals. For example, creative methods may be applied in health and engineering research organizations.

This project was more creative because of the iterative character, visualizing things, and we conducted kind of action research, while I normally start with ready-to-test interventions (SQUEALED).

We have connections with industrial design, but we have also conducted creative activities ourselves. The creative exercise was not the same as a Van Gogh painting, but it was rather rational. The creativity includes translating how we could use creative ideas in a gamification setting (GOAL).

According to Akkerman and Bruining (2016) boundary-crossing learning mechanisms can operate on the organizational, the interpersonal and the intrapersonal levels. The organizational level includes differences in interests between organizations, for example between creative and health research organizations. The interpersonal level includes differences within a project, such as those that occur when a social sciences researcher operates within a health organization or when a technical business consultant works at an engineering faculty at a university. The intrapersonal level includes differences between people, such as those that may be observed when a non-designer engages in creative activity. Building on the work of Akkerman and Bruining (2016), Figure 5.1 illustrates how the diversity of practices within organizations, for example in the means or procedures that are employed to align distinctive participatory positions in multiple practices, can be a part of individuals.



Figure 5.1: Intrapersonal aspects (adapted from Akkerman & Bruining, 2016).

We also observed differences between practices within individual domains, for example between design practice and design research. In the Create-Health projects, academics who work in design research and professionals who work in design practice exchanged knowledge, but they did not always speak the same language: *'I know what to tell my community, to design researchers, but I'm not that sure how I can tell it to design professionals, so it becomes relevant' (FoodSampler).*

As far as the collaboration between design practice and design research is concerned, researchers and/or students conducted design activities in five projects. The other five projects included professionals from creative agencies with backgrounds in the arts and user-interface technology, graphic design and user-interface technology, and information technology (Zielhuis et al., 2022). In brief, these creative agencies offer service design and develop applications, interactive experience platforms and games. Some of the design professionals who were involved in design research realized and explained that their approach and contribution to the project differed from that of design researchers: 'This project feels very academic to me, I see this more as building up knowledge for follow-up projects than that it helps us now' (Design professional, PACO). In addition, design and health professionals explained that the real-life timeline and the academic timeline do not run in parallel. Companies asked whether they must really wait for years until they can obtain insights into the knowledge that had been generated, but the researchers pointed out that it is difficult to share articles that are not published yet. Therefore, some projects included intermediate dissemination activities, such as press releases (GOAL) as well as symposia and events that were organized during the course of the project in order to communicate evolving insights to other researchers, practice partners and (representatives of) the target group.

We found that having a design professional on board does not guarantee knowledge transfer. Knowledge from academic design research projects does not always help design

professionals to improve their work, and potential differences between academic and practice audiences in design are not always recognized (Zielhuis et al., 2022). Zielhuis and colleagues (2022) suggest that researchers and funding agencies should address impacts on design practice, articulate the specific needs of design practice, operationalize design-practice roles and reach out to a broader design-practice group from the start. Recognizing such differences can have two effects. First, the involvement of design professionals within a consortium requires the identification of appropriate activities as well as benefits that may accrue to them. Secondly, when insights are transferred to the design practice as an audience or as a target group outside of a project, it is necessary to determine what content is interesting for them and which forms are suitable for that audience.

5.3 COORDINATION OF TASKS WITHIN RESEARCH PROJECTS

The second boundary-crossing mechanism, 'coordination,' has to do with creating cooperative and routinized exchanges between practices and with transcending boundaries. In focusing on activity, it mainly reflects a practice-based learning process (Akkerman & Bakker, 2011). The Create-Health research call demanded that creative and health research organizations work together and that the main applicant be supported by one of two research organizations. The research organizations had to involve representatives of the target group (e.g. via patients, clients, or organizations of the elderly), and practice partners (undertakings) in the consortium. This consortium had to be built on (preferably pre-existing) public-private partnerships between parties with complementary expertise that would jointly contribute needed knowledge, supplemented with additional expertise where necessary or desired. The structure of the consortium had to reflect the different perspectives of the various parties, which are important for the development of the project. Those parties would jointly guarantee the input of knowledge that was required for the whole project.

CONNECTING PRACTICES

In the 10 Create-Health research projects, we report three findings on connecting practices: 1) projects differ in the coordination of a core team in one or two research organizations, 2) projects differ in the way that governance is organized (steering committees and such like) and in the frequency with which governance institutions meet, and 3) projects differ in the number of practices that they are connected with. Table 5.5 summarizes these findings, which we will discuss shortly by relating them to the relevant literature.

TOPIC	INSIGHT	LITERATURE	EMERGENT PRACTICE FROM THE PROJECTS OR LITERATURE		
Connecting prace Important for BC	CONNECTING PRACTICES: BEING LESS SHAPED BY AND/OR INVESTING TIME IN RESEARCH SEEM IMPORTANT FOR BOUNDARY SPANNING.				
Coordinating a core team in one or two research organizations	Some projects have a core team that is an- chored in creative and health research orga- nizations that connect practices from different domains. Other projects have a core team within one (creative or health) research organization. In both project struc- tures, we observed researchers in the core teams who acted as boundary spanners.	Brokers, in- termediaries, and boundary spanners (Wat- ling Neal et al., 2021) Mediating role for junior de- sign research- ers (Godfroij et al., 2022)	 PhD researchers often act as boundary spanners, as do post- doctoral researchers and assis- tant or associate professors. PhD researchers are less influ- enced by a distinctive discipline than senior researchers. Investing time in research seems important for acting as a boundary spanner. 		
The way gover- nance is orga- nized	Projects differ in ways in which governance is organized, and researchers explained their roles as coordi- nating, communicating and translating various perspectives.		 Projects are organized in different ways, e.g.: There may be a steering committee in which all disciplines are represented, Core researchers can have regular meetings about daily issues, Professors from the various disciplines can supervise the main researchers, Responsibility for coordination may be joint, One individual or partner may be responsible for all coordination, A repository may be used as a means of coordination. 		
The number of connected practices	Projects differ in the number of practices that they connect with.	RPM (Van Beest et al., 2021)	• Projects are connected to prac- tices in different ways, but all cross boundaries in the theo- retical, the conceptual, and the real-life-practice context (see Chapter 7).		

Table 5.5: Insights on the coordination of tasks, which is related to connecting practices, in Create-Health research consortia. Emergent practices from the projects and references to the relevant literature are added.

COORDINATING A CORE TEAM IN ONE OR TWO RESEARCH ORGANIZATIONS

Some projects have a core team that is anchored in creative *and* health research organizations that connect the practices of different domains through the collaboration of researchers from the different organizations. For example, *DDD* coordinated a core team of researchers in two research organizations and combined social behavioral scientists and design researchers in the core team. In this project, practices such as longitudinal field research and design research (co-design sessions) were conducted in parallel and informed each other. Researchers crossed boundaries by exchanging insights on both practices in order to keep each other in the loop and abreast of different perspectives. The other projects had a core team in one (creative *or* health) research organization and made connections to practice and research partners outside of the core team: *'We were really the core team. The supervisors and the partners around us were sparring partners' (GOAL)*. In line with the requirements of the call, all projects involved (representation from) the target group and the practice partners (undertakings) that were involved in the consortium.

In both project structures, we observed researchers in the core teams that acted as brokers or boundary crossers, building relationships between domains, supporting the knowledge-sharing process and information flows, facilitating learning, and enabling exchange between the production and use of knowledge (Watling Neal et al., 2021). In at least three of the projects (DDD, PACO and GOAL), a PhD researcher in the core team connected disciplines by, among others, working intensively together with the project partners from the other domain while being supervised and coached by project partners from their own domain (DDD) or by becoming familiar with individuals from both worlds: 'I do not have hardcore knowledge of health and I am neither a designer. I know a bit about both, and I know people in both worlds. I try to navigate between them' (GOAL). DDD researchers explained that a PhD student acted as a mediator because a postdoc and an assistant or associate professor may have been influenced more by their discipline (Godfroij et al., 2022). In other projects (e.g. SQUEALED), we observed assistant or associate professors as boundary spanners. They were responsible for the coordination of the project. The latter was described as 'a unique structure because of a concentrated core team without PhDs, with the advantage of investing your own time in the research, which makes it easier to involve practice partners like the municipality of Rotterdam' (Healthy Storytelling). Therefore, researchers on some projects explained that the prerequisite of boundary spanning is 'not being shaped that much by a distinctive discipline,' while others explained that investing time is important.

THE WAY IN WHICH GOVERNANCE IS ORGANIZED AND FREQUENCY OF MEETINGS

We observed six different ways of organizing governance. More than one structure can occur in a project. The first structure that we observed is the use of a steering committee in which all disciplines are represented. For instance, in the *DDD* project, the steering committee met every six weeks and coordinated, monitored and structured the project. The second way is to install a project team that consists of the core researchers. That team convenes regularly to discuss daily issues (e.g. *Healthy Storytelling*). The third way entails the use of a core team, with professors from the various disciplines supervising the main researchers (e.g. *GOAL*). The fourth way of organizing governance is to make researchers from both disciplines share responsibility for project management, coordination and communication (e.g. *DDD*). A fifth approach is to make one individual or partner responsible for all coordination (e.g. *SQUEALED*). This person then often acts as a boundary spanner. Finally, researchers on one project (*FoodSampler*) explicitly mentioned the use of a repository to share progress documentation and deliverables as a means of coordination.

The frequency of consortium meetings differed between projects, ranging from one meeting every six weeks (e.g. *DDD*) to one meeting per annum (e.g. *SQUEALED*). Advice could be sought from consortium members between meetings: 'She expected me to work independently, but if I got stuck, which happened recently, I went to ask her for advice,' (DDD) and 'she gave useful input at consortium meetings. In between, she was easily accessible, so we could have asked her more, but that was not necessary' (SQUEALED).

The core team drew on complementary expertise where necessary or desired between meetings: 'Everyone together, that was once a year, but in between, it went back and forth' (SQUEALED). The work on SQUEALED that took place between meetings included individual information-sharing meetings between an engineering partner, who delivered graphs of technical data, and two core-team researchers, who provided contextual information about the domestic situations of the participants (elderly individuals), which facilitated understanding and interpreting the data. The core team of researchers (who were working at the same research institute) were concerned with progression and continuing the project. They explained their roles as entailing the coordination of communication and as translating various perspectives: 'if you bring technical people and, for example, doctors together at one table, you hear very nice things and very different perspectives regarding technical possibilities. On [a] more abstract level, we researchers try to understand how things are useful' (SQUEALED).

THE NUMBER OF CONNECTED PRACTICES

All projects were connected with the practices of creative and health research, with professionals, and with (representatives of) the target group. However, some projects involved academic practice to a greater extent (e.g. *DDD* and *Track*, *Trace*, *Trigger*), while others were more closely connected to practice, health organizations, governance, design and/or development (e.g. *GOAL*, *NATALIE* and *SQUEALED*). The case descriptions in Chapter 2 reveal differences in the number of practices involved in each case. Chapter 3 describes the different practices. Despite these differences in the involvement of practices, all projects entailed collaboration in the theoretical, the conceptual and the real-life-practice context, as is evident from the research paths (Van Beest et al., 2021) of *GOAL* and *DDD* (see Figure 5.4). Researchers on the *GOAL* project conducted eight experiments with practice and involved many governance-practice partners (e.g. municipalities in the Netherlands and Belgium as well as secondary schools). The *DDD* project combined academic research from two domains and involved research practice, namely a mental health organization and a local innovation network, to conduct user research and co-design with the target group.



Figure 5.2: Various activities in the top two rows of the RPM matrix: 'understand & create' and 'explore & test.' Examples of research paths: GOAL (left) and DDD (right).

Figure 5.2 shows a simplified RPM of *GOAL* (left). Researchers on that project started their research path with 'understand and create' by performing a literature review (theoretical context), exploring different configurations for health games (conceptual context) and adding creative ideas to an existing eHealth platform called GameBus incrementally (practical context). Then, in 'explore and test,' they applied their ideas in eight experiments (practical context). Based on the findings of the experiments, they explored and tested a prototype of a toolbox called SciModeler (conceptual context) and captured the relevant insights in knowledge building blocks (theoretical context). The simplified research path of *DDD* (see the right-hand side of Figure 5.4) also started with 'understand and create,' in that a literature review (theoretical context) was followed by longitudinal interviews with individuals in their own environments (real-life-practice context) and co-design interviews and sessions that aimed to create a prototype of an eHealth tool (conceptual context). This step was followed by 'explore and test.' The researchers conducted a longitudinal field evaluation to test the prototype in the domestic setting of the target group (real-life-practice context) then wrote reports and articles in 'deliver and implement' (theoretical context). These examples show that projects are connected to practices in different ways and have different research pathways, but all cross boundaries in the theoretical, the conceptual and the real-life-practice context. Chapter 7 describes the RPM as a tool.

5.4 REFLECTION ON COLLABORATION

The mechanism of *'reflection'* has to do with expanding one's perspectives on the practices and with transcending boundaries. Like identification, it mainly reflects meaning-oriented learning processes by focusing on perspectives and identities. It involves the explication and visibility of perspectives, and it seems to be a condition for transformation (Akkerman & Bakker, 2011).

EVOLVING PERSPECTIVES

In the 10 Create-Health projects, we report one main finding about the exchange of evolving perspectives: reflecting on collaboration dynamics in research projects fosters the integration of knowledge between disciplines. Table 5.6 summarizes the finding, which we will discuss shortly in relation to the relevant literature.

TOPIC	INSIGHT	LITERATURE	EMERGENT PRACTICE FROM THE PROJECTS OR LITERATURE
REFLECTING KNOWLEDG	GON COLLABORATION: E-GENERATION PROCE	REFLECTION, WITH PROTOTYPING AS P SS, FOSTERS COLLABORATION BETWEE	art of the En disciplines.
Collab- oration dynamics	Reflecting on collaboration dy- namics in research projects fosters the integration of knowledge be- tween disciplines.	Reflection is part of the process through which transferrable knowl- edge is developed (Durrant et al., 2017). Coupling, interweaving and decou- pling dynamics (Basballe & Halskov, 2012). Understanding changing dynamics in an interdisciplinary RtD project (Godfroij et al., 2022). Ways to foster interdisciplinary col- laboration in research (Andriessen et al., 2020). Boundary objects such as design tools, maps or prototypes (Carlille, 2002; 2004; Star, 2010; Star & Griesemer 1989) can support the collaboration process (Reay et al., 2017).	 Search for integrated approach by means of reflection sessions There may be a search for additional or missing expertise in addition to the – preferably existing – partnerships between parties with complementary expertise Prototyping, as a part of the knowledge-generation process, may facilitate collaboration and support boundary crossing

Table 5.6: Insights on reflection on collaboration, which is related to evolving perspectives. Emergent practices from the projects and references to relevant literature are added.

REFLECTING ON COLLABORATION DYNAMICS

In various projects, we observed that reflection was a very time-consuming activity that involved confrontation and continuous joint work. However, it also generates surprising collaboration, which may be integrated into practice. For example, the *DDD* project provides insights into the manner in which practices from different disciplines gradually merge, transitioning from longitudinal field research that informs the design process towards an approach that integrates tools from research design. The project explicitly paid attention to the crossover between disciplines, and the fostering of knowledge integration was mentioned in the research proposal. By applying a Research through Design approach that combines retrospective and prospective analysis (Zimmerman & Forlizzi, 2014). More specifically: (a) Design activities and research in the project will take benefit of knowledge generated through conducting a technology acceptance study with people with dementia [...]; (b) Technology acceptance research will benefit from research-through-design activities.

Turning to the literature, Andriessen and colleagues (2020, p.17) found that 'strategies to foster interdisciplinary collaboration in research recommended in literature do not easily fit the unpredictability of design research projects and the complexity that comes from doing research in health practice.' By explicitly reflecting on collaboration dynamics that foster knowledge integration between disciplines, the *DDD* project provides an example of interdisciplinary collaboration in research and a detailed example of the learning mechanism that Akkerman and Bakker (2011) call 'reflection.' A sub-study within this single case yielded insights into the changing dynamics in interdisciplinary RtD, and, specifically, into the role of design researchers within the complex structure of actors who merge research and design activities across domains (Godfroij et al., 2022).

Research on the *DDD* project entailed reflection on the part of all researchers who were involved in it. They reflected on the dynamics of collaboration and found that they needed a more integrated approach, which they developed during the project, and that it was not the case that *'the psychologist tells the designer what to do, and the design researcher makes things'* (*DDD*). They reflected on the collaboration dynamics that are defined by Basballe and Halskov (2012), namely coupling, interweaving and decoupling. Basballe and Halskov (2012) define *'coupling'* as a dynamic that unites interests from various disciplines and establishes a common point of departure. They define *'interweaving'* as a dynamic where-by one activity or material informs the interests of various interests and *'decoupling'* as a dynamic that modifies foci by turning the interests of one discipline into the salient focus of the process.

There was less emphasis on reflection and collaboration dynamics in the other Create-Health projects, although we observed some dynamics, and the researchers explained them. For example, in the *FoodSampler* project, extensive attention was paid to interdisciplinary differences in approaches to analyzing research findings and means of enabling both disciplines to make use of alternative perspectives. Moreover, in most Create-Health projects, we observed that prototyping or the use of research artefacts fosters collaboration. The use of visual maps in the *DDD* project and the use of graphs and visualizations in *SQUEALED* provide two salient examples. The latter were made by a technical and a design partner of *SQUEALED* and were based on energy-consumption data that was gathered through longitudinal field research. The visual maps, graphs and visualizations fostered the integration of knowledge between domains (social sciences, technology and service design) and functioned as 'boundary objects,' a concept that Star and Griesemer (1989) developed for 'objects which are both plastic enough to adapt to local needs and the constraints of the several parties employing them, yet robust enough to maintain a common identity across sites [...]. These objects may be abstract or concrete. They have different meanings in different social worlds but their structure is common enough to more than one world to make them recognizable, a means of translation' (Star & Griesemer, 1989).

Based on conversations with design professionals, so also visually with the graphs, we developed the vignettes with input from them, based on what they found and what was possible according to them. The nice thing about working with them is that they also must ask us a lot of questions because they only see that data. Of course, they do not know the content of those meanings from the care side. So, it is precisely this exchange, even if it is about data by means of graphs, that you still get further and gain new insights (SQUEALED).

The aim of this fundamental research project was knowledge generation in the field of eHealth innovation with a view to preventing loneliness among the frail elderly and not the development of an eHealth product. In the *SQUEALED* project, prototyping was (implicitly) part of the process, which, according to Reay and colleagues (2017), facilitates collaboration and supports the boundary-crossing process. Moreover, reflection is part of the process of developing transferable knowledge (Durrant et al., 2017). In this knowledge-generation process, prototypes (artefacts and boundary objects) are research artefacts and not project goals in themselves (Godin & Zahedi, 2014). By developing knowledge building blocks, *SQUEALED* contributed to broader processes rather than just focusing on the creation of prototypes and products, which aligns with contemporary design (research) practice (Godfroij & Van der Lugt, 2020, Godfroij, 2021).

In *SQUEALED*, we also observed an evolving dynamic of coupling, interweaving and decoupling that was based on reflection during the project. When using the energy-consumption data that was collected from older individuals who were living independently, the analyses made it clear that many false notifications would be sent to family members and formal caregivers after the implementation of the system (see case portrait in Chapter 2). Therefore, the core team of researchers decided that it would not be useful to develop the eHealth system further into a product. In the last year of the project, they engaged a service design agency as a practice partner in the research process. In other projects, we also observed that additional partners had been engaged along the way and that the role of some project partners became less prominent over time. Therefore, the identification of practices (see Section 5.2) evolved during the projects as a result of reflections on collaboration and growing insights, with practices coupled, interwoven and decoupled over time.

5.5 TRANSFORMATION BY INTERDISCIPLINARY LEARNING

'Transformation,' the fourth mechanism, has to do with collaboration and the co-development of (new) practices, as well as with transcending boundaries. It involves confrontation and continuous joint work in which boundaries need to be encountered and contested. Like coordination, transformation mainly reflects a practice-based learning processes by focusing on activity, on which identification and reflection seem conditional (Akkerman & Bakker, 2011). Previously, the four learning mechanisms have been used successfully to identify the specific learning potential of various hybrid learning practices in vocational education (Bakker et al, 2016). Learning is defined as the process through which knowledge is acguired. The learning potential of an interdisciplinary research project stems from the fact that researchers can learn from another discipline and from practice partners that work at the boundary between research and practice. Engeström (1987) and Wenger (1998) stress that working at a boundary and boundary crossing engender strong learning potential. Transformation refers to ongoing two-way actions and interactions between contexts, such as the two-way knowledge transfer in the Create-Health collaborations in which research and design (both practical and academic) were merged. Most projects refer to this tendency as RtD (Stappers & Giaccardi, 2017; Zimmerman & Forlizzi, 2014). The merging of practices in the co-development of new ones is closely related to the 'interweaving' that Basballe and Halskov (2012) describe as a process in which one activity or material informs the interests of various disciplines, in which boundaries are encountered and contested.

INTEGRATING PRACTICES

Some Create-Health projects involved more transformation than others. Although the call favored pre-existing partnership consortia and even though we observed that most partnerships consisted of individuals who had worked together in the past, getting to know processes together and learning from another discipline still takes time and effort: 'We had to learn from each other's method' (FoodSampler). However, transformation, if it occurs, is inspiring, and it does generate surprising collaborations. We observed some projects in which there was intensive collaboration between disciplines and in which much (unscheduled) time was spent on sense-making and creating shared understandings. In these projects, we found transformation by interdisciplinary learning on two levels: within the projects on the organizational level and within individuals on the intrapersonal level (Akkerman & Bruining, 2016). The other projects, in which transformation occurred less, entailed information exchange between disciplines, and the project partners mainly worked in parallel on their own tasks and within their own disciplines. Table 5.7 summarizes the findings on the organizational and the intrapersonal level.
TOPIC

INSIGHT

LITERATURE

EMERGENT PRACTICE FROM THE PROJECTS OR LITERATURE

INTEGRATING PRACTICES: INTERDISCIPLINARY LEARNING ACROSS BOUNDARIES REQUIRES AN OPEN MINDSET TO LEARN FROM THE OTHER S KNOWLEDGE AND SKILLS.

Transforma- tion on the or- ganizational/ project level	In interdisciplinary research collabora- tions, it takes time and effort to get to know the process to- gether (also in existing partnerships) and to learn from the other discipline. However, transformation, if it occurs, is inspiring and generates surpris- ing collaborations.	Interdisciplinary RtD is a complex collaboration pro- cess (Godfroij et al., 2022). Design activities play a formative role in the knowl- edge-generation process (Stappers & Giaccardi, 2017). Transferring, translating and transforming to manage knowledge across boundar- ies (Carlille, 2002, 2004).	 Researchers from the other discipline may not feel comfortable in applying methods in a specific way. It may be difficult to find publication outlets or to identify interdisciplinary out- lets that can be approached. Pushing each other may be helpful. Codeveloping (new) interdisciplinary practices is a complex process.
Transforma- tion on the interpersonal level	In transformation on the individual level, we observed open mindsets, openness to learning, enthusiasm and changes to ways of working, such as being more systemat- ic or more creative.	Transformative learning re- quires critical (self-)reflection and openness to change (Baartman & De Bruin, 2011). Value is perceived experi- entially and differently by different actors (Vargo et al., 2020).	• Transformative learn- ing in interdisciplinary collab- oration across boundaries de- mands that knowledge, skills and attitudes be integrated, which can produce a valuable collaboration in which value is perceived experientially and differently by different actors.

Table 5.7: Insights on transformation by interdisciplinary learning, which is related to integrating practices. Emergent practices from the projects and references to the relevant literature are added.

TRANSFORMATION ON THE ORGANIZATIONAL OR PROJECT LEVEL

We observed organizational transformation in, for example, the *FoodSampler*, *DDD* and *PACO* projects. An example that illustrates the merging of activities that are conducted by creative and health research organizations that work together and in which the dietary expertise (health research) of the University of Applied Sciences and the contextual inquiry (creative research) of a technical university were deployed to set up and conduct interviews jointly. Finding a method that suited both parties demanded considerable effort.

Their understanding of running interviews was very different from ours. For them, an interview was more a matter of closed questions, very structured, so they could compare the different participants. We work with contextual interviews, cameras and pencils so they can map things. It took a bit to find our combination, but I think it was one of the nicest studies we did in terms of [the] data it generated. We had to learn from each other's method and to come to a method that we were both comfortable with to run. The main issue was not [that] they did not believe in what we were doing but [that] they did not feel comfortable running something like that (FoodSampler).

In this project, boundaries were encountered and contested, but codeveloping a method that suited both parties was also worthwhile: 'It took a bit to find our combination [...] it was one of the nicest studies we did.' Furthermore, this example emphasizes learning potential: 'We had to learn from each other's method. The lead researcher in FoodSampler explained this learning mechanism as follows:

Knowledge, that I learned a lot from them. In every meeting, I was like 'oh my god, that makes sense.' You start putting things together. So, that is knowledge. Of course, I knew with whom I was working. So, I knew they were more systematic in how they do their research. They do more extensive analyses and literature reviews. All these things, they are very strict into that. So, that I knew, too, but it was also kind of nice to collaborate with that kind of different approaches. (FoodSampler)

The researchers on this project also conducted analysis sessions together. In those sessions, they had to integrate different data analysis practices.

We have a clear idea about the knowledge we want to gather. We would also let ourselves a bit open to see what other things were interesting within that path. So, I really highlight it, but the fact that I could work with health care researchers and practitioners... For them, in the beginning, it was a bit like... They couldn't really grasp it. They were like, 'just set up an experiment, run the experiment and that's it.' Not much qualitative approach. But they were also so open because they also saw the need... that they need to go beyond that. So, the ways of working were quite open to explore this. They were fascinated! Every time I organize a session here to run qualitative analysis, like wall analysis or thematic analysis, they would just enjoy, or more than that. They were taking pictures of themselves with all the post-its on the wall to send to their colleagues. They would be like 'ah, there is another day of this' (FoodSampler).

Another example that illustrates the merging of activities that are conducted by creative and health research organizations that work together originates from the *DDD* project. In that project, the lengthy discussions and reflections in the evaluation sessions included confrontations but also many occasions on which researchers from both disciplines were 'pleasantly surprised and inspired' (DDD). This project aimed to develop a (new) method, longitudinal co-design, which turned out to be very difficult and was a part of a larger process. The researchers in this project concluded that 'longitudinal co-design is complex and research findings need to be more fine-grained' (DDD). However, the project encountered and contested boundaries successfully, and ways of working wound up interwoven. The lead researcher made the following observation:

I resisted for a long time to apply design tools within my field research, but insights have become increasingly sharp, also about what we cannot know. For example, whether a lack of initiative is a problem for people with dementia or whether they may not want to take the initiative at all... So, that's why those design methods are so good for getting ahead (DDD).

Thus, the (re)construction of boundaries ('I did not want participants of my field study to be influenced by design activities') as described in the section on identification (see Section 5.2) were transcended after lengthy discussions and reflections within research teams, resulting into the integration of practices: 'An integrated work package WP1.5 that combines the work packages 1 and 2' (DDD). The lead researcher on DDD stopped 'chopping in the sand' because they became aware that they wanted to learn from the other discipline and that they 'will see where to publish articles.'

The researchers on the *PACO* project also explained that they had contributed to method development by exploring steps in collaboration with other disciplines. In other projects, transformation, interweaving and the co-development of new practices were less dominant, although we observed that researchers engaged in a time-consuming sense-making process in various projects. They experienced it as a valuable process that they had not foreseen. The researchers were happy to learn from other disciplines, regardless of the time it took. For example, 'It was one of the nicest studies we did' (FoodSampler) and 'it generated many occasions [on] which researchers from different disciplines were pleasantly surprised and inspired' (DDD). Their experiences, which they explained, indicated that intensive collaboration is needed to produce outputs for interdisciplinary outlets. For example, publishing in domain-specific journals was difficult because 'the journals are nicely multi-disciplinary for psychological terms, but not that creative, like "let's all start using design methods"' (DDD). The researchers realized that they really needed to collaborate to reach interdisciplinary outlets: 'I feel like we are pushing each other to do that. So, not letting go to achieve that' (DDD).

TRANSFORMATION ON THE INTRAPERSONAL LEVEL

On the individual level, we observed that some of the researchers became more open minded in their attitude towards research and that they now work more systematically or more creatively. For example, in the DDD project, the lead researcher changed their attitude: 'Over time, [the design researcher] convinced me to apply creative methods in my field [of] research, and I just wanted to do what seems good to me and to learn from a different approach' (DDD). The members of the FoodSampler consortium already had an wscenario when they collaborated: 'We engaged even more happily in it' (Researcher, FoodSampler). In this project, the lead researcher explained that they had a mindset that was very open to collaborating and to learning from each other: 'We would also let ourselves a bit open to see what other things were interested within that path [...] they were also so open because they also saw the need... that they need to go beyond' (FoodSampler).

Furthermore, the researchers on some projects explained that they had started to work more iteratively and creatively or that they had become more systematic.

I have started to work in a much more structured way. I tried to develop myself on that because if you're not on the same wavelength from the start, it stops right away' (GOAL).

The data-management plan, that is my biggest learning. I don't start any project without it anymore, and we really use it as a live document. I think [it] is a nice learning. I sort of learned from [the research partners] but also, because of the demands of the project... that we had to really move in that direction. There is a lot of structure in [the] ways they sort their data, and also, because of the project, because we were asked to use the more fair principles and to publish all these data sets... The combination of these two things, I think that is my biggest learned, and I am very proud of it now. I am super-structured in how I collect my data, and I also try to publish it in that way (FoodSampler).

In addition, the enthusiasm of the individual actors who were involved in the projects and learning potential were important for interdisciplinary collaboration: 'Everybody is super-enthusiastic about the project and that is noticeable in the collaboration and during meetings. Everybody puts in all the time that is needed' (SQUEALED).

Therefore, on the intrapersonal level, we observed transformation processes with an open mindset and a learning attitude. Baartman and De Bruin (2011) suggest that a learning process is transformative when it requires critical (self-)reflection and openness to change. They refer to a learning process that is directed towards the integration of knowledge, skills and attitudes. Building on Chapter 4, in SD logic, service is a reciprocal process between actors, not an intangible product. For example, value creation and interweaving may emerge from

a data analysis session that informs researchers from various disciplines about their research interests. In this analysis session, design researchers may be interested in the thematic analysis on the wall, while medical and social-science researchers may be interested in coding data fragments. The two are not mutually exclusive: the analysis session, as an activity, and transcripts or other data sources, as material, can be used to inform various interests, and researchers can apply different competences (knowledge, skills and attitudes) to create value for everyone. Create-Health researchers may learn from each other's knowledge, skills and methods in such analysis sessions, which can produce a valuable collaboration in which 'value is perceived experientially and differently by diverse actors' (Vargo et al., 2020, p.17).

5.6 CONCLUSIONS: COLLABORATION WITH INTERSECTING IDENTITIES AND PRACTICES IN EHEALTH INNOVATION

This chapter provides insights into the learning mechanisms of boundary crossing in Create-Health collaborations by answering the following question: *how do project partners in Create-Health innovation collaborate across boundaries, and how does it add value to interdisciplinary collaboration*? The Create-Health call focused on research collaborations between creative and health research organizations that involve representation from the target group (e.g. organizations of the elderly) and practice partners (undertakings) in the consortium. The findings improve the understanding of the processes of (re)constructing boundaries, connecting practices, exchanging evolving perspectives and integrating practices.

In order to explain how the project partners in Create-Health innovations collaborate across boundaries, we must first report the finding that the process of creating shared understandings about the differences between diverse practices is time consuming. We found many nuances in and overlaps between research disciplines with different interests on the organizational, the interpersonal and the intrapersonal level as well as between practice and research. Boundaries between the specific practices were (re)constructed for each project, for example in evaluation sessions and lengthy discussions as well as by listing differences. Recognizing differences has an impact on the involvement of various professionals from practice and research and in transferring insights to creative and health practice.

Secondly, we found that PhD and postdoctoral researchers play a crucial role in organizing and smoothening collaboration between practices. Being less influenced by a distinct discipline than senior researchers and investing much time in the research project seem to be prerequisites for boundary spanning. We observed researchers who acted as boundary spanners, brokers or mediators, both in projects in which a core team was coordinated by one research organization and in projects that required two research organizations to coordinate a core team. Projects differ in the ways in which governance is organized, and the researchers explained their roles as coordinating, communicating and translating various perspectives. The examples of forms of governance include steering committee in which all disciplines are represented, daily meetings between core researchers, professors from various disciplines supervising the main researchers, joint responsibility for coordination, entrusting one individual or partner with all coordination, and/or the use of repositories as a means of coordination.

Thirdly, we found that reflection fosters knowledge integration between disciplines and that (the identification of) the partners that are involved in the project may change over time because of evolving perspectives and emerging insights. Practices couple, interweave and decouple over time, based on reflection, and there can be a search for an integrated approach and/or additional or missing expertise. Reflection on collaboration dynamics, prototyping or the use of research artefacts may facilitate the knowledge-generation and boundary-crossing processes.

Fourthly, we observed that, even in pre-existing partnerships between individuals who knew each other had worked together in the past, getting to know the process of Create-Health innovation together, learning from the other discipline and integrating practices in combined or new ways of working still required time and effort. Not all researchers were able to work continuously and jointly at the boundaries of change or to create (new) practices together. However, in some projects, we observed intensive collaboration between disciplines, with much unforeseen time being spent on sense-making and creating shared understandings. In this process, we found transformation on two levels: within the project on the organizational level, for example in the form of an integrated work package or an attempt to become comfortable with combined ways of working, and within people on the intrapersonal level, in forms such as changes in mindset, openness to learning from another discipline and beginning to work more systematically or creatively.

To answer the question of how Create-Health collaboration across boundaries adds value to interdisciplinary research projects, we report two main findings that concern the importance of 1) learning potential and 2) the ability to generate and communicate knowledge. Learning potential demands an open mindset, enthusiasm and inspiration. Researchers explained that they wanted to learn from different approaches and had to study the methods of others. Being able to generate and communicate knowledge requires reflection on collaboration dynamics as well as coupling, interweaving and decoupling practices that are based on growing insight. Intensive collaboration across boundaries creates value because interdisciplinary outlets that transcend specific domains can be reached. Domain-specific journals are not always open to creative research methodologies, and integrated methods are not always accepted. Therefore, researchers must work together to connect and integrate practices as well as to engage in dissemination activities and to publish works.

Despite the finding that many aspects of collaborating across boundaries are unforeseeably time consuming, all that time and effort are worthwhile! Researchers explained that the study was among the most pleasant that they had participated in and that it had resulted in surprising collaborations.

CONCLUSIONS

DAAN ANDRIESSEN

For over three years, we followed 10 research projects in which researchers and practitioners from the creative and the health sector combined forces to develop knowledge building blocks for eHealth applications. We wanted to learn how the Create-Health ways of working that were employed within these research projects contributed to innovation, research processes and the endeavors of the parties that were involved. This chapter summarizes our main findings on three topics: the difficulty of defining 'creative ways of working,' the many ways of creating impact for the public and the importance of recognizing and using the unique contribution and way of working of each partner. Based on these insights, we provide recommendations for researchers and funding organizations on getting the best out of Create-Health collaborations in research (Section 6.4).

When we started our research, the funding agency invited us to look specifically at 'creative ways of working.' However, 'creative ways of working' are difficult to define, an important insight. In the 10 projects, we found many ways of working that, in one way or another, can be described as 'creative,' but they do not necessarily have one common characteristic. Therefore, there does not seem to be an intrinsic defining characteristic of 'creative ways of working.' In addition, it is also not possible to associate a way of working with a person's background. We did not find typical creative or healthcare ways of working. Many participants in the 10 projects were from mixed backgrounds and could not be classified easily as creative researchers, professional or health researchers or as professionals (Zielhuis et al., 2020).

At the same time, we find that collaboration between parties who work in the creative and health sectors does produce some interesting and maybe even unique approaches to innovation and research in which ways of working are 'mixed.' Therefore, instead of trying to identify typical creative ways of working, we opted to study the ways of working within the projects as Create-Health ways of working. This should not be taken to imply that there are no differences between the creative domain and the health domain.

Blandford and colleagues (2018)and developing and evaluating each intervention. Two of the central areas of expertise required are Health (broadly defined pinpoint several differences, drawing on their personal experiences of deploying effective interactive digital-health interventions. For example, they state that development processes in health tend to be viewed as linear and evidence based, while in human-computer interaction and software engineering, they tend to be much more iterative and focused on fitness for purpose. However, in the projects that we studied, this boundary was not so clear cut. In all 10 Create-Health projects, the development process was iterative, even when it was not a designer but a health researcher or a behavioral scientist who had taken the lead. In all projects, there was also an emphasis on creating evidence and using theory. Therefore, although differences in methodological approach exist, when creatives and the health sector work together to create knowledge building blocks, they adopt a common approach with elements from both domains, resulting in Create-Health ways of working.

Combining elements from both domains is not always easy. As we reported elsewhere (Zielhuis et al., 2020), such combinations can be challenging. For example, the open approach that is often required in design processes might interfere with the need to submit clear plans to ethics committees. At the same time, combining elements may lead to a best-ofboth-worlds approach, for example when a survey study and an explorative co-design study are combined to create personas.

Our research question is as follows: what ways of working are employed within research projects on eHealth innovations in which research and practice partners from both the creative and the health sector collaborate, and how do stakeholders benefit from these ways of working? We answer this question by highlighting three aspects of Create-Health ways of working that together provide a comprehensive overview of the main characteristics of the 10 projects:

- The use of specific ways of doing research and innovation,
- The way beneficiaries and other actors who are involved are included in research activities, and
- The way collaboration between creative and health actors takes shape.

We summarize our main findings on all three aspects.

6.1 CREATE-HEALTH: WAYS TO INNOVATE AND CONDUCT RESEARCH

Several practices and worlds come together in research projects on eHealth such as those that are presented in this book. Since various challenges can arise when an attempt is made to integrate the ways of working that predominate in these worlds (Blandford et al, 2018; Groeneveld et al., 2019), researchers need more guidance on operationalizing their integration. We found several ways in which integration can be improved, and we discovered that this diversity of ways of working can be organized along the following dimensions:

- the goals of a project, which are directed towards meaningful eHealth innovation, and, in light of these goals:
- the mix of methods
- the role of prototypes
- the role of iterations.

What are the goals of the projects? The Create-Health call from ZonMw strongly focuses on theory development. Since earlier programs apparently focused too much on delivering solutions for practice, projects were sought that would produce *knowledge building blocks* in order to contribute to solving problems in healthcare practice. Many of the researchers who were involved welcomed this focus on theory development as an alternative to *jumping to solutions*. Design practice partners, as well as care organizations and care professionals, understand the need for an evidence-based approach to grounding healthcare innovations. However, most projects aim for a combination of generic knowledge and relevance to practice.

Knowledge outcomes can have three different objects: knowledge about a phenomenon, knowledge about design approaches or knowledge about solutions. They can also have different orientations—knowledge about how things are now as opposed to knowledge about possible futures. The various project partners within a project sometimes display a different interest in these various knowledge outcomes. Since most projects aim to impact multiple audiences, it is difficult to combine the different interests of those audiences, and this can complicate matters. Although this issue was resolved successfully in the 10 projects, we recommend discussing the various interests of the partners in relation to the goals of the research at the stage at which a consortium is formed. The tool in Chapter 7, which is based on the RPM, can help structure this conversation.

How do the projects combine methods? All 10 projects combine methods from different disciplines, but they do so in various ways. We identified four such ways. In a sequential approach, methods are combined sequentially. This approach can be recognized as multidisciplinary (Choi & Pak, 2006)interdisciplinary and transdisciplinary teams are increasingly encouraged in health research, services, education and policy. This paper is the second in a series. The first discussed the definitions, objectives, and evidence of effectiveness of multiple disciplinary teamwork. This paper continues to examine the promotors, barriers, and ways to enhance such teamwork. Methods. The paper is a literature review based on Google and MEDLINE (1982-2007. For example, an iterative process of developing a proto-type by using qualitative evaluations may precede a quantitative effect study. Both types of methods benefit the project. The challenge in this approach is the quality of the handover—each party needs to understand the input of the other.

Different methods are combined in strategies for *integrating the results* of different activities into joint results. This approach can be recognized as *interdisciplinary* (Choi & Pak, 2006) interdisciplinary and transdisciplinary teams are increasingly encouraged in health research, services, education and policy. This paper is the second in a series. The first discussed the definitions, objectives, and evidence of effectiveness of multiple disciplinary teamwork. This paper continues to examine the promotors, barriers, and ways to enhance such teamwork. Methods. The paper is a literature review based on Google and MEDLINE (1982-2007. For example, the outcomes of a systematic literature review may be combined with the results from context mapping (Sleeswijk Visser et al., 2005). The challenge is to ground a design process in previously developed models and theories *while* keeping an open mind and learning from users.

In a strategy that *integrates expertise* into methods, the method that is used requires expertise and inputs from different disciplines. This approach can be recognized as *trans-disciplinary* (Choi & Pak, 2006)interdisciplinary and transdisciplinary teams are increasingly encouraged in health research, services, education and policy. This paper is the second in a series. The first discussed the definitions, objectives, and evidence of effectiveness of multiple disciplinary teamwork. This paper continues to examine the promotors, barriers, and ways to enhance such teamwork. Methods. The paper is a literature review based on Google and MEDLINE (1982-2007. For example, the creative partners may introduce user research as a method, and the health partners can provide access to the target group and domain expertise. The challenge is to find a method that suits both parties.

A *personal approach* involves an N=1 (or N=low) study, which is a common method in both domains. Instances in which a first prototype is tested on one older adult provide a salient example. This personal approach helps frame the research for various domains. Because experts in both domains are familiar with this approach, using it in Create-Health projects is unproblematic.

We found combinations of these four strategies in all of the projects. This implies that a Create-Health project can sometimes be a multidisciplinary study and sometimes a transdisciplinary one, depending on the phase of the project and the objective of that phase. Therefore, Choi and Pak's (2006) well-known distinction between multidisciplinary, interdisciplinary and transdisciplinary research is not as clear cut as it seems. It is important that the research partners discuss various methods, particularly methods from the design domain, which are not clearly defined (Sanders, 2008). It is also necessary to discuss which level of disciplinarity is necessary for particular activities. This helps to understand what the activities entail, and it sheds light on the means of combining the various methods.

Publishing studies that are based on these mixed-methods approaches to research can be challenging, especially when the researchers wish to reach an audience beyond the design field, for instance by publishing in health journals. The co-design approach, in particular, can be difficult to frame in a manner that is appropriate for such journals.

What is the role of prototypes in the 10 projects? We found that all projects worked with some sort of prototype. This is a characteristic element of Create-Health ways of working. Making tangible things can facilitate shared understanding considerably. However, the manner in which prototypes are used can lead to confusion. Partners from other domains might think of prototypes only as tools that will be developed further into products. This is the prototype as a *product as a solution*. However, prototypes also have other functions. Some prototypes are used mainly to elicit responses and interactions. This is the prototype as a *research artefact*. In this program, we found several examples of such research artefact, such as *provotypes*, which are intended to provoke (Mogensen, 1992). Finally, we saw several examples of prototypes that are *technology sketches* or proofs of concept. A single project can contain prototypes with different roles. To complicate matters, some prototypes start out as research artefacts but end up as products as solutions.

It is important to discuss the purpose of the prototypes in a project, especially with the practice partners but also with the funding agency. Will a care institution end up with a working solution for their local situation? Can they keep the high-fidelity prototype that their clients love? Will the technology be supported in the long run, or will the research artefact have to be returned once the study concludes? Do the researchers intend to develop a

prototype further after the project, and how much development will this take? Alternatively, is the prototype meant to describe the design case as a demonstrator that illustrates a theory?

What is the role of iteration in the 10 projects? Although iteration is present in each of the 10 projects, we found two different ways in which it took shape. This has to do with the goal of the project and the role of the prototype. Projects that are aimed at developing an eHealth prototype result in work on a product-as-a-solution prototype and use iteration to integrate the insights from user research and scientific evidence into the product. We find that this approach is particularly popular in projects in which researchers from healthcare or the behavioral sciences take the lead and try to develop evidence-based interventions. In projects that are aimed at developing theory, design cases are used as iterations or loops through which a theoretical framework is gradually developed. This type of iterative approach is used in projects in which design researchers take the lead. These projects are particularly emergent and difficult to plan.

Whether the focus of a project is on developing a working prototype or on developing a theoretical framework, the research pathway never proceeds from theory to practice in a linear fashion. There is a constant process of hopping between the theoretical, the conceptual and the practical context. The theoretical context is used to ground a prototype in theory, and the practical context is used to ground it in user experience. However, combining both types of grounding in a single project is difficult within a three-year timeframe. Additional research into the effectiveness of the prototype is always needed.

It is important to consider the point at which a research path ends and the steps that need to be taken to produce a result that can be used in practice. Iterating towards an evidence-based prototype often requires further validation but also a practice partner that can facilitate its financial, organizational and technical implementation. When iterating towards theory, it is often necessary to translate theoretical insights into guidelines or insights that practice partners can use. It is important to discuss what the research pathway should look like, what the role of all partners in the various research steps will be and what the final destination of the research path will be with potential consortium partners. Clarity about the research pathway is needed to manage expectations and to prevent disappointment along the way. The RPM tool that is described in Chapter 7 can facilitate such conversations.

To summarize, we found several means of combining ways of working in Create-Health collaborations. We provided practical guidance for dealing with the challenges of such collaborations. It is also important that research partners make sure that they involve the right partners (e.g. professional designers), who must possess appropriate expertise on particular ways of working, in order to be successful. We have highlighted several ways of attaining this objective.

6.2 CREATE-HEALTH: WAYS TO INVOLVE INDIVIDUALS IN RESEARCH PROJECTS ON EHEALTH

The creative sector has a long tradition of *co-design*: involving end users and stakeholders in design processes. The health sector has developed a tradition of *patient participation* in research. According to Blandford and colleagues (2018)and developing and evaluating each intervention. Two of the central areas of expertise required are Health (broadly defined, there is an important difference between user involvement in the creative domain and user involvement in the health domain. In the creative sector, researchers are trained to focus on the user, reflecting the presupposition that the user is an expert on their own life. In the health sector, researchers who are developing an intervention start with existing knowledge and expertise, and their task is to get the user to engage, adhere and comply.

Because of these differences, we investigated the involvement of end users and stakeholders in the 10 Create-Health projects. Brett and colleagues (2014) show that the participation of patients in research is conducive to the attainment of user-focused research objectives, the development of user-relevant research questions, the identification of more appropriate recruitment strategies for studies, the generation of consumer-focused interpretations of data and to improvements in the implementation and dissemination of study results. In light of these potential benefits of end-user and stakeholder involvement in research, one of the terms of the Create-Health call referred to the inclusion of four types of stakeholders in each of the 10 projects: research partners from the healthcare domain, research partners from the design domain, creative partners (at least one undertaking) and healthcare or welfare organizations. Another condition was to involve end users and health professionals in the formation and execution of the research. In our research, we look at the benefits of end-user and stakeholder involvement that accrue to the participants. What value do they get out of participating in the projects?

An important finding is that all projects encountered difficulties with recruiting and involving members of the target group, namely overweight individuals (especially with a lower SES),

individuals with dementia or individuals who experience loneliness. It is difficult to find and contact such individuals and to keep them engaged during the study. In addition, researchers on some projects had to change their approaches during the COVID-19 lockdown. We identified several ways to find members of the target group during social distancing (Godfroij et al., 2020). The following approaches may also be valid under ordinary circumstances.

- The target group can be expanded to spare the vulnerable.
- Co-design sessions can be moved from a lab setting to the domestic setting of the target group.
- Small-scale research (sometimes even N=1) in homes can provide rich insights. Generally, individuals feel more comfortable sharing their experiences at home.
- Online contact with the target group (via video calls) increases flexibility and can be alternated with home visits and/or telephone contact.
- It helps to involve the target group remotely to capitalize on existing resources and technology with which they are already familiar.
- It is possible to teach members of the target group how to deal with technology, for example by sending out manuals or by calling them and making the digital transition together with them. It is helpful to inform the members of the target group of the way in which the session works beforehand.

How are the target group and other stakeholders involved in the 10 projects? We distinguish between involvement in the theoretical context, involvement in the conceptual context and involvement in the practice context of the research. We find that the involvement of members of the target group and other stakeholders in activities in the *theoretical context*, such as desk research, lab experiments or the publication of scientific results, was limited. These activities are mainly conducted by researchers. Furthermore, lab experiments often take place in the early stages of research, and what is tested is often still very abstract or too technically underdeveloped to be tested directly with the target group. Consequently, a more neutral audience (such as students) is often used. Furthermore, ethical considerations, such as the need to avoid placing too heavy a burden on participants, are taken into account when a decision is made not to involve the target group directly.

Creative partners, healthcare organizations, healthcare professionals and the target group (with their relatives and/or informal caregivers) are much more involved in the *conceptual context*. They help to create the basic concepts for prototypes by participating in interviews, co-design sessions, visualization routines, focus groups and such like. When it comes to exploring and testing a prototype, healthcare partners are more involved than creative partners. As a result, in most projects, the creative partners had little interaction with health partners. The exploration and testing of a prototype are usually conducted through workshops with care professionals or the target group, usability studies with the target group, mock-up studies and so on. The involvement of the target group and their relatives in the conceptual context is mixed, ranging from the more passive roles of listeners or co-thinkers to the more active ones of partners who execute a part of the research. In this context, healthcare professionals are mostly involved as facilitators of the target group or as co-thinkers in interviews or focus groups.

As can be expected, the involvement of the target group and healthcare partners is even more intensive when the research projects enter the *practice context*, that is, when an attempt is made to understand the practice situation and to test prototypes in a real-life setting. All but one project included these groups in testing in the real-life-practice context. Here, too, ethical considerations play an important role. When there is no tangible prototype to test, one needs to consider whether it is ethical to involve a vulnerable target group in testing. There is also the risk that the target group and health professionals expect the hifi prototype to become available as a commercial product, while that might not be the case. This tendency requires expectations to be managed carefully. Involving a target group of individuals who are no longer completely independent poses further challenges. Often, the presence of another individual is necessary, but it may cause participants to give responses that they believe to be socially desirable.

How do the participants from the target group benefit from the projects? It is clear that the researchers and the research projects benefit from the participation of the target group in terms of data gathering and the validation of prototypes. We find that researchers on several projects also tried to make sure that the participants would benefit from participating in the research. The examples of such activities include conducting several interviews over a longer period of time in order to build a relationship with individuals with dementia, inviting the elderly for coffee breaks or adopting a personal approach to interviews with individuals who are experiencing loneliness. These are examples of two-way value exchanges between researchers and members of the target group.

How do the health sector partners benefit from the research? The health sector is under enormous pressure, which increased considerably when the pandemic started. This pressure makes it even more difficult to involve health professionals in activities that are not directly related to healthcare practice. This said, health professionals benefitted directly from their participation in several projects, especially when the research entailed the introduction of a way of working that complemented their daily practice, such as workshops for the elderly and their relatives. However, these benefits mainly accrue at the personal level, and new ways of working were not implemented systematically at the organizations.

How do the other creative sector partners benefit from the research? Creative partners

are mostly involved in the development of prototypes. Most creative partners invest in the design and development of a prototype with an eye on potential tangible outcomes in the future. There is no immediate return in the short term. In part, this is a consequence of the call and its focus on the development of knowledge building blocks.

To summarize, our research shows that the value of participation differs between the various stakeholder groups. For the beneficiary stakeholders, value is found in the research activities that occur during the project. For social stakeholders, the value lies in the integration of activities into the daily healthcare process during the project. Economic stakeholders derive value from the long-term development of eHealth applications. It is important that researchers think about the ways in which participating parties can benefit from the research while it is ongoing and not only from potential future products, especially in fundamental research projects such as the ones that are presented in this book.

Our research refines the idea of patient participation in research. The projects show that increasing target-group participation in all phases of research is not always needed or desirable. We find that the researchers on a number of projects deliberately chose not to work with patients or clients but instead focused on informal care givers, relatives or students as future beneficiary actors. Our research also refines the idea that creative ways of working always entail co-designing solutions together with the target group. Members of the target group were involved in the 10 projects, but not in every stage of the research and not always as partners or decision-makers. The reasons are often valid. Researchers who coordinate research projects on eHealth should ask themselves who they would like to involve as well as when, how and why they should involve specific actors from the three actor groups (beneficiary actors, social actors and economic actors). The decision process should not focus only on methodological considerations—ethical issues and the question of how a collaboration can be turned into a win-win situation should also be taken into account. Thinking about the research project as a process of mutual value exchange can help. In the Research Pathway Tool that is presented in Chapter 7, we pay special attention to the roles of the various partners and to the services that can be exchanged during a project.

6.3 CREATE-HEALTH: WAYS OF COLLABORATING

Intensive collaboration between different disciplines is often not without obstacles, and healthcare and creative professionals come from different worlds that do not automatically align. Partners from different fields participated in the 10 projects, as depicted in Figure 6.1. In our research, we looked at the manner in which this collaboration unfolded, especially the ways in which actors 1) created understandings of the differences between diverse

practices, 2) connected practices and organized smooth collaboration between practices, 3) explicitly redefined and exchanged evolving perspectives over time and made these processes visible and 4) changed practices or even created new in-between ones through continuous joint work at the boundary.



Figure 6.1: When creative and health sectors unite

How do the various actors in the 10 projects come to know what the diverse practices are and how they differ? First, we find that many researchers do not identify strictly with a single domain and that creativity can also be a part of individuals, regardless of the research discipline in which they work. It is important to identify the competences, capabilities and skills that individuals possess, irrespective of their formal position. That is not to say there are no boundaries between disciplines. It does take time to get to know the ways of working of others. Moreover, different partners may have different interests or points of view. For example, companies do not want to wait for access to knowledge, but researchers do not want to share articles that are not published yet. Design researchers want participants in field studies to take part in design activities, while behavioral researchers do not want them to be influenced by such studies because that influence may reduce the validity of measurements.

How do the 10 projects create cooperative and routinized exchanges between practices and transcend boundaries? The first element of coordination entails assigning the coordinating role. In most projects, coordination was conducted by a researcher from the research organization of the main applicant. Some projects aimed for shared project management that would be anchored at two research organizations in order to equalize contributions

and to stimulate boundary crossing. The position of the coordinator is another element of coordination. In many cases, the coordinator was a PhD student. Some of those students deliberately tried to act as boundary brokers by moving physically between two institutions. In other cases, the coordinators were associate professors, and this allowed them to spend more time on the project then they would if one of their PhDs had taken the lead. The way in which the governance of the projects is organized constitutes the third element of coordination. Six different ways of organizing were used, and some projects applied more than one. The first is the use of a steering committee in which all disciplines are represented. The steering committee meets at key moments and coordinates, monitors and structures the project. The second way is to establish a project team that consists of core researchers who hold regular meetings about daily issues. The third way is to use a core team of professors from various disciplines who supervise the main researchers. The fourth organizational approach is to share responsibility for project management, coordination and communication between researchers from both disciplines. A fifth way is to make one individual or partner responsible for all coordination. This person then often acts as a boundary broker. Finally, one project explicitly mentions the use of a repository to share progress documentation and deliverables as a means of coordination.

How do actors in the 10 projects redefine their perspectives during the collaboration? Collaborating in Create-Health projects is not 'the health researcher [telling] the designer what to do, and the design researcher [making] things.' We observe an evolving dynamic of coupling, interweaving and decoupling within the projects. For example, in most projects, we observe that additional partners were engaged along the way and that the role of other project partners would become less prominent over time. The reasons included a need for additional expertise or a realization that it was not useful to develop an eHealth tool further, which rendered the technical practice partner obsolete. It is desirable to realize that there may be a search for additional or missing expertise that ought to extend beyond the (preferably pre-existing) partnerships between parties with complementary expertise that were sought in the call.

Researchers on one of the Create-Health projects explicitly reflected on collaboration dynamics. They found that they needed a more integrated approach, which they developed during the project. Explicitly reflecting on collaboration dynamics to foster knowledge integration between disciplines is an appropriate means of optimizing interdisciplinary collaboration in research, especially since the strategies that are recommended in the literature do not easily fit the unpredictability of design research projects and the complexity that results from conducting research on health practice.

Beyond reflection on the knowledge-integration process, we also find that prototypes are used as boundary objects to bridge gaps in understanding between disciplines. These objects can include graphs and visualizations as well as physical prototypes. Creating an object to which various disciplines can attach meaning helps to create common understanding and to foster knowledge integration between domains.

How do the 10 projects co-develop new practices and transcend boundaries? The co-development of practices and the overcoming of boundaries involves confrontation and continuous joint work in which said boundaries must be encountered and contested. This transformation occurs more frequently in some projects than it does in others. We observe transformation on two levels: within the projects on the organizational level and within people on the intrapersonal level. On the organizational level, transformation occurs when two parties from different backgrounds work together to find a method for data gathering that suits both. Although boundaries are encountered and contested, this can be worthwhile, and it can offer a learning opportunity. The prerequisites are a willingness to get to know each other, time to learn each other's trades and a belief that the other is better at something. For example, a health research partner who is known to be more systematic in the manner in which they conduct their research, with more extensive literature reviews and stricter ways of working, may collaborate with a design research partner who uses more open contextual interviews as well as cameras and pencils to map problems. In various projects, we observe that researchers went through a time-consuming sense-making process that they did not foresee but which turned out to be valuable. That process may induce irritation and misunderstandings, but it can also create tangible benefits such as design tools that can be used to gain a better understanding of the manner in which members of the target group experience certain situations. We find that researchers are happy to learn from other disciplines, regardless of the process being time consuming.

On the personal level, transformation occurs when researchers become more open minded in their attitudes and when they start to work more systematically and creatively. Design researchers admit that they have started to work more systematically than before, and health researchers report that they have become more iterative and creative. The prerequisites are enthusiasm about a project, seeing its learning potential, critical (self-)reflection and openness to change. Although it takes time and effort to get to know the process together and to learn from the other discipline, transformation, when it occurs, does generate surprising collaborations and inspiration. This may lead to intensive collaboration between disciplines in which a lot of unscheduled time is spent on sense-making and creating shared understandings.

To summarize, we found several ways to operationalize the collaboration process, which are not always easy. Success requires an appropriate mix of competences, capabilities and skills that individuals bring. It also requires coordination and a coordinator who acts as a boundary broker. This process requires a willingness to get to know each other, time to learn the trades of the other and a belief that the other is better at something. It also requires enthusiasm about a project, a recognition of the learning potential of individuals, critical (self-) reflection and openness to change. Most importantly, these projects show how collaboration can create better results and personal learning.

6.4 CONCLUSIONS AND RECOMMENDATIONS

We had a unique opportunity to follow 10 research projects on eHealth innovation, on which research and practice partners from the creative and the health sector collaborated. We set out to describe the characteristics of their ways of working, to assess the benefits of the projects for stakeholders and to derive recommendations for future cooperation between the two sectors. We encountered 10 unique projects in which ways of working from the design sciences, the health sciences and the behavioral sciences were combined into unique mixtures of research.

WE DRAW THREE MAIN CONCLUSIONS

Creative ways of working are difficult to define and cannot be captured by terms such as 'iterative' and 'designerly.' We found that the ways of working of researchers from different disciplines are more similar than we thought in some respects, which provides opportunities for collaboration. However, one cannot always pinpoint differences in advance. A tool or method can have the same name and the same basic characteristics but a different paradigmatic foundation. For instance, design *tools*, which have become widespread by the notion of 'design thinking,' are not similar to designerly *ways of working*. This carries the risk that the *essence* of some ways of working can be overshadowed. Understanding Create-Health ways of working requires all partners to develop a contextual understanding of *why* and *how* methods and tools are used.

Our research refines the idea of patient participation. Often, the direct participation of end users in the innovation process is promoted as a way to improve the quality of outcomes. However, determining *if* and *when* the target group should participate directly is a balancing exercise. We show that Create-Health research projects typically exert an impact on many different levels, on different actor groups and audiences (which can be both academic and practical), during the process and with different outcomes. In our view, it is not only through direct participation in every phase of every project that end users benefit. The impact that a project has on the academic and the professional parties involved will, eventually, also induce older individuals to adopt healthy living practices and improve their quality of life.

Finally, it is important that research partners take time to recognize and use their unique contributions and ways of working. This means that the collaboration process will always require explicit effort and attention, even in projects in which the partners have collaborated in the past. The design and health disciplines have different customs when it comes to reporting results and sharing findings. This said, they can find each other in several ways when they operationalize their research efforts. Both draw on various fields. Nevertheless, we argue that that both fields need to emphasize their unique contributions and press on to share the results of multidisciplinary cooperation beyond the limits of professional fields.

RECOMMENDATIONS

There are several things that the parties who are involved in Create-Health projects on eHealth innovation can do to improve their chances of success.

In the formation process of the research, make sure to clarify the following elements of the research with all partners:

- 1. The goals of the research and the role of prototypes in reaching those goals;
- 2. The required mix of methods, the role of iterations in mixing them and the intensity of the research collaboration in each phase;
- 3. The roles and tasks of the project partners and the way in which the collaboration is organized across disciplines;
- 4. The way in which the target group will be involved.

The RPM can be used as a tool to facilitate this conversation. The tool is described in Chapter 7.

For the collaboration to be successful, it is important that all participating parties, including end users, benefit from the research. This collaboration begins by posing the following question to each participating party at the start of the project: what is in it for you? Thinking about the project as a service exchange may help in this conversation. This approach then shapes the way in which the project is set up and the methods that are used. Being conscious of added value for end users may lead to the conclusion that their participation in all phases of the research is not always wise.

The foregoing yields a recommendation for funding agencies. There seems to be a tendency among funding agencies to demand participation by end users in every piece of research and in every phase. That may not always be beneficial for the participants or for the research. Careful consideration of the pros and cons of patient participation is needed. Moreover, optimizing the contribution of end users to research requires time for everyone to get to know each other. Budgets should allow for that.

In executing the research, several measures can be taken to ensure cross-fertilization across disciplines. Working across disciplines requires time to learn about others' paradigmatic views on knowledge and ways of working and time to reflect regularly on the cross-disciplinary collaboration and the learning processes that take place. One or more specifically appointed members of the team could assume the role of boundary broker. This also means that funding agencies should recognize that the cross-disciplinary effort should be reflected, in terms of time and money, in proposals.

Finally, we promote learning communities as practices that can take this cross-disciplinary collaboration beyond the scope of the project, as we saw in some of the projects that are presented in this book. Learning communities of this type unite various disciplines and contextualize insights from research and practice. These communities provide a platform for sharing project outcomes, which are often contextualized.

These recommendations could help research partners to get the best out of Create-Health collaborations and to provide actionable knowledge, both for those partners and for academic and professional audiences, so that the solutions that the public need can be created.

HOW TO USE THE RESEARCH PATHWAY TOOL

THIS CONVERSATION TOOL HELPS TO MAP YOUR RESEARCH PATHWAY AND TO DEFINE THE SCOPE OF YOUR RESEARCH.



Figure 7.1 Layers of the Research Pathway Model

INTRODUCTION

The Research Pathway Tool (RPT) is designed to structure conversations between research partners about the research pathway. This research pathway consists of the research steps and the activities that are undertaken to achieve the goals of the research. The tool helps to explore and articulate the many different goals that the partners who are involved have adopted in relation to the project as well as the various ways of getting there that they identify. By doing this, partners are able to pinpoint difficulties in advance.

After you have worked with this tool, you will know more about your research. For instance, you will be able to answer the following questions:

- What do all parties have in mind with this project?
- What (common) goals are you working towards?
- What (research) activities will you conduct?

- In what order will you achieve the goals that you have set?
- Who is involved in which research step?

Consequently, you could also use this tool to plan your project, to discuss changes during the project, to evaluate and/or to map activities to outcomes retrospectively. Doing so could help determine your next steps, such as learning more about what the problem your next research project may need to solve, be it practical problems, further prototype development or delivering knowledge.



Figure 7.2 Research Pathway Tool

BACKGROUND

The Research Pathway Model (RPM) is a process model that explains the type of outcome that follows from different research steps. This helps to create a better shared understanding of a research project. The model consists of two axes (research contexts on the horizontal axis and research activities on the vertical axis) and nine research steps (Figure 7.1).

We discern three research contexts.

- The theoretical context, in which the research is focused on creating, exploring and delivering a better understanding of problems and related propositions for solutions that are advanced and not yet verified in practice.
- The conceptual context, which translates the presupposed solutions into a more specific prototype that is created, explored and made. In this context, researchers, project partners and other stakeholders provide experiential knowledge to translate theory to prototype by themselves, together with the consortium and with or without the end users, but always in a protected space such as a brainstorm room, a pilot environment or in the context of a pilot organization.
- The real-life-practice context, which concerns the context in which the prototype is tested in the world of professional practice and/or the living environment of the end users.

In addition to the three research contexts, the model identifies three overarching research activities.

- Understand and Create refers to research goals, with the aim of improving the understanding of the problem and creating an idea that can be used to solve it. These goals can be realized by engaging in research activities throughout the whole research pathway, whereby it is possible to iterate a research goal during the iterative process. The examples of research activities include performing a literature review ('create theoretical understanding'), co-designing a prototype ('create a concept') and observing end users in their domestic setting ('create understanding of real-life practice').
- Explore and Test refers to research goals, with the aim of exploring an idea, concept, construct or solution by performing research activities. The examples include laboratory research and tests in a controlled environment ('explore theory or a concept in a controlled situation'), evaluating a prototype in a pilot ('explore a concept') and testing a prototype in the environment of the end user ('explore a solution in real-life practice').

• Deliver and Implement refers to goals that are related to the delivery of insights as part of the research pathway. The examples include publishing research papers or grey literature ('deliver theory or knowledge'), delivering a prototype ('proof of concept') to developers or another research project ('deliver a concept') and working on an implementation strategy for an organization ('deliver change in real-life practice').

This results in a 3x3 matrix of nine squares. Each square represents a particular type of research step in a particular context. The matrix does not prescribe an ideal sequence of steps. The resulting pathway might well be an iterative process in which a specific square appears in various steps. The RPM can be used as a process model to map activities, patterns and the linkages between them. By using such a process model, researchers are able to design their research pathways prior to planning their actual research activities. In addition, it helps them to engage in monitoring along the way and to evaluate activities after concluding the research project. The matrix makes it possible to map the activities that contribute to the implementation of the innovation project, whereby contributions to the real-life-practice context, the conceptual context and theoretical context can be made explicit.

	THEORETICAL	CONCEPTUAL	PRACTICAL
	CONTEXT	CONTEXT	CONTEXT
UNDERSTAND & CREATE	CREATE THEORETICAL UNDERSTANDING	CREATE A CONCEPT	CREATE UNDER- Standing of the Real-life practice
EXPLORE & TEST	EXPLORE A THEORY OR Concept in a Controlled Environment	EXPLORE A CONCEPT	EXPLORE A SOLUTION IN REAL-LIFE PRACTICE
DELIVER	deliver a theory or	deliver a concept	deliver a change in
& IMPLEMENT	knowledge		Real life practice

Figure 7.3: Research Pathway Model

HOW TO START WITH THE RESEARCH PATHWAY TOOL?

You start with a research question or a real-life problem. With this tool, you will put together the pathway for this project. Therefore, ask yourself what the ultimate goal and the purpose of the project are. This may be the same goal, but more often it is not. Most innovations take longer to develop. For each phase, it is important to consider what the goals are, who will be involved and what (research) activities are appropriate for a given phase of the innovation process.

The following five steps will help you to create a research pathway for your own project. You can also use the tool to explore a specific part of your research project, such as the involvement of the target group. Another important point to note is that the RPM mandates no specific order, obligation or purpose for all the steps. The same is true of this tool. We can imagine instances in which it is more appropriate to start by defining the target group or to restrict the use of the tool to the conversation about who does what within the consortium or project team.

By asking yourselves questions in each step or each phase, you could spark a conversation between the research partners.





Figure 7.4: Components of the RPT

STEP I : IDENTIFY STAKEHOLDERS, INCLUDING THE TARGET GROUP

The tool is designed to facilitate conversations with stakeholders. Therefore, start the conversation with each other. Who is at the table, and why? What are the different motives of each of the participants in the research project? What are their different fields of expertise and their goals in the project? For who are you going to develop an innovation, product, service or change? Are they the individuals for whom the innovation is intended? Are they already involved in setting goals?

In Chapter 3, we discussed the goals that projects and their stakeholders have in relation to the research pathway. The RPT can help to clarify the goals of each stakeholder and their involvement during the process. For example, are stakeholders primarily involved in the real-life-practice context because they have a role in understanding the context for which an innovation is intended, or do they have a role in the co-design process in the conceptual context, for example because they want to be involved in the formulation of ideas about a solution?

In Chapter 4, we discussed the values of projects and their stakeholders within the research pathway. Will any value accrue to them directly within the process of the research project?

The RPM could also help to identify stakeholders. When you have an overall picture of the research pathway, it becomes clear which stakeholders are essential to each of the steps of the research (Figure 7.2).



Figure 7.5: Stakeholders

Discuss the context(s) in which everyone who is involved would be engaged and the ones in which they feel most comfortable. Again, for each actor, you could discuss everyone's preferences about the nine steps in which they can be involved. Use a figure for each actor in the steps below, and discuss the pathway together.

In which steps is the target group involved?

- How will they be involved?
- How will they contribute to the project?
- Will there be any direct value for them within the process of the research project?

In which steps are the other stakeholders involved? Use the figure.

- How will they be involved?
- How will they contribute to the project?
- Will there be any direct value for them within the project?



Figure 7.6: Stakeholders placed in the RPT

STEP 2: DEFINE YOUR RESEARCH GOALS

Each square raises a question about the research goals of every step of the research. Do you plan or envision that you will perform a certain step at some point in the research project? If the answer to this question is 'yes,' the research pathway will involve this square. If the answer is a definite 'no,' the square is not relevant to this particular project. If all or some partners disagree, then discuss the question further.



Figure 7.7: Checked boxes for the steps you want to explore

When you have printed the tool (Figure 7.4), you can remove the steps you do not use and move on to the ones you want to explore further.

STEP 3: DETERMINE ACTIVITIES FOR EACH RESEARCH GOAL

How can you make sure that all of the activities are performed so as to support the ultimate goal of the research?

You could take the following actions:

- Incorporate previous research results or project outcomes;
- Outsource an activity and collaborate with an organization, an entrepreneur or a professional;
- Define the activities, desired results, the stakeholders that are involved and the person who is responsible for each step of the research pathway.

To work towards the goals, you can combine methods from different disciplines in various ways (See Chapter 3 for examples from the projects). These varied activities can be organized according to their orientation in time: they can be aimed at the future or at the present. The future orientation manifests clearly in the various generative methods that can be used to help individuals express their needs and desires, for example by *co-constructing stories*.

The present orientation is focused on evidence-based and pre-structured methods, for example by introducing a hi-fi prototype to practice. Both orientations provide appropriate means of gaining information on user needs, but the orientations and the expectations of the participants are different.

You can discuss the following questions:

- What kind of methods will you use?
- Are these methods future oriented or now oriented?
- Are the methods typical of a specific research discipline?
- Are the methods particularly relevant to science, practice or both?



Figure 7.8: Research activities and research results, and stakeholders involved placed in the RPT

STEP 4: THINGS

Many projects work with 'things.' Three different types of things can be distinguished, namely products as solutions, research artefacts and proofs of concept (see Chapter 3). They have different purposes and require different ways of working. Therefore, it is important to discuss the kinds of things that you want to develop or use as well as your reasons.



Figure 7.9: Type of 'things' you want to develop

You can discuss the following questions:

What type of 'things' will you develop or use during the project?

• Do you see them as means or as ends within a square? Position the figure in the relevant squares, next to the 'methods' or 'results' cards.

Do you (fore)see any differences in goals or expectations?



Figure 7.10: The 'things' placed in the RPT

STEP 5: PATHWAY OVER TIME

Iterating is an activity that is characteristic of innovative research projects. However, there are different ways of iterating. Chapter 3 provides examples from the 10 projects that show how different goals require different ways of iterating. The RPM helps to highlight these differences because you can map the differences in the manner in which research pathways move between the three contexts.

In the research that we presented in this book, two main ways of moving between the three contexts, which are non-linear processes, emerged: 1) the iterative development of a specific evidence-based innovation into a product (see Example 1, Figure 7.11) and 2) the iterative process of fleshing out a conceptual framework by conducting design explorations (see Example 2, Figure 7.12). The examples are described in Chapter 3. However, other pathways are also possible as long as they are helpful in your project.



Figure 7.11: Example of the iterative development of a product



ITERATING ON A FRAMEWORK

Figure 7.12: Example of the iterative development of a conceptual framework

You could discuss the following questions:

- Which pathway through the 3x3 matrix will you take? Map a path and discuss it.
- Will there be parallel paths?
- Where will the iterations take place?

During the project, research pathways, stakeholders, actors and partners may change, for example because the results of a particular research step call for a revision of the plan. You can use the Research Pathway Tool to consider your plans and to discuss the changes that are needed to arrive at the desired contributions to and benefits from the project as a whole and in relation to each individual actor.

In the following pages you will find all components of the Research Pathway Tool in a printable format. You could print out these specific pages on A4 or A3 paper size. A pair of scissors will do the rest. **LANDER STATE**




CONTEXT

¿DNILLES GETTINC?



EXPLORE & TEST

THEORETICAL CONTEXT **WANILATE OF AN UNA CONTRACTIVE OF AN UNA CONTRACTIVA CONTRACTIVA CONTRACTIVA CONTRACTIVACTIVA CONTRACTIVA CONTRACTIVA CONTRACTIVA C**

IMPLEMENT DELIVER &



SIGNATE







¿DNILLES (GELEVEN) LOTIC VOU USE A PILOT (CREATED) SETTINC?

& TEST & TEST

CONCEPTUAL

CONCEPTUAL

E SALLOLOU A VANITAD NOV OU DO VOU DELIVER A PROTOTYPE?

DELIVER & IMPLEMENT DO YOU CREATE UNDERSTANDING OF REAL-LIFE PRACTICE?

DO YOU CREATE UNDERSTANDING OF REAL-LIFE PRACTICE?





¿DNILLES **BILLES BILLES** DO YOU USE A **REAL-LIFE** SETTINC?

EXPLORE & TEST

PRACTICAL CONTEXT

23012024 NI SADANO Salasa Un Sadano Salasa Ni Sadano Salasa Salas





























LIST OF FIGURES, TABLES AND BOXES

CHAPTER I

- Figure 1.1 When creative and health sectors unite
- Figure 1.2 The Research Pathway Model (RPM)

- Figure 2.1 Actor Map Venn Diagram
- Figure 2.2 Project Actors
- Figure 2.3 RPM
- Figure 2.4 Actor map Everyday Sounds of Dementia
- Figure 2.5 Timeline Everyday Sounds of Dementia
- Figure 2.6 The "Soundboard"
- Figure 2.7 Interaction with the "Tumbler"
- Figure 2.8 Actor map DDD
- Figure 2.9 Timeline DDD
- Figure 2.10 "LivingMoments"
- Figure 2.11 Participant receives a postcard via "LivingMoments"
- Figure 2.12 Actor map NATALIE
- Figure 2.13 Timeline NATALIE
- Figure 2.14 Actor map Growing Roots
- Figure 2.15 Timeline Growing Roots
- Figure 2.16 Virtual Nature Space in a room
- Figure 2.17 Virtual Nature Space in VR
- Figure 2.18 Actor map SQUEALED
- Figure 2.19 Timeline SQUEALED
- Figure 2.20 Ambient Living Environment Data (Technolution bv)
- Figure 2.21 Actor map TTT
- Figure 2.22 Timeline TTT
- Figure 2.23 Unobtrusive Monitoring to Support Home-Based Dementia Care
- Figure 2.24 Actor map PACO
- Figure 2.25 Timeline PACO
- Figure 2.26 "Herman, the cook" in PACO-app
- Figure 2.27 Actor map Healthy Storytelling

- Figure 2.28 Timeline Healthy Storytelling
- Figure 2.29 Bal & Spriet (Ball & Stalk)
- Figure 2.30 Co-design session to develop "Torries"
- Figure 2.31 Gameboard Bal & Spriet (Ball & Stalk)
- Figure 2.32 The "Torries" plushies
- Figure 2.33 Actor map FoodSampler
- Figure 2.34 Timeline FoodSampler
- Figure 2.35 Research participant engaged in contextual interview
- Figure 2.36 Research participant engaged in home lab setting
- Figure 2.37 Actor map GOAL
- Figure 2.38 Timeline GOAL
- Figure 2.39 Research participant engaged with mHealth app
- Figure 2.40 Research participant engaged in healthy task
- Figure 2.41 Research participant engaged in an experiment

- Figure 3.1 Four themes of ways of working
- Figure 3.2 One of the experiments with health games ran at schools
- Figure 3.3 The various activities for the top two rows in the RPM matrix
- Figure 3.4 The sound box that was personalized in the co-design sessions in the Tumbler study.
- Figure 3.5 'Perspective' as an extra layer to the RPM
- Figure 3.6 The stuffed toys named 'Torries'
- Figure 3.7 One of the virtual agents in the developed app
- Figure 3.8 The persona creation in the project was fed by both the longitudinal interviews and the context mapping activities
- Figure 3.9 Research participant in interaction with the VITA-cushion
- Figure 3.10 Research participant in interaction with an experimental prototype in a lab setting
- Figure 3.11 Research participants with the virtual nature environment
- Figure 3.12 The iterative pathway of the project PACO
- Figure 3.13 The iterative pathway of the project healthy Storytelling
- Figure 3.14 Two ways of iterating and moving between contexts
- Figure 3.15 The timeline of the project Sounds of Dementia
- Figure 3.16 Tips in four themes for ways of working in eHealth projects
- Table 3.1 The insights on the ways of working to deal with goals in eHealth projects

- Table 3.2 The insights on the ways to mix methods in eHealth projects
- Table 3.3The insights on the ways to deal with the role of prototypes in
eHealth projects
- Table 3.4
 The insights on the ways to iterate in eHealth projects
- Box 3.1 Direct impact in a theoretical project
- Box 3.2 A design game to make theory actionable
- Box 3.3 Generative ways of doing user research
- Box 3.4 Qualitative evaluating on three levels
- Box 3.5 Different perspectives in evaluating a virtual agent
- Box 3.6 A ping-pong approach for longitudinal research
- Box 3.7 Vita cushion: direct interaction between researcher and participant using a prototype
- Box 3.8 Experiential prototypes
- Box 3.9 Adapting and finetuning the VR environment
- Box 3.10 Studying sounds & dementia in various contexts

- Figure 4.1 Exchange of knowledge, skills, and tools between disciplines
- Figure 4.2 Co-creation session project PACO
- Figure 4.3 Types of stakeholders in the ten projects
- Figure 4.4 VR environment on a laptop screen
- Figure 4.5 VR environment in a room
- Figure 4.6 Involved stakeholders in the conceptual context
- Figure 4.7 Involvement as a layer in the Research Pathway Model
- Table 4.1Stakeholder involvement in research projects on eHealth
- Table 4.2Research Pathway Model (Van Beest et al., 2021)
- Table 4.3The theoretical context of the Research Pathway Model
- Table 4.4The conceptual context of the Research Pathway Model
- Table 4.5The real-life practice context of the Research Pathway Model
- Box 4.1: Attempts to approach people of the target group
- Box 4.2: Test with representatives of the target group
- Box 4.3: First step into the world of real-life practice

CHAPTER 5

- Figure 5.1 Intrapersonal aspects
- Figure 5.2 The various activities for the top two rows in the RPM
- Table 5.1
 Boundary crossing learning mechanisms
- Table 5.2
 Structure of this chapter, following the four learning mechanisms
- Table 5.3The insights on identification of practices in Create Health r
esearch collaborations
- Table 5.4
 Differences in interests of behavioral and design researchers
- Table 5.5The insights on coordination of tasks in Create Health research consortia
- Table 5.6
 The insights regarding reflection on collaboration
- Table 5.7
 The insights on transformation by interdisciplinary learning

CHAPTER 6

Figure 6.1 When creative and health sectors unite

- Figure 7.1 Layers of the Research Pathway Model
- Figure 7.2 Research Pathway Tool
- Figure 7.3 Research Pathway Model
- Figure 7.4 Components of the RPT
- Figure 7.5 Stakeholders
- Figure 7.6 Stakeholders placed in the RPT
- Figure 7.7 Checked boxes for the steps you want to explore
- Figure 7.8 Research activities, research results, and stakeholders involved, placed in the RPT
- Figure 7.9 Type of 'things' you want to develop
- Figure 7.10 The 'things' placed in the RPT
- Figure 7.11 Example of the iterative development of a product
- Figure 7.12 Example of the iterative development of a conceptual framework

CHAPTER I

- Akkerman, S.F., & Bakker, A. (2011). Boundary crossing and boundary objects. Review of Educational Research, 81(2), 132–169.
- Jonas, W. (2006). Research through DESIGN through research a problem statement and a conceptual sketch. In *Design Research Society* (Vol. 36, pp. 1362–1380). Lisbon. doi. org/10.1108/03684920710827355
- Simonse, L., Albayrak, A., & Starre, S. (2019) Patient journey method for integrated service design, *Design for Health*, 3:1, 82-97, DOI: 10.1080/24735132.2019.1582741
- Nelson, H.G. & Stolterman, E. (2003) "Design judgement: decision-making in the 'real'world." The Design Journal 6.1: 23-31.
- Yin, R.K. (2009) Case study research: Design and methods. Vol. 5. SAGE, Thousand Oaks, USA.
- Van Beest. W, Boon P.C.W., Andriessen D., Pol H., Van der Veen G., & Moors E.H.M. (2021). A Research Pathway Model for evaluating the implementation of practice-based research: The case of self-management health innovations, *Research Evaluation*, rvab023, https://doi.org/10.1093/reseval/rvab023
- Vargo, S. & Lusch, R. (2004) Evolving to a new dominant logic in marketing. *Journal of Marketing*, 68 (1): 1-17.
- ZonMw. (2018). Subsidieoproep: Overkoepelend project Create Health. Evaluatie van het gebruik van creatieve werkwijzen die tot innovaties in de zorg leiden. Den Haag: ZonMw.

CHAPTER 2

PROJECT EVERYDAY SOUNDS OF DEMENTIA

FURTHER READING

- Houben, M., Brankaert, R., Bakker, S., Kenning, G., Bongers, I. & Eggen, B. (2019). Foregrounding Everyday Sounds in Dementia. In *Proceedings of the 2019 Conference on Designing Interactive Systems (DIS '19)*
- Houben, M., Brankaert, R., Kenning, G., Eggen, B. & Bongers, I. (2020). The Perspectives of Professional Caregivers on Implementing Audio-Based Technology in Residential Dementia Care. In International Journal of Environmental Research and Public Health 2020, 17, 6333. (IJERPH)
- Houben, M., Brankaert, R., Bakker, S., Kenning, G., Bongers, I. & Eggen, B. (2020). The Role of Everyday Sounds in Advanced Dementia Care. In *Proceedings of the 2020 Conference on Human Factors in Computing Systems (CHI '20)*

- Houben, M., Brankaert, R., Kenning, G., Bongers, I. & Eggen, B. (2022a). Designing for Everyday Sounds at Home with People with Dementia and their Partners. In *Proceedings* of the 2022 Conference on Human Factors in Computing Systems (CHI' 22)
- Houben, M., Brankaert, R., Dhaeze, E., Kenning, G., Bongers, I. & Eggen, B. (2022b). Enriching Everyday Lived Experiences in Dementia Care. In *Proceedings of the 2022 Conference on Tangible, Embedded, and Embodied Interaction (TEI '22)*

- Brankaert, R. (2016). Design for Dementia: A design-driven living lab approach to involve people with dementia and their context. PhD, Eindhoven University of Technology.
- Hummels, C. & Frens, J. (2011). Designing Disruptive Innovative Systems, Products and Services: RTD Process. 10.5772/22580.
- Kenning, G. (2017). Making It Together: Reciprocal Design to Promote Positive Wellbeing for People Living with Dementia. University of Technology Sydney
- Schafer, R. & Murray. (1977). The Soundscape: Our Sonic Environment and the Tuning of the world. Rochester, VT: Destiny Books. Schiffer, Michael Brian. 1991.
- Treadaway, C., Kenning, G. & Coleman, S. (2015). "Sensor e-Textiles: Designing for persons with late stage dementia." *ThirdEuropean Conference on Design4Health*, Sheffield UK.

PROJECT DEMENTIA DYNAMICS IN DESIGN (DDD)

FURTHER READING

- Diks, S. H., Muyrers, T. H. C., Huang, T-J., Chen, G., Thoolen, M. & Brankaert, R. G. A., 8 May (2021). CoasterChat: Exploring Digital Communication between People with Early Stage Dementia and Family Members Embedded in a Daily Routine. Extended Abstracts of the 2021 CHI Conference on Human Factors in Computing Systems, CHI EA 2021: Extended Abstracts of the 2021 CHI Conference on Human Factors in Computing Systems. Association for Computing Machinery, Inc, p. 1-7 7 p. 402
- Godfroij, B., Van Beest, W. & Van Gessel, C. (2020). *Co-design in de anderhalvemetermaatschappij*, Whitepaper. Utrecht University of Applied Sciences, Utrecht, The Netherlands.
- Godfroij, B., Peek, S., van der Lugt, R., & Brankaert, R. (2022). Changing Roles in Interdisciplinary Research-through-Design. In G. Goldschmidt & E. Tarazi, (Eds.), Expanding the frontiers of design: A blessing or a curse? *Proceedings of the 13th Design Thinking Research Symposium* (p.410 – 429). DesignTech, Israel Institute of Technology, Haifa.
- Kolasinska, A.B., Thoolen, M., Peek, S., Lu, Y., & Brankaert, R. (2021). Co-creating Design Opportunities for Social Technology in the Context of Dementia. *Dementia Lab 2021: Supporting Ability Through Design*, Springer International Publishing, 125-141.
- Thoolen, M., Brankaert, R. & Lu, Y., (2019). Sentic: a tailored interface design for people with dementia to access music. *DIS 2019 Companion - Companion Publication of the 2019 ACM Designing Interactive Systems Conference*. New York: Association for Computing Machinery, Inc, p. 57-60 4 p.

- Thoolen, M., Brankaert, R. G. A. & Lu, Y., (2020a). *AmbientEcho: Exploring Interactive Media Experiences in the Context of Residential Dementia Care.* DIS '20: Proceedings of the 2020 ACM Designing Interactive Systems Conference: Association for Computing Machinery, Inc, p. 1495-1508 14 p.
- Thoolen, M., Brankaert, R. G. A. & Lu, Y., (2020b). Designing Sentic: Participatory Design with People Living with Dementia. *HCI and Design in the Context of Dementia*. Springer Nature Switzerland AG: Springer, Vol. 1. p. 269-288 20 p.

- Peek, S. (2017). Understanding technology acceptance by older adults who are aging in place: A dynamic perspective. Ipskamp.
- Peek, S.T.M. Luijkx, K.G., Rijnaard, M.D., Nieboer, M.E., Van der Voort, C.S., Aarts, S., Van Hoof, J., Vrijhoef, H.J.M, Wouters, E.J.M., (2016). Older Adults' Reasons for Using Technology while Aging in Place. Gerontology, 62(2), pp.226–237.
- Peek, S.T.M., Luijkx, K.G., Vrijhoef, H.J.M, Nieboer, M.N., Aarts, S., van der Voort, C.S., Rijnaard, M.D., & Wouters, E.J.M. (2017a). Changes and Stability in the Use of Technologies by Independent-Living Seniors Over Time. Manuscript submitted for publication.
- Peek, S.T.M., Luijkx, K.G., Vrijhoef, H.J.M, Nieboer, M.N., Aarts, S., van der Voort, C.S., Rijnaard, M.D., & Wouters, E.J.M. (2017b). Origins and Consequences of Technology Acquirement by Independent-Living Seniors: Towards an Integrative Model. BMC Geriatrics 17, 189 (2017).
- Peek, S.T.M., Wouters, E.J.M., van Hoof, J., Luijkx, K.G., Boeije, H.R., & Vrijhoef, H.J.M. (2014). Factors influencing acceptance of technology for aging in place: A systematic review. International Journal of Medical Informatics, 83(4), 235–248.

PROJECT NATALIE: PERSUASIVE TECHNOLOGY TO SUPPORT COMMUNICATION AND NEEDS ARTICULATION OF PEOPLE WITH DEMENTIA

FURTHER READING

Wartena B.O., Tabak R. & Van 't Veer J.T. (2019). NATALIE (Need Articulation Through Autonomy Loss In Elderly). *Supporting Health by Technology 2019*: Martiniplaza Groningen.

Wartena BO, van 't Veer J.T., Visch, V.V. & Hekkert, P.P.M. (Pre-print) How to balance autonomy resonance design for older adults; a design research viewpoint, *JMIR Aging*

REFERENCES

- Dijkstra, A., Buist G., Dassen Th. (1996). Nursing-Care Dependency: Development of an assessment scale for demented and mentally handicapped patients. *Scandinavian Journal of Caring Science* 10: 137-143.
- Dijkstra, A. (1998). Care Dependency; an assessment instrument for use in long-term care facilities. Academisch proefschrift. Groningen: De Regenboog.

- Dupuis, S.L., Gillies, J., Carson, J. & Whyte, C. (2011). Moving beyond patient and client approaches: Mobilizing 'authentic partnerships' in dementia care, support and services. *Dementia*: 11, p. 427-452.
- Henderson, V. (1966) The Nature of Nursing. New York, Macmillan
- Lindqvist, E., Nygard, L., & Borell, L. (2013). Significant junctures on the way towards becoming a user of assistive technology in Alzheimer's disease. *Scand. J. Occup. Ther.* 20 (5), 386–396.
- Malinowsky, C., Nygård, L., & Kottorp, A. (2013). Using a screening tool to evaluate potential use of e-health services for older people with and without cognitive impairment. *Aging Ment. Health* 18, 340–345.
- Span, M., Smits, C., Jukema J., Groen-van de Ven, L., Janssen R., Vernooij-Dassen, M., Eefsting, J. & Hettinga, M. (2015). An interactive web tool for facilitating shared decisionmaking in dementia-care networks: a field study. *Frontiers in Aging Neuroscience* 7:128.
- Span, M. (2016). Developing an interactive web tool to facilitate share decisionmaking in dementia care networks: a participatory journey. Academisch proefschrift, Vrije Universiteit Amsterdam.
- Van den Berg, M. (2013). *De usability and acceptability van Lable Care in kaart gebracht*. (Bachelor's thesis, University of Twente, The Netherlands). Retrieved from: http://www. lablecare.nl/media/LableCare-Usability-and-Acceptability.pdf

PROJECT GROWING ROOTS: CONNECTING ELDERLY THROUGH VIRTUAL NATURE SPACES

SCIENTIFIC OUTPUT

- Van Houwelingen-Snippe, J., Ben Allouch, S., & van Rompay, T. J. L. (2021). Virtual Reality Representations of Nature to Improve Well-Being amongst Older Adults: A Rapid Review. Journal of Technology in Behavioral Science. https://doi.org/10.1007/s41347-021-00195-6
- Van Houwelingen-Snippe, J., van Rompay, T. J. L., & Allouch, S. B. (2020). Feeling connected after experiencing digital nature: A survey study. *International journal of environmental research and public health*, 17(18), [6879]. https://doi.org/10.3390/ ijerph17186879
- Van Houwelingen-Snippe, J., van Rompay, T. J. L., de Jong, M. D. T., & Allouch, S. B. (2020). Does digital nature enhance social aspirations? An experimental study. *International journal of environmental research and public health*, 17(4), [1454]. https://doi. org/10.3390/ijerph17041454

REFERENCES

Aron, A., Aron, E. N., & Smollan, D. (1992). Inclusion of other in the self-scale and the structure of interpersonal closeness. *Journal of Personality and Social Psychology*, 63, 596–612.

- Baños, R., Etchemendy, E., Castilla, D., García-Palacios, A., Quero, S., & Botella, C. (2012). Positive mood induction procedures for virtual environments designed for elderly people. Interacting with Computers, 24,131–138.
- Farris, M., Bates, R., Resnick, H., & Stabler, N. (1994). Evaluation of computer games' impact upon cognitively impaired frail elderly. *Computers in Human Services*, 11, 219–228.
- Fredrickson, B. L. (2001). The role of positive emotions in positive psychology. The 723 broaden-and-build theory of positive emotions. American Psychologist, 56, 218–226
- Graaf, de M., Ben Allouch, S., & Van Dijk, J. (2017). Long-term evaluation of a social robot in real 731 homes. Interaction Studies, 17(3), 462-491.
- Goldstein, J., Cajko, L., Oosterbroek, M., et al. (1997). Video games and the elderly. *Social Behavior, and Personality*, 25, 345-352.
- Jaschinski, C., & Ben Allouch, S. (2015a). An extended view on benefits and barriers of ambient assisted living solutions. *International Journal on Advances in Life Sciences*, 7, 40-53.
- Jaschinski, C., & Ben Allouch, S. (2015b). Understanding the user's acceptance of a sensor-based ambient assisted living application. In A. A. Salah, B. J. A. Kröse, & D. J. Cook (Eds.), *Proceedings of the 6th International Workshop on Human Behavior Understanding*, LNCS, vol. 9277, p. 13-25. Switzerland: Springer International Publishing.
- Jung, Y., Yeh, S., McLaughlin, M., Rizzo, A., & Winstein, C. (2009). Three-dimensional game 752 environments for recovery from stroke. In U. Ritterfeld, M. Cody, & P. Vorderer (Eds.), Serious games: 753 Mechanisms and effects. New York: Routledge, Taylor and Francis, p. 413-428.
- Kaplan, R., & Kaplan, S. (1989). The experience of nature: A psychological perspective. New York: Cambridge University Press.
- Kaplan, R., & Kaplan, S. (2011). Well-being, reasonableness, and the natural environment. Applied Psychology: Health and Well-being, 3, 304-321.
- Lange B., Requejo P., & Flynn S., (2010). The potential of virtual reality and gaming to assist successful aging with disability. Physical Medicine and Rehabilitation Clinics of North America, 21, 339–356.
- Piff, P.K., Dietze, P., Feinberg, M., Stancato, D.M., & Keltner, D. (2015). Awe, the small self, and prosocial behavior. *Journal of Personality and Social Psychology*,108, 883–899.
- Tech4People (2016). Digital nature helps people to recover. UT Nieuws, 9, 31-32 (https://www.utoday.nl/science/63493/Digital_nature_helping_patients_to_recover).
- Ulrich, R. S. (1984). View through a window may influence recovery from surgery. *Science*, 224, 42-421.
- Van Rompay, T.J.L., & Jol, T. (2016). Wild and free: Unpredictability and spaciousness as predictors of creative performance. *Journal of Environmental Psychology*, 48, 140-148.
- Zelenski, J.M., & Nisbet, E.K. (2014). Happiness and feeling connected: The distinct role of nature 801 relatedness. Environment and Behavior, 46, 3-23.

PROJECT SQUEALED: SUPPORT QUALITY-CARE FOR ELDERLY USING AMBIENT LIVING ENVIRONMENT DATA

FURTHER READING & REFERENCES

Kelsey, J.L., Berry, S.D., Procter-Gray, E., Quach, L., Nguyen, U-S.D.T., Li, W., Kiel, D.P., Lipsitz, L.A., & Hannan, M.T. (2010). Indoor and outdoor falls in older adults are different: The maintenance of balance, independent living, intellect, and Zest in the Elderly of Boston Study. J Am Geriatr Soc, 2010. 58 (11): p. 2135-41

PROJECT TRACK, TRACE & TRIGGER: UNOBTRUSIVE SENSING TECHNOLOGIES TO MONITOR AND COACH ELDERLY WITH DEMENTIA

FURTHER READING

- Bastoni, S., Wrede, C., da Silva, M.C., Sanderman, R., Gaggioli, A., Braakman-Jansen, A. & van Gemert-Pijnen L (2021). Factors Influencing Implementation of eHealth Technologies to Support Informal Dementia Care: Umbrella Review. *JMIR Aging*;4(4). doi: 10.2196/30841
- Klein Brinke, J., Chiumento, A., & Havinga, P. (2021). Personal Hygiene Monitoring Under the Shower Using WiFi Channel State Information. In 1st Workshop on Computer Human Interaction in IoT Applications, CHILOT 2021 – Virtual event.
- Sharma, N., Klein Brinke, J.K., van Gemert-Pijnen, L.J.E.W.C., & Braakman-Jansen, L.M.A. (2022). Implementation of Unobtrusive Sensing Systems for Older Adult Care: Scoping Review. JMIR Aging 2021;4(4). doi: 10.2196/27862
- Wrede, C., Braakman-Jansen, A., & van Gemert-Pijnen, L. (2021). Requirements for Unobtrusive Monitoring to Support Home-Based Dementia Care: Qualitative Study Among Formal and Informal Caregivers. *JMIR Aging*; 4(2). doi: 10.2196/26875

PUBLICATIONS (TO BE) SUBMITTED

- Sharma, S., Klein Brinke, J., Braakman-Jansen, L.M.A., Croockewit, J.H., van Gemert-Pijnen, J.E.W.C., Havinga, P., & Duc, L.V. (2022) Wi-Gitation: replica dataset for physical agitation activities recognition with Wi-Fi CSI signal. Proceedings of the ACM on Interactive, Mobile, Wearable and Ubiquitous Technologies (IMWUT), submitted February 2022
- Wrede, C., Braakman-Jansen, L.M.A., & van Gemert-Pijnen, L.J.E.W.C. (2022) How to create value with monitoring technology in home-based dementia care: a multimethod qualitative study among stakeholders.

CONFERENCE PAPERS

Bastoni, S., Wrede, C., Da Silva, M. C., Sanderman, R., Gaggioli, A., & van Gemert-Pijnen, L. J. E. W. C. (2021). Determinants of Implementation of eHealth Solutions to Support Informal Dementia Care: An Umbrella Review. Poster session presented at 10th Annual Conference of the Association for Researchers in Psychology and Health (ARPH) 2021.

- Wrede, C., Jansen, L., Havinga, P.J.M., Croockewit, J.H., & van Gemert-Pijnen, L.J.E.W.C. (2019). Unobtrusive sensing technologies to monitor and coach elderly with dementia: Track, Trace & Trigger. In E. Kyza, P. Karpinnen, E. Karapanos, K. T. Win, & H. Oinas-Kukkonen (Eds.), Adjunct Proceedings of the 14th International Conference on Persuasive Technology, Limassol, Cyprus (pp. 43-45) (poster session)
- Wrede, C., Braakman-Jansen, A., & van Gemert-Pijnen, L. (2019a). Niet-hinderlijke monitoring- en coaching technologie in de dementie thuiszorg. Oral presentation at *Dementie Experience* Liberein 2019, Enschede, NL.
- Wrede, C., Braakman-Jansen, A., & van Gemert-Pijnen, L. (2019b). Privacy perspectives on unobtrusive in-home monitoring of elderly with dementia: Challenges in design and user acceptance. Oral presentation at *Mind Your Data Conference 2019*: Privacy and Legal Matters in E-health. Leiden, The Netherlands.
- Wrede, C., Braakman-Jansen, A., & van Gemert-Pijnen, L. (2019c). Involving dementia practice in research. Oral presentation at *4th Dementia Lab Conference* 2019. Eindhoven, The Netherlands.
- Wrede, C., Braakman-Jansen, A., & van Gemert-Pijnen, L. (2020a). Unobtrusive sensing technologies to support independent living of people with dementia. Oral presentation at *Personalized eHealth Technology Symposium 2020*, Enschede, NL.
- Wrede, C., Braakman-Jansen, A., & van Gemert-Pijnen, L. (2020b). Exploring views and needs of (in)formal caregivers towards unobtrusive monitoring in home-based dementia care. Poster session presented at *30th Alzheimer Europe Conference* (AEC), 2020.
- Wrede, C., Braakman-Jansen, A., & van Gemert-Pijnen, L. (2021a). An integrated digital care collaboration platform to support home-based dementia care: A mixed-methods evaluation. Poster session presented at *31th Alzheimer Europe Conference (AEC)*, 2021.
- Wrede, C., Braakman-Jansen, A., & van Gemert-Pijnen, L. (2021b). Voices of Caregivers: Key Demands Towards AI-driven Home Monitoring in Community-based Dementia Care. Poster session presented at the 2021 GSA Annual Scientific Meeting. Innovation in Aging, 5(Suppl. 1), 664. https://doi.org/10.1093/geroni/igab046.2489.

WORK IN PROGRESS

- Klein Brinke J. et al: Collaborative sensing and communication in a Wi-Fi enabled system (in progress 2022)
- Klein Brinke J. et al: Using Federated Learning to enable multi-transmitter and multi-receiver systems using Wi-Fi in a home setting (in progress 2022)
- Sharma N. et al: Leveraging transfer learning for physical agitation activity recognition using Wi-Fi CSI signal (in progress 2022).
- Vuijk, J., Sharma N, Klein Brinke J: Comparing different RF-based sensing technologies for monitoring the effect of music on a healthy person (MSc thesis 2022, potentieel paper)

REFERENCES

Kamminga, J., Le, D.V., & Havinga, P. (2016). Ambient sound-based collaborative localization of indeterministic devices. *Sensors*, 16(9), 1478.

- Le, D.V., Kamminga, J. W., Scholten, H., & Havinga, P. J. (2016, May). Error bounds for localization with noise diversity. In 2016 International Conference on Distributed Computing in Sensor Systems (DCOSS) (pp. 83-92). IEEE.
- Le, D.V., Kamminga, J. W., Scholten, H., & Havinga, P. J. (2016, March). Nondeterministic sound source localization with smartphones in crowdsensing. In 2016 IEEE International Conference on Pervasive Computing and Communication Workshops (PerCom Workshops) (pp. 1-7). IEEE.
- Le, D.V., Scholten, H., & Havinga, P. (2013, September). Flead: Online frequency likelihood estimation anomaly detection for mobile sensing. In *Proceedings of the 2013 ACM conference on Pervasive and ubiquitous computing adjunct publication* (pp. 1159-1166).
- Le, D.V., Scholten, H., Havinga, P. M., & Ngo, H. (2014, January). Location-based data dissemination with human mobility using online density estimation. In 2014 IEEE 11th Consumer Communications and Networking Conference (CCNC) (pp. 450-457). IEEE.
- Nguyen T., Nguyen V., Salim F.D., Le D.V. & Phung D. (2017). A Simultaneous Extraction of Context and Community from pervasive signals using nested Dirichlet process. *Pervasive and Mobile Computing*. 38,2: 396-417.
- Salomons, E. L., Havinga, P. J., & Van Leeuwen, H. (2016). Inferring human activity recognition with ambient sound on wireless sensor nodes. *Sensors*, 16(10), 1586.
- Van Kleunen, W. A., Le D.V., & Havinga, P. J. (2016, October). Calibration-free signal-strength localization using product-moment correlation. In 2016 IEEE 13th International Conference on Mobile Ad Hoc and Sensor Systems (MASS) (pp. 210-218). IEEE.

PROJECT PACO: PERSUASIVE DIGITAL AGENTS TO COACH OLDER ADULTS TOWARDS DIETARY BEHAVIOR CHANGE

FURTHER READING

- Kramer, L.L. (2021). Meet the peer and the cook: Design and evaluation of persuasive Embodied Conversational Agents to support healthy ageing. Wageningen University. https://doi.org/10.18174/545783
- Kramer, L.L., Blok, M., Van Velsen, L., Mulder, B.C., & De Vet, E. (2021). Supporting eating behavior of community-dwelling older adults: co-design of an embodied conversational agent. *Design for Health*, 1-20
- Kramer, L.L., Mulder, B.C., Van Velsen, L., & De Vet, E. (2021). Use and effect of web-based embodied conversational agents for improving eating behavior and decreasing loneliness among community-dwelling older adults: Protocol for a Randomized Controlled Trial. JMIR Research Protocols 10 (1), e22186
- Kramer, L.L., Ter Stal, S., Mulder, B.C., De Vet, E. & Van Velsen, L. (2020). Developing embodied conversational agents for coaching people in a healthy lifestyle: Scoping review. *Journal of medical Internet research* 22 (2), e14058
- Ter Stal, S., Kramer, L.L., Tabak, M., Op den Akker, H., & Hermens, H. (2020) Design features of embodied conversational agents in eHealth: a literature review. *International Journal of Human-Computer Studies* 138, 102409

- Battagolino, C., Bickmore, T. Increasing the Engagement of Conversational Agents through CoConstructed Storytelling. In: Intelligent Narrative Technologies and Social Believability in Games: Papers from the AIIDE 2015 Joint Workshop. 2015.
- Petty, R.E. & Cacioppo, J.T. (1986). The Elaboration Likelihood Model of Persuasion, in Communication and Persuasion: Central and Peripheral Routes to Attitude Change. Springer: New York.
- Vardoulakis, L.P., Ring, L., Barry, B., Sidner, C.L., Bickmore, T. Designing relational agents as long term social companions for older adults. In: *Intelligent Virtual Agents* (Lecture Notes in Computer Science), 2012, 289–302.

PROJECT HEALTHY STORYTELLING FOR EHEALTH

FURTHER READING

- Van Boeijen, A.G.C., Visch, V.T., & Vegt, N.J.H. (expected). Storytelling as a design element for healthcare.
- Van Eijk, S. (2019) Big Fat Nonsense A discussion tool to reduce weight-related stigmas in the direct social environment. Master thesis, Faculty of Industrial Design Engineering, Delft.
- Vegt, N.J.H., van Boeijen, A.G.C., Spooren, W.P.M.S., Evers, A.W.M., van Rossum, E.F.C, & Visch, V.T. (expected). Can storytelling erase stigmas? A framework for designing interactive storytelling spaces that reduce health-related stigma.
- Vegt, N.J.H., Visch, V.T., van Boeijen, A.G.C., & Spooren, W.P.M.S. (2021). 'Er is nog één smal stoeltje vrij. Wat doe je? – Spel maakt stigmatisering van overgewicht bespreekbaar'. Onze Taal, 90 (1), 18-20.
- Vegt, N.J.H, Visch, V.T., van Boeijen A.G.C. et al. (expected). Hosting the Torries. A case study on increasing health literacy through interactive narrative in low SES families.
- Visch, V.V., Vegt, N.J.H, van Boeijen, A.G.C., & Verstappen, C. (expected). Playing Ball & Stalk to discuss obesity stigmatization

REFERENCES

- Haigh, C. & Hardy, P. (2011). Tell me a story a conceptual exploration of storytelling healthcare education. *Nurse Education Today*, 31, 408-411.
- Horvath, B. (1987). Text in conversation: Variability in storytelling texts. Variation in Language: NWAVXV at Stanford. 212-23.
- Macaulay, R. (2002). Extremely interesting, very interesting, or only quite interesting? Adverbs and social class. *Journal of sociolinguistics*, 6, 398-417.
- Gray, J. (2009). The power of storytelling: using narrative in the healthcare context. *Journal* of Communication in Healthcare, 2(3), 258-273.
- Slater, M., Buller, D., Waters, E., Archibeque, M. & LeBlanc, M. (2003). A test of conversational and testimonial messages versus didactic presentations of nutrition information. *Journal of Nutrition Education Behavior*, 35(5), 255-259.

PROJECT FOODSAMPLER: DIETARY REPORTING IN DAILY LIFE

FURTHER READING

- Romero Herrera, N., Davis-Owusu, K., van Oers, S., de van der Schueren, M., Alberts, J., & Vastenburg, M. (2018, May). FoodSampler: engaging people to contextualise food behavior: Mixed methods for monitoring choices and triggers of eating habits. In Proceedings of the 12th EAI International Conference on Pervasive Computing Technologies for Healthcare (pp. 269-273). ACM
- Van Oers, S., Davis-Owusu, K., Romero Herrera, N., Vastenburg, M., Verwijs, M., van Brakel, L., Beeren, I., Holtkamp, M., Helder, S., Vreugdenhil, A., & van der Schueren, M. (2019). Foodsampler: de context als onderdeel van voedingsinnameregistratie. *Nederlands Tijdschrift voor Voeding en Dietiek*, 74(3), 21-28.

REFERENCES

- Creswell, J., & Piano, V. (2011). Designing and Conducting Mixed Methods Research. Sage Publications.
- Hektner, J. M., Schmidt, J. A., & Czikszentmihalyi, M. (2007). Experience sampling method: Measuring the quality of everyday life. Sage Publications.
- Huber, M., van Vliet, M., Giezenberg, M., Winkens, B., Heerkens, Y., Dagnelie, P. C.,
 & Knottnerus, J. A. (2016). Towards a 'patient-centred' operationalisation of the new dynamic concept of health: a mixed methods study. *BMJ open*, 6(1).
- Kahneman, D., Krueger, A. B., Schkade, D. A., Schwarz, N., & Stone, A. A. (2004). A survey method for characterizing daily life experience: The day reconstruction method. *Science*, 306(5702), 1776–1780.
- Krogstie, J., (2012) Bridging Research and Innovation by Applying Living Labs for Design Science Research, Nordic Contributions in IS Research, Springer Berlin Heidelberg, pp.161–176
- O'Brien, H. L., & Toms, E. G. (2008). What is user engagement? A conceptual framework for defining user engagement with technology. *Journal of the American Society for Information Science and Technology*, 59(6), 938-955.
- Romero Herrera, N. (2017a) In-Situ and Mixed-Design Interventions. D. Keyson, D. Lockton and O. GuerraSantin (eds) *Living Labs: Design and Assessment of Sustainable Living*, pp.157-167, Springer, The Netherlands.
- Romero Herrera, N. (2017b). The emergence of living lab methods. In *Living Labs* (pp. 9-22). Springer, Cham.

PROJECT GOAL: GAMIFICATION FOR OVERWEIGHT PREVENTION AND ACTIVE LIFESTYLE

SCIENTIFIC OUTPUT

d'Hondt J.E., Nuijten R.C.Y., Van Gorp P.M.E. (2019) Evaluation of Computer-Tailored Motivational Messaging in a Health Promotion Context. In: Bella G., Bouquet P. (eds) Modeling and Using Context. CONTEXT 2019. Lecture Notes in Computer Science, vol 11939. Springer, Cham. https://doi.org/10.1007/978-3-030-34974-5_11

- Haghighi, E.H., Nuijten, R., Van Gorp, P. (2021). Towards Hybrid Profiling: combining digital phenotyping with validated survey questions to balance data entry effort with predictive power. Proceedings 2021 International Symposium on Electrical, Electronics and Information Engineering, ISEEIE 2021. Seoul, Zuid-Korea, 19-21 February 2021.
- Khanshan, A., Van Gorp, P., Nuijten, R., Markopoulos, P. (2021). Assessing the Influence of Physical Activity Upon the Experience Sampling Response Rate on Wrist-Worn Devices. Int. J. Environ. Res. Public Health 2021, 18, 10593. https://doi.org/10.3390/ ijerph182010593
- Nuijten, R.C.Y. (2022). Energize: Exploring the impact of gamification strategies on engagement with mobile health apps. Eindhoven University of Technology.
- Nuijten, R., Kaymak, U., & van Gorp, P. M. E. (2019). Evaluation of the impact of the social proof principle and extrinsic rewards in a health promotion campaign. Postersessie gepresenteerd op *ICT OPEN 2019*, Hilversum, Nederland.
- Nuijten, R., Kaymak, U., Van Gorp, P., Simons, M., van den Berg, P., & Le Blanc, P. (2019). Fuzzy modeling to 'understand' personal preferences of mHealth users: a case study. In V. Novak, V. Marik, M. Stepnicka, M. Navara, & P. Hurtik (editors), 2019 Conference of the International Fuzzy Systems Association and the European Society for Fuzzy Logic and Technology (EUSFLAT 2019) pp. 558-565. Atlantis Studies in Uncertainty Modelling; Vol. 1. Atlantis Press. https://doi.org/10.2991/eusflat-19.2019.77
- Nuijten, R. & Van Gorp, P. (2021). SciModeler: A Metamodel and Graph Database for Consolidating Scientific Knowledge by Linking Empirical Data with Theoretical Constructs. In Proceedings of the 9th International Conference on Model-Driven Engineering and Software Development - MODELSWARD, pp. 314-321. DOI: 10.5220/0010315503140321
- Nuijten R.C.Y., Van Gorp, P., Borghouts, T., Le Blanc, P., Van den Berg, P., Kemperman, A., Hadian, E., & Simons, M. (2021). Preadolescent Students' Engagement With an mHealth Intervention Fostering Social Comparison for Health Behavior Change: Crossover Experimental Study. J Med Internet Res 2021;23(7)
- Nuijten, R., Van Gorp, P., Hietbrink, J., Le Blanc, P., Kemperman, A., van den Berg, P. & Simons M (2022). Pilot Evaluation of the Impact of Lottery-Based Incentives on Engagement Levels of Male Low SES Vocational Students With an mHealth App. Front. Digit. Health 3:748588. doi: 10.3389/fdgth.2021.748588
- Nuijten, R.C.Y., Van Gorp, P.M.E., Kaymak, U., Le Blanc, P.M., Kemperman, A.D.A.M., van den Berg, P.E.W., & Simons, M. (2018). Gamification for Overweight prevention and Active Lifestyle (GOAL). In Proceedings of the 2018 *Data Science Summit* (DSSE)
- Nuijten, R., Van Gorp, P.M.E., Kaymak, U., Le Blanc, P.M., Kemperman, A.D.A.M., van den Berg, P.E.W., & Simons, M. (2019). Join the GOAL-movement. Postersessie gepresenteerd op Data Science Summit 2019, Eindhoven, Nederland.
- Nuijten, R., Van Gorp, P., Kaymak, U., Simons, M., Kemperman, A., Van Den Berg, P.E.W. (2019). Evaluation of the impact of extrinsic rewards on user engagement in a health promotion context. Proceedings of the Annual International Conference of the IEEE Engineering in Medicine and Biology Society, EMBS. July 2019, Article number 8856296, Pages 3600-3604. 41st Annual International Conference of the IEEE Engineering in Medicine and Biology Society, EMBC 2019; Berlin; Germany; 23 – 27 July 2019.
- Nuijten, R.C.Y., Van Gorp, P., Khanshan, A., Le Blanc, P.M., van den Berg, P. E. W., Kemperman, A.D.A.M., & Simons, M. (2021/Geaccepteerd/In druk). Evaluating the impact of personalized goal setting on engagement levels of government staff with a gamified mHealth tool: results from a two-month randomized intervention trial. *Journal of Medical Internet Research*.
- Nuijten, R., Van Gorp, P., Khanshan, A., Le Blanc, P., Kemperman, A., van den Berg, P., & Simons, M. (2021). Health Promotion through Monetary Incentives: Evaluating the Impact of Different Reinforcement Schedules on Engagement Levels with a mHealth App. Electronics. 2021; 10(23):2935. https://doi.org/10.3390/electronics10232935
- Nuijten, R.C.Y., Van Gorp, P.M.E., Van den Dungen, K.C.R., Kaymak, U., Le Blanc, P.M., & Dijstelbloem, F. (2018). Applying gamification to improve quality of life: exploration of the effectiveness of a game that empowers users to improve their quality of life. Postersessie gepresenteerd op *ICT.OPEN 2018*, Amersfoort, Nederland.
- Zhang C, van Gorp P, Derksen M, Nuijten R, IJsselsteijn WA, Zanutto A, Melillo F, & Pratola R. (2021). Promoting Occupational Health through Gamification and E-Coaching: A 5-Month User Engagement Study. International Journal of Environmental Research and Public Health. 2021; 18(6):2823. https://doi.org/10.3390/ijerph18062823

REFERENCES

- Cugelman, B. (2013). Gamification: What It Is and Why It Matters to Digital Health Behavior Change Developers. JMIR Serious Games 2013;1(1):e3 doi: 10.2196/games.3139
- DeSmet A., Van Ryckeghem D., Compernolle S., Baranowski T., Thompson D., Crombez G., Poels K., Van Lippevelde W., Bastiaensens S., Van Cleemput K., Vandebosch H., & De Bourdeaudhuij, I. (2014). A meta-analysis of serious digital games for healthy lifestyle promotion. Prev Med. 2014 Dec;69:95-107. doi: 10.1016/j.ypmed.2014.08.026. Epub 2014 Aug 27. PMID: 25172024; PMCID: PMC4403732.
- Deterding, S., Dixon, D., Khaled, R., Nacke, L., (2011) From game design elements to gamefulness," in Proceedings of the 15th International Academic MindTrek Conference on Envisioning Future Media Environments - MindTrek '11, ACM Press, 2011. doi: 10.1145/2181037.2181040.
- Khanshan, A., Van Gorp, P., Nuijten, R., & Markopoulos, P. (2021). Assessing the Influence of Physical Activity Upon the Experience Sampling Response Rate on Wrist-Worn Devices. Int. J. Environ. Res. Public Health 2021, 18, 10593. https://doi.org/10.3390/ ijerph182010593
- Nuijten, R.C.Y. (2022). Energize: Exploring the impact of gamification strategies on engagement with mobile health apps. Eindhoven University of Technology.
- Nuijten, R. & Van Gorp, P. (2021). SciModeler: A Metamodel and Graph Database for Consolidating Scientific Knowledge by Linking Empirical Data with Theoretical Constructs. In Proceedings of the 9th International Conference on Model-Driven Engineering and Software Development - MODELSWARD, pp. 314-321. DOI: 10.5220/0010315503140321
- Nuijten, R.C.Y., Van Gorp, P.M.E., Kaymak, U., Le Blanc, P.M., Kemperman, A.D.A.M., van den Berg, P.E.W., & Simons, M. (2018). Gamification for Overweight prevention and Active Lifestyle (GOAL). In Proceedings of the 2018 Data Science Summit (DSSE).
- Tabak M., Dekker-van Weering M., van Dijk H., & Vollenbroek-Hutten M. (2015). Promoting Daily Physical Activity by Means of Mobile Gaming: A Review of the State of the Art. Games Health J. 2015 Dec;4(6):460-9. doi:10.1089/g4h.2015.0010. Epub 2015 Sep 23. PMID: 26397179.

- Amabile, T.M. & Pratt, M.G. (2016). "The dynamic componential model of creativity and innovation in organizations: Making progress, making meaning". Research in Organizational Behavior. 36: 157–183.
- Ausin, J., & Van Dijk, J. & Drossaert, C. (2020). When theory meets users in co-design: four strategies towards synergy between bottom-up and top-down input. Design Research Society, DRS 2020: Synergy - Online, Brisbane, Australia. 10.21606
- Beck, J., & Stolterman, E. (2016). Examining the Types of Knowledge Claims Made in Design Research. She Ji: The Journal of Design, Economics, and Innovation, 2(3), 199–214. https://doi.org/10.1016/j.sheji.2017.02.001
- Blandford, A., J. Gibbs, N. Newhouse, O. Perski, A. Singh, & E. Murray. (2018). "Seven Lessons for Interdisciplinary Research on Interactive Digital Health Interventions." *Digital Health* 4:1-13
- Brankaert, R., & den Ouden, E. (2017). The Design-Driven Living Lab: A New Approach to Exploring Solutions to Complex Societal Challenges. *Technology Innovation Management Review*, 7(1), 44–51.
- Burns, P.B., Rohrich, R.J. & Chung, K.C. 2015. "The Levels of Evidence and Their Role in Evidence-Based Medicine". *Plastic and Reconstructive Surgery* 128 (1):305–10.
- Choi, B., & Pak, A. (2006). Multidisciplinarity, interdisciplinarity and transdisciplinarity in health research, services, education and policy: 1: Definitions, objectives and evidence of effectiveness. *Clin Invest Med*, *29*(6), 351–364.
- Council for Public Health and Society (2017). No evidence without context. About the illusion of evidence based practice in healthcare. Report.
- Durrant, A.C., Vines, J., Wallace, J., & Yee, J.S.R. (2017). Research through design: Twenty-first century makers and materialities, *Design Issues*, 33(3), 3–10.
- Gaver, W., & Bowers, J. (2012). Annonated Portfolios. Interactions, 40–49.
- Godfroij, B.K. (2021). A process model on changing design practice: An industrial design case in a healthcare innovation journey. *12th International Process Symposium*. Rhodes Greece, online, 1–5 September, 2021.
- Godfroij, B., Peek, S., Van der Lugt, R., Brankaert, R. (accepted). Changing Roles in Interdisciplinary Research-through-Design. *The 13th Design Thinking Research Symposium* (DTRS), 2022 Israel.
- Godfroij, B., Van Beest, W. & Van Gessel, C. (2020). Co-design in de anderhalvemeter-maatschappij, Whitepaper. Utrecht University of Applied Sciences, Utrecht, The Netherlands.
- Groeneveld, B., Dekkers, T., Boon, B. & D'Olivo, P. (2019) Challenges for design researchers in healthcare, *Design for Health*
- Hevner, A.R. (2007) A Three Cycle View of Design Science Research. Scandinavian Journal of Information Systems, 19/2: 87-92.
- Hermsen, S., Van Essen, A., Van Gessel, C., Bolster, E., Van der Lugt, R., & Bloemen, M. (2020). Are agile design approaches useful in designing for health? A case study. Proceedings of the 6th International Conference on Design4Health Amsterdam, Netherlands, 1st - 3rd July 2020. Editors: Christer, C., Craig, C. & Chamberlain, P. Lab4Living, Sheffield Hallam University

- Hoök, K., & Lowgren, J. (2012). Strong concepts: Intermediate-level knowledge in interaction Design research. In ACM Transactions on Computer-Human Interaction (Vol. 19).
- Horst, W. (2011). *Prototypes as platforms for participation* (PhD dissertation). University of Southern Denmark, Sønderborg.
- Howick, J., P. Glasziou, & J.K. Aronson (2009). "The Evolution of Evidence Hierarchies: What Can Bradford Hill's 'guidelines for Causation' Contribute?" *JRSM* 1 (102(5)):186–94.
- Keller, A. I. (2005). For inspiration only. Designer interaction with informal collections of visual material. Technical University, Delft, The Netherland.
- Koskela-Huotari, K., Friedrich, P., & Isomursu, M. (2013). Jungle of "co". In. Proceedings of the 2013 Naples Forum on Service.
- Koskinen, I., Zimmerman, J., Binder, T., Redstrom, J., & Wensveen, S. (2011). Design Research Through Practice: From the Lab, Field, and Showroom. Elsevier. https://doi. org/10.1016/B978-0-12-385502-2.00010-9
- Löwgren, J. (2013). Annotated Portfolios and Other Forms of Intermediate- Level Knowledge. Interactions, (February), 30–34.
- Mann, D., Kuppin Chokshi, S. & Kushniruk, A. (2018) Bridging the Gap Between Academic Research and Pragmatic Needs in Usability: A Hybrid Approach to Usability Evaluation of Health Care Information Systems. *JMIR Human Factors* 5, no. 4 (November 28): e10721. doi:10.2196/10721.
- Mattelmäki, T. & Sleeswijk Visser, F. (2011), Lost In Co-X: Interpretations Of Co-Design And Co-Creation. *IASDR2011, The 4th World Conference on Design Research*. Delft, the Netherlands
- McCann, T.V., & Clark, E. (2003). Grounded theory in nursing research: Part 1–Methodology .
- McComb, C., & Jablokow, K. (2022). A Conceptual Framework for Multidisciplinary Design Research with Example Application to Agent-Based Modeling. *Design Studies*.
- Moors, E.H.M. (2013). *Duurzaam innoveren: de kunst van het verbinden*. Inaugurele rede. Utrecht:Universiteit Utrecht
- Peffers, K., Tuunanen, T., Rothenberger, M. A., & Chatterjee, S. (2008). "A Design Science Research Methodology for Information Systems Research," *Journal of Management Information Systems* (24:3), pp. 45-77.
- Portney, L.G., & M.P. Watkins. 2000. Foundations of Clinical Research: Applications to Practice. New Jersey: Prentice Hall.
- Reay, S., G. Collier, J. Kennedy-Good, A. Old, R. Douglas, & A. Bill. (2017). Designing the Future of Healthcare Together: Prototyping a Hospital Co-Design Space. *CoDesign* 13 (4): 227–244
- Roggema, R. (2017). Research by Design: Proposition for a Methodological Approach. Urban Science, 1(2), 1–19.
- Sanders, L. (2008). An evolving map of design practice and design research, XV(December), 1–7.
- Sleeswijk Visser, F. (2018). Structuring roles in Research through Design collaboration. In C. Storni, K. Leahy, M. McMahon, P. Lloyd, & E. Bohemia (eds.), *Proceedings of DRS* 2018 (pp. 368–380), Limerick, Ireland, 25–28 June, 2018.

- Smeenk, W., Sturm, J., & Eggen, B. (2022). A comparison of existing frameworks leading to an empathic formation compass for co-design A Comparison of Existing Frameworks Leading to an Empathic Formation Compass for Co-design, *13*(2019), *53–68*.
- Stappers, P. & Giaccardi, E. (2017). Research through Design. In M. Soegaard, & R. Friis-Dam (eds.), The Encyclopedia of Human-Computer Interaction, (2nd edition, pp. 1–94). The Interaction Design Foundation.
- Stappers, P. J., Sleeswijk Visser, F., & Keller, A. I. (2015). The role of prototypes and frameworks for structuring explorations by research through design. *The Routledge Companion to Design Research*, 167–174. https://doi.org/10.4324/9781315758466-20
- Stappers, P. J., & Van der Lugt, R. (2006). Design and the Growth of Knowledge Best Practices and Ingredients for Successful Design Research.
- Stokes, D. (1997). Pasteur's quadrant: Basic science and technological innovation. Washington D.C.: Brookings Institution Press.
- Wensveen, S., & Matthews, B. (2015). Prototypes and Protoyping in Design Reserch. *Routledge Companion to Design Research*, (January), 262–276.
- Van Aken, J. E. (2005). Management research as a design science: Articulating the research products of Mode 2 knowledge production in management. *British Journal of Management*, 16: 19–36.
- Van de Ven, A.H., Polly, D.E., Garud, R., Venkataraman, S. (1999). The Innovation Journey. Oxford University Press, Inc. New York.Wensveen, S., & Matthews, B. (2014). Prototypes and prototyping in design research. In P.A. Rodgers & J. Yee (Eds.), *Routledge Companion to Design Research* (pp. 262--276) London: Routledge.
- Van der Lugt, T. & Van der Laan, T. (2017) Design facilitators' journeys through the jungle of Co- in healthcare, *The Design Journal*
- Wensveen, S., & Matthews, B. (2015). Prototypes and Protoyping in Design Reserch. Routledge Companion to Design Research, (January), 262–276.
- Zielhuis, M., Sleeswijk Visser, F., Andriessen, D., & Stappers, P. J. (2022). Making design research relevant for design practice: what's in the way? *Design Studies*, https://authors. elsevier.com/sd/article/S0142694X21000740
- Zielhuis, M., Sleeswijk Visser, F., Andriessen, D., Stappers, P.J. (in press). What makes design research more useful for design professionals? An exploration of the research-practice gap. *Journal of Design Research.*
- Zimmerman, J., Forlizzi, J., & Evenson, S. (2007) Research through design as a method for interaction design research in HCI. SIGCHI Conference on Human Factors in Computing Systems (pp. 493–502).

- Aarikka-Stenroos, L., Jaakkola, E., Harrison, D. & Mäkitalo-Keinonen, T. (2017). How to manage innovation processes in extensive networks: A longitudinal study. *Industrial Marketing Management*, 67: 88-105.
- Arnstein, S.R. (1969) A ladder of citizin participation. J Am Inst Plann. 1969;35(4):216–24.
- Boaz, A., Hanney, S., Borst, R. et al. (2018). How to engage stakeholders in research: design principles to support improvement. *Health Res Policy Sys*, 16, 60.

- Concannon, T.W., Grant, S., & Welch, V. (2019) Practical Guidance for Involving Stakeholders in Health Research. *J Gen Intern Med* 34, 458–463. https://doi.org/10.1007/s11606-018-4738-6.
- D'Este P., Ramos-Vielba I, Woolley R. & Amara N. (2018). How do researchers generate scientific and societal impacts? Toward an analytical and operational framework. *Science and Public Policy*, 45(6): 752-763.
- Dewaele A., Vandael K., Meysman S., & Buysse A. (2021). Understanding collaborative interactions in relation to research impact in social sciences and humanities: A meta-eth-nography, *Research Evaluation*, 30, 2:179–190, https://doi.org/10.1093/reseval/rvaa033
- De Wit M.P.T., Kvien T.K., & Gossec L. (2015) Patient participation as an integral part of patient-reported outcomes development ensures the representation of the patient voice: a case study from the field of rheumatology. *RMD Open.*; 1(1). https://doi.org/10.1136/ rmdopen-2015-000129.
- Geissbuhler, A. (2013). Lessons learned implementing a regional health information exchange in Geneva as a pilot for the Swiss national eHealth strategy. International Journal of Medical Informatics, 82(5), e118-e124.
- Hevner, A. R. (2007) 'A Three Cycle View of Design Science Research', Scandinavian Journal of Information Systems, 19: 87–92.
- Joiner, K. & Lusch, R. (2016). Evolving to a new service-dominant logic for health care. Innovation and Entrepreneurship in Health. 3: 25-33.
- Kok, M.O., & Schuit, AJ. (2012) 'Contribution Mapping: A Method for Mapping the Contribution of Research to Enhance Its Impact', *Health Research Policy and Systems*, 10: 21.
- Lemmens, L., De Bruin, S., Struijs, J., Rijken, M., Nijpels, G. & Baan, C. (2015). Patient involvement in diabetes care: experiences in nine diabetes care groups. Int J Integr Care;15. https://doi.org/10.5334/ijic.2207.
- Merkel S., & Kucharski A. (2019). Participatory Design in Gerontechnology: A Systematic Literature Review, *The Gerontologist*, 59, 1: 16–25. Nicolas G., Foote J., Kainz K., Midgley G., Prager K., Zurbriggen C. (2019). Towards a heart and soul for co-creative research practice: a systemic approach. *Evidence & Policy*, 3: 353-370.
- Nielsen, A.J., & Mathiassen, L. (2013). Interpretive Flexibility in Mobile Health: Lessons from a Government-Sponsored Home Care Program. *Journal of Medical Internet Research*, 15(10): e236.
- Nijland, N. (2011). Grounding eHealth: towards a holistic framework for sustainable eHealth technologies. University of Twente. https://doi.org/10.3990/1.9789036531337
- Pagliari, C. (2007) Design and evaluation in eHealth: challenges and implications for an interdisciplinary field. *J Med Internet Res* ;9(2).
- Peine, A., Faulkner, A., Jæger, B. & Moorsa, E. (2014). Science, Technology and the 'Grand Challenge' of Ageing – Understanding the Socio-Material Constitution of Later Life. Technological Forecasting and Social Change 93: 1–9.
- Pieterse, M., Kip, H. & Cruz-Martinez, R.R. (2018). The complexity of eHealth Implementation: A theoretical and practical perspective. Chapter 12.
- Reypens, C., Lievens, A. & Blazevic, V. (2016). Levering value in multi-stakeholder innovation networks: A process framework for value creation and capture. *Industrial Marketing Management*, 56: 40-50.
- Saapen, J., & van Drooge, L. (2011) 'Productive Interactions as a Tool for Social Impact Assessment of Research', *Research Evaluation*, 20: 211–8.

- Sanders, E.B.N., & P.J. Stappers (2008). Co-Creation and the New Landscapes of Design. *CoDesign* 4 (1): 5–18.
- Smits D.W., Van Meeteren K., Klem M., Alsem M. & Ketelaar M. (2020). Designing a tool to support patient and public involvement in research projects: the Involvement Matrix. *Res Involv Engagem* 6, 30 (2020). https://doi.org/10.1186/s40900-020-00188-4.
- Van Beest. W, Boon P.C.W., Andriessen D., Pol H., Van der Veen G., & Moors E.H.M. (2021). A Research Pathway Model for evaluating the implementation of practice-based research: The case of self-management health innovations, *Research Evaluation*, 2021, https://doi.org/10.1093/reseval/rvab023
- Van Gemert-Pijnen L., Kelders S.M., Kip H., & Sanderman R. (2018) *eHealth Research Theory and Development, A Multidisciplinary Approach*. Londen and New York: Routledge Taylor & Francis Group.
- Van Gemert-Pijnen, L., Nijland, N., van Limburg, M., Ossebaard, H.C., Kelders, S.M., Eysenbach, G., & Seydel, E.R. (2011) A Holistic Framework to Improve the Uptake and Impact of eHealth Technologies. J Med Internet Res;13(4)
- Van Limburg, M., Wentzel, J., Sanderman, R., & van Gemert-Pijnen, L. (2015). Business Modeling to Implement an eHealth Portal for Infection Control: A Reflection on Co-Creation with Stakeholders. *JMIR Research Protocols*, 4(3): e104
- Vargo, S. & Lusch, R. (2004) Evolving to a new dominant logic in marketing. *Journal of Marketing*, 68 (1): 1-17.
- Vargo, S. & Lusch, R. (2008) Service-dominant logic: Continuing the evolution. *Journal of the Academy of Marketing Science*, 36 (1): 1-10.
- Vargo, S. & Lusch, R. (2016) Institutions and axioms: an extension and update of service-dominant logic. *Journal of the Academy of Marketing Science*, 44:5-23.
- Vargo S., Koskela-Huotari K & Vink J. (2020). Service-Dominant Logic Foundations and Applications. The Routledge Handbook of Service Research Insights and Ideas.

- Akkerman, S.F., & Bakker, A. (2011). Boundary crossing and boundary objects. *Review of Educational Research*, 81(2), 132–169.
- Akkerman, S. & Bruining, T. (2016). Multilevel Boundary Crossing in a Professional Development School Partnership, *Journal of the Learning Sciences*, 25:2, 240-284.
- Andriessen, D., Zielhuis, M., Greven, K., Van Beest, W., Godfroij, B., & Van der Lugt, R. (2019). Seven ways to foster interdisciplinary collaboration in research involving healthcare and creative research disciplines. In R. Brankaert & W. IJsselsteijn (eds.). Communications in Computer and Information Science, 1117(2019), 174–177.
- Andriessen, D., Godfroij, B., Greven, K., & Zielhuis, M. (2020). Collaborating in Complexity: Strategies for interdisciplinary collaboration in design research. In K. Christer, C. Craig, & P. Chamberlain, (eds.), Proceedings of the 6th International Conference on Design-4Health, (Vol.1, pp. 11–17). Sheffield: Sheffield Hallam University.
- Baartman, L.K.J., & De Bruijn, E. (2011). Integrating knowledge, skills and attitudes: Conceptualising learning processes towards vocational competence. *Educational Research Review*, Vol.6 (2), pp.125-134.

- Bakker, A., Zitter, I., Beausaert, S., & De Bruijn, E. (2016). Tussen opleiding en beroepspraktijk: het potentieel van boundary crossing. Assen: Koninklijke Van Gorcum.Bakker, A. & Akkerman, S.F. (2013). A boundary-crossing approach to support students' integration of statistical and work-related knowledge. Educ Stud Math86, 223–237
- Basballe, D.A., & Halskov, K. (2012). Dynamics of research through design. In *Proceedings* of the Designing Interactive Systems Conference, pp. 58–67. New York: Association for Computing Machinery.
- Carlille, P.R. (2002). A Pragmatic View of Knowledge and Boundaries: Boundary Objects in New Product Development.
- Carlille, P. (2004). Transferring, Translating, and Transforming: An integrative Framework for Managing Knowledge Across Boundaries.
- Durrant, A.C., Vines, J., Wallace, J., & Yee, J.S.R. (2017). Research through design: Twenty-first century makers and materialities, *Design Issues*, 33(3), 3–10.
- Engeström, Y. (1987). Learning by Expanding: An Activity-theoretical approach to developmental research. Helsinki, Finland: Orienta-Konsultit.
- Engeström, Y., Engeström, R., & Kärkkäinen, M. (1995). Polycontextuality and boundary crossing in expert cognition: Learning and problem solving in complex work activities. *Learning and Instruction*, 5, 319–336.
- Godfroij, B.K. (2021). A process model on changing design practice: An industrial design case in a healthcare innovation journey. *12th International Process Symposium*. Rhodes Greece, online, 1–5 September, 2021.
- Godfroij, B., Peek, S., Van der Lugt, R., & Brankaert, R. (2022). Changing Roles in Interdisciplinary Research-through-Design. *The 13th Design Thinking Research Symposium* (DTRS), 2022 Israel.
- Godfroij, B., van Beest, & van Gessel (2020). *Co-design in de anderhalvemetermaatschappij* Whitepaper. Utrecht University of Applied Sciences, Utrecht, The Netherlands.
- Godfroij, B., & van der Lugt, R. (2020). Change and Novelty for Industrial Designers in Complex Design Projects for Healthcare. In K. Christer, C. Craig, & P. Chamberlain, (eds.), Proceedings of the 6th International Conference on Design4Health, Vol.1, pp. 186–193. Sheffield: Sheffield Hallam University.
- Godin, D., & Zahedi, M. (2014). Aspects of Research through Design: A Literature Review. In Y. Lim, K. Niedderer, J. Redström, E. Stolterman, & A. Valtonen, (eds.), *Design's Big Debates - DRS International Conference 2014*, Umeå, Sweden, 16–19 June, 2014.
- Stappers, P. & Giaccardi, E. (2017). Research through Design. In M. Soegaard, & R. Friis-Dam (eds.), The Encyclopedia of Human-Computer Interaction, 2nd edition, pp. 1–94. The Interaction Design Foundation.
- Star, S. L. (1988). The structure of ill-structured solutions: Boundary objects and heterogeneous distributed problem solving. In *Readings in distributed artificial intelligence*, ed. M. Huhns & L. Gasser. Menlo Park, CA: Kaufman.
- Star, S. L. (2010). This is not a boundary object: Reflections on the origin of a concept. *Science*, *Technology*, & *Human Values*, 35(5), 2010, 601–617.
- Star, S.L., & Griesemer, J.R. (1989). Institutional Ecology, 'Translations' and Boundary Objects: Amateurs and Professionals in Berkeley's Museum of Vertebrate Zoology, 1907-39. Social Studies of Science, Vol.19(3), pp.387-420.
- Suchman, L. (1994). Working relations of technology production and use. *Computer Supported Cooperative Work*, 2, 21–39.

- Van Beest, W., Boon, W.P.C., Andriessen, D., Pol, H., van der Veen, G., & Moors, E.H.M. (2021). A Research Pathway Model for evaluating the implementation of practice-based research: The case of self-management health innovations, *Research Evaluation*, Vol.31(1), pp.24–48.
- Van der Lugt, R., & Van der Laan, T. (2017). Design facilitators' journeys through the jungle of Co- in healthcare. *The Design Journal, 20*(2017), 2057–2067.
- Vargo, S.L., Koskela-Huotari, K. & Vink, J. (2020). Service-Dominant Logic: Foundations and Applications. In E. Bridges & K. Fowler (eds.), *The Routledge Handbook of Service Research Insights and Ideas*, (1st edition, pp. 1–21). London: Routledge.
- Vargo, S.L., Lusch, R.F. (2008). Service-dominant logic: continuing the evolution. J. of the Acad. Mark. Sci.36, 1–10.
- Watling Neal, J., Neal, Z., & Brutzman, B. (2021). Defining brokers, intermediaries, and boundary spanners: a systemic review. Evidence & Policy A Journal of Research Debate and Practice. Policy Press, Vol.18 (1), pp.7-24.Wenger, E. (1998). Communities of Practice: Learning as a social system. Systems Thinker. Vol.9(5), pp.1-10
- Zielhuis, M., Van Gessel, C., Van der Lugt, R., Godfroij, B. & Andriessen, D. (2020). Grounding Practices: How researchers ground their work in create-health collaborations for designing e-health solutions. In K. Christer, C. Craig, & P. Chamberlain, (eds.), *Proceedings* of the 6th International Conference on Design4Health (Vol.3, pp. 142–152). Sheffield: Sheffield Hallam University.
- Zielhuis, M., Sleeswijk Visser, F., Andriessen, D., Stappers, P.J. (2022). Making design research relevant for design practice: What is in the way?. *Research Notes. Design Studies.* Vol.78, No.C January 2022.
- Zimmerman, J., & Forlizzi, J. (2014). Research Through Design in HCI. In J.S. Olson & W.A. Kellogg (eds.), *Ways of Knowing in HCI* (pp. 167–189). New York: Springer.

- Blandford, A., Gibbs, J., Newhouse, N., Perski, O., Singh, A., & Murray, E. (2018). Seven lessons for interdisciplinary research on interactive digital health interventions. *Digital Health*, 4, 205520761877032.
- Brett, J., Staniszewska, S., Mockford, C., Herron-Marx, S., Hughes, J., Tysall, C., & Suleman, R. (2014). Mapping the impact of patient and public involvement on health and social care research: A systematic review. *Health Expectations*, 17(5), 637–650.
- Choi, B.C.K. & Pak, A.W.P., (2006). Multidisciplinarity, interdisciplinarity, and transdisciplinarity in health research, services, education and policy: 1. Definitions, objectives, and evidence of effectiveness. *Clinical and Investigative Medicine*, *29*(6), 351–364. https:// doi.org/10.25011/cim.v30i6.2950
- Mogensen, P. (1992). Towards a provotyping approach in systems development. Scandinavian Journal of Information Systems, 4(1), 5.
- Sleeswijk Visser, F., Stappers, P.J., van der Lugt, R., & Sanders, E.B.-N. (2005) Contextmapping: Experiences from practice. CoDesign: International Journal of CoCreation in Design and Arts, 1(2), 119–149.

Zielhuis, M., Van Gessel, C., van der Lugt, R., Godfroij, B. K., & Andriessen, D. (2020). Grounding practices: How researchers ground their work in create-health collaborations for designing e-health solutions. In K. Christer, C. Craig, & P. Chamberlain (Eds.), *Proceedings of the 6th European Conference on Design4Health*, 1st-3rd July 2020 Amsterdam. Amsterdam.

